UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

\mathbf{F}	\mathbf{O}	R	M	8-	.K
1	\mathbf{v}	1	Т ₹ Т	U-	- T Z

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report: (Date of earliest event reported): January 13, 2023

Calyxt, Inc.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation)

following provisions:

001-38161 (Commission File Number) 27-1967997 (IRS Employer Identification No.)

2800 Mount Ridge Road Roseville, MN 55113-1127

(Address and zip code of principal executive offices)

(651) 683-2807

(Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the

\boxtimes	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)			
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)			
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))			
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))			
Secu	Securities registered pursuant to Section 12(b) of the Act:			
	Title of each class	Trading Symbol(s)	Name of exchange on which registered	
Common Stock, par value \$0.0001 per share		CLXT	The NASDAQ Global Market	
Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter). Emerging growth company				
If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.				

Item 1.01. Entry into a Material Definitive Agreement.

Merger Agreement

On January 13, 2023, Calyxt, Inc., a Delaware corporation ("Calyxt"), and Calypso Merger Subsidiary, LLC, a Delaware limited liability company and wholly-owned subsidiary of Calyxt ("Merger Subsidiary") entered into an Agreement and Plan of Merger (the "Merger Agreement," and the transactions contemplated thereby, the "Transactions") with Cibus Global LLC, a Delaware limited liability company ("Cibus"); New Ventures I Holdings, LLC, a Delaware limited liability company ("Blocker 1"); BCGF CB Holdings LLC, a Delaware limited liability company ("Blocker 3"); BCGFK CB Holdings LLC, a Delaware limited liability company ("Blocker 5"); FSBCGF CB Holdings LLC, a Delaware limited liability company ("Blocker 5"); FSGRWCO CB Holdings LLC, a Delaware limited liability company ("Blocker 7"); GROWTHCO CB Holdings LLC, a Delaware limited liability company ("Blocker 7"); GROWTHCO CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); Blocker 9. Blocker 9. Blocker 9, the "Blocker 9, the "Blockers").

The Merger Agreement provides that, among other things and upon the terms and subject to the conditions set forth therein (including receipt of Calyxt stockholder approval and Cibus member approval), on the date (the "Closing Date") of the closing of the Transactions (the "Closing"): (a) each of the Blockers will merge with and into Calyxt (each a "Blocker Merger," and collectively the "Blocker Mergers"), in each case with the Blockers ceasing to exist and Calyxt surviving each such Blocker Merger (the resulting company, the "Resulting Public Company"), (b) at the effective time of the Blocker Merger with Blocker 1 (the "First Blocker Merger Effective Time"), Calyxt's amended and restated bylaws will be amended and restated in a manner to be mutually agreed by Calyxt and Cibus (as amended and restated, the "Amended Bylaws") and Calyxt's amended and restated, the "Amended Certificate of incorporation will be amended and restated in a manner to be mutually agreed by Calyxt and Cibus (as amended and restated, the "Amended Certificate of Incorporation"), (c) following the Blocker Mergers, Merger Subsidiary will merge with and into Cibus (the "Cibus Merger" and, collectively with the Blocker Mergers, the "Mergers"), with Cibus as the surviving company and Merger Subsidiary ceasing to exist. As consideration for the Mergers, the equity holders of the Blockers and the equity holders of Cibus will receive the Merger Consideration (as defined below). In connection with the Mergers, the Resulting Public Company will contribute all of its assets and liabilities to Cibus, as a contribution to the capital of Cibus, in exchange for newly issued membership units of Cibus under Cibus's Amended Operating Agreement (as defined below) ("Cibus Common Units"), pursuant to a contribution agreement in a form to be agreed by Calyxt and Cibus.

The Closing is subject to the approval of Calyxt's stockholders, the approval of Cibus's members, the receipt of required regulatory approvals (to the extent applicable) and satisfaction of other customary closing conditions, as more particularly described below and in the Merger Agreement. The Closing is currently expected to occur in the second quarter of 2023.

Merger Consideration

The Amended Certificate of Incorporation, once adopted at the First Blocker Merger Effective Time, will create two classes of Resulting Public Company common stock: (i) Class A Common Stock, par value \$0.0001 (the "Class A Common Stock"), which shares will have full voting and economic rights and (ii) Class B Common Stock, par value \$0.0001 (the "Class B Common Stock"), which shares will have full voting, but no economic rights. Upon adoption of the Amended Certificate of Incorporation, each share of Calyxt common stock existing and outstanding immediately prior to the First Blocker Merger Effective Time shall automatically convert to a share of Class A Common Stock at the First Blocker Merger Effective Time without any further action by any current holders of Calyxt common stock (the "Redesignation").

At the applicable effective time of each Blocker Merger, the limited liability company interests of the applicable Blocker will be cancelled and converted into the right to receive shares of Class A Common Stock corresponding to that Blocker Merger (the "*Blocker Merger Consideration*"), in each case, as set forth in the allocation schedule attached to the Merger Agreement (as such allocation schedule shall be amended and revised from time to time prior to the Closing, the "*Allocation Schedule*").

At the effective time of the Cibus Merger, all of the issued and outstanding membership units of Cibus, and all warrants or options to purchase membership units of Cibus, will be cancelled and converted into the right to receive the following, as described in further detail in the Merger Agreement (the "Cibus Merger Consideration" and together with the Blocker Merger Consideration, collectively, the "Merger Consideration"):

- any pre-Closing holder (excluding the Blockers, but including any investor who has delivered an irrevocable commitment as of the date of the Merger Agreement to purchase Cibus membership units prior to Closing) of Cibus membership units (other than profits interest units of Cibus) (collectively, the "*Top 99 Holders*") shall have the option to elect, on a per unit basis, to receive Merger Consideration in the form of either (a) the number of shares of Class A Common Stock set forth on the Allocation Schedule in exchange for all such holder's Cibus membership units covered by such election or (b) a number of shares of Class B Common Stock, with an equal number of non-voting Cibus Common Units set forth on the Allocation Schedule (a non-voting Cibus Common Unit and a share of Public Company Class B Common Stock, collectively, an "*Up-C Unit*"), provided, that such holder will not be entitled to exchange more than 95% of such holder's existing Cibus membership units for Up-C Units and the remainder of such holder's Cibus membership units will be exchanged for shares of Class A Common Stock (all Top 99 Holders who elect to receive Up-C Units will be referred to as "*Electing Members*");
- all pre-Closing holders of Cibus membership units (excluding Blockers) other than Top 99 Holders shall receive the number of shares of Class A Common Stock set forth on the Allocation Schedule in exchange for their Cibus membership units (other than profits interest units of Cibus);
- all pre-Closing warrants to purchase Cibus membership units shall automatically be exchanged for the right to receive the number of shares of Class A Common Stock set forth on the Allocation Schedule; and
- all pre-Closing profits interest units of Cibus will, (a) to the extent not a "restricted unit" (under the award agreements and governing documents applicable to such profits interest unit), be automatically cancelled and converted into the right to receive the number of shares of Class A Common Stock set forth on the Allocation Schedule and (b) to the extent a "restricted unit" (or unvested unit), be automatically cancelled and converted into the right to receive the number of restricted shares of Class A Common Stock pursuant to a Resulting Public Company stock plan, as set forth on the Allocation Schedule and subject to the same vesting schedule as was applicable to such profits interest unit prior to Closing.

As set forth in further detail on the Allocation Schedule:

- Each issued and outstanding Cibus membership unit immediately prior to the Closing will, effective as of the Closing, be converted into the right to receive, subject to the election of such holder (to the extent a Top 99 Holder), 3.17046 shares of either (i) Public Company Class A Common Stock or (ii) Public Company Class B Common Stock (along with a corresponding number of Cibus Common Units).
- Each issued and outstanding Cibus profits interest unit as of immediately prior to the Closing will, effective as of the Closing, be converted into the right to receive a number of shares of Public Company Class A Common Stock. The number of shares of Public Company Class A Common Stock to be issued in consideration for Cibus profits interest units was calculated based on a Black Scholes valuation of such Cibus profits interest units, with the exercise price input used in such Black Scholes valuation being a present value calculation of the applicable threshold value associated with the underlying Cibus profits interest unit. The conversion factor for each Cibus profits interest unit will be determined by dividing the Black Scholes valuation by \$2.00 (the price at which Cibus most recently sold Cibus Series F Preferred Units prior to the date of the Merger Agreement).

• Each issued and outstanding warrant to purchase Cibus membership units immediately prior to the Closing will, effective as of the Closing, be exchanged for the right to receive a number of shares of Public Company Class A Common Stock. The number of shares of Public Company Class A Common Stock to be issued in consideration of such warrants will be calculated based on a Black Scholes valuation of such warrant. The conversion factor for each such warrant will be determined by dividing the Black Scholes valuation by \$2.00 (the price at which Cibus most recently sold Cibus Series F Preferred Units prior to the date of the Merger Agreement).

Immediately following the Cibus Merger, (i) each pre-Closing Cibus membership unit and pre-Closing warrant to purchase Cibus membership units, in each case, held by Resulting Public Company as a result of the Blocker Mergers will be converted or exchanged for a number of Cibus Common Units set forth in the Allocation Schedule, (ii) each pre-Closing warrant to purchase Cibus membership units held by Resulting Public Company as a result of the warrant exchanges described above will be converted or exchanged for a number of Cibus Common Units set forth in the Allocation Schedule, and (iii) Cibus will issue a number of Cibus Common Units to Resulting Public Company in an amount equal to the sum of (A) the number of shares of Class A Common Stock issued to former Cibus equity holders as Cibus Merger Consideration and (B) the number of shares of Class A Common Stock resulting from the Redesignation.

Following the Closing and the issuance of the Merger Consideration and the issuance of Cibus Common Units discussed above, the combined company will be organized in an "Up-C" structure, and Resulting Public Company's only material assets will consist of Cibus Common Units. Upon the Closing, pre-Closing Calyxt stockholders are expected to own approximately 5.0% of issued and outstanding common stock of Resulting Public Company, on a fully-diluted, as converted basis, and pre-Closing holders of Cibus membership units and warrants are expected to own approximately 95.0% of the issued and outstanding common stock of Resulting Public Company, on a fully-diluted, as converted basis, in each case, subject to dilution in connection with any additional equity issued by Cibus in connection with a contemplated equity financing between signing the Merger Agreement and Closing and certain employee option grants which Calyxt is authorized to issue between sign and close.

Immediately following the Closing, all of the issued and outstanding Cibus Common Units will be held by Resulting Public Company and the Electing Members (through their ownership of Up-C Units), with the exact ownership interest held by Resulting Public Company and the Electing Members being subject to the elections to be submitted by Top 99 Holders prior to the Closing. At the Closing, Resulting Public Company and the Electing Members will enter into an Exchange Agreement (as defined below), a Tax Receivable Agreement (as defined below) and Cibus's third amended and restated limited liability company agreement (the "Cibus Amended Operating Agreement"), while Resulting Public Company and the Electing Members will enter into a Registration Rights Agreement (as defined below). The Up-C Units will generally be exchangeable for shares of Class A Common Stock on a one-for-one basis pursuant to the terms of the Exchange Agreement, subject to certain procedures and restrictions to be set forth therein.

Representations, Warranties and Covenants

The Merger Agreement contains customary representations and warranties of the parties thereto, as more particularly set forth in the Merger Agreement. Additionally, the Merger Agreement contains customary pre-Closing covenants of the parties, including (i) the obligation of Calyxt, Cibus and their respective subsidiaries to conduct their respective business in the ordinary course and to refrain from taking certain specified actions, subject to certain exceptions, without the prior written consent of certain counterparties to the Merger Agreement and (ii) certain restrictions on Calyxt's and Cibus's ability to solicit alternative acquisition proposals from third parties, provide information to third parties and to engage in discussions with third parties, in each case, in connection with alternative acquisition proposals or inquiries, subject to certain exceptions (the "No-Shop Provision").

Interim Funding

Beginning at the earlier of March 15, 2023 and the date Calyxt's unrestricted cash balance first drops below \$1,500,000, Calyxt can request, and Cibus has agreed to provide, an unsecured, interest-free revolving line of credit of up to \$3,000,000 in cash, which amount may be increased to \$4,000,000 if Cibus elects to extend the Outside Date (as defined below) to June 30, 2023 (the "*Interim Funding*"). Funds can be drawn by Calyxt in \$500,000 increments and may only be used to fund operating expenses incurred in the ordinary course of business consistent with past practice and consistent with the negative covenants in the Merger Agreement. The full outstanding balance of the Interim Funding will be reduced to zero in connection with the Closing if the Mergers are consummated. The full

outstanding balance of the Interim Funding will be forgiven by Cibus if the Merger Agreement is terminated for any reason other than (a) a termination by Cibus resulting from the fraud or willful breach of the Merger Agreement by Calyxt (in which case Calyxt will no longer be permitted to make draws on the line of credit the outstanding balance will become due and payable on the date that is two years from the date funds are first drawn), (b) a termination by Calyxt if it accepts a Superior Offer (in which case Calyxt will no longer be permitted to make draws on the line of credit and the outstanding balance will be due and payable within 30 days after termination) or (c) termination by Cibus in the event Calyxt shall have effected a Public Company Adverse Recommendation Change or Calyxt materially breaches its obligations with respect to soliciting an Acquisition Proposal or Acquisition Inquiry (in which case Calyxt will no longer be permitted to make draws on the line of credit and the outstanding balance will be due and payable within 30 days after termination). The Interim Funding is subject to acceleration in connection with certain bankruptcy events.

Conditions to the Parties' Obligations to Consummate the Transactions

The consummation of the Mergers is subject to certain closing conditions, including (a) the approval of Calyxt's stockholders (the "Calyxt Stockholder Approval") of (i) the Blocker Mergers, (ii) the issuance of the shares of Class A Common Stock and Class B Common Stock to be issued as Merger Consideration pursuant to the Merger Agreement and in accordance with the rules of the Nasdaq Stock Market LLC ("Nasdaq"), (iii) the Amended Certificate of Incorporation and Amended Bylaws and (iv) any other matters necessary or desirable to consummate the Transactions (collectively, the "Calyxt Stockholder Matters"), (b) the receipt of written consents evidencing the approval of the Transactions by the requisite number of pre-Closing holders of Cibus membership units and of the Blockers, (c) the effectiveness of the registration statement on Form S-4 (the "Form S-4") to be filed by Calyxt with the U.S. Securities and Exchange Commission (the "SEC") pursuant to the Merger Agreement, registering the issuance of shares of Class A Common Stock to be issued in the Mergers, (d) the approval by Nasdaq for listing of the shares of Class A Common Stock to be issued as Merger Consideration, (e) the entry into an amendment to the License Agreement, dated July 25, 2017, between Calyxt and Cellectis S.A., (f) the absence of any judgment or law issued by any governmental entity enjoining or otherwise prohibiting the consummation of the Mergers, (g) execution of the Exchange Agreement, the Tax Receivable Agreement, the Cibus Amended Operating Agreement and the Registration Rights Agreement, and (h) other customary conditions specified in the Merger Agreement.

As of the date hereof, Calyxt and Cibus do not anticipate that the Transactions will require a merger control filing under the Hart-Scott Rodino Antitrust Improvements Act of 1976 (the "HSR Act"), however, if the parties determine prior to Closing that such a filing is required by law, the Closing will be subject to the expiration or termination of the waiting period under the HSR Act and receipt of any other specified merger control consents or clearances.

Pursuant to the terms of the Merger Agreement, each of Calyxt, Cibus and the Blockers are required to use reasonable best efforts to consummate the Mergers, including with respect to satisfaction of the relevant closing conditions. Prior to obtaining the Calyxt Stockholder Approval, Calyxt's board of directors (the "Calyxt Board") may, in certain limited circumstances, withhold, amend, qualify, withdraw or modify its recommendation that Calyxt's stockholders adopt the Merger Agreement in the event of unforeseen changes in circumstances (as further described in the Merger Agreement) or recommend or otherwise declare advisable a Superior Offer (as defined in the Merger Agreement) (a "Calyxt Board Recommendation Change"), subject to complying with notice and other specified conditions, including giving Cibus the opportunity to propose revisions to the terms of the transaction contemplated by the Merger Agreement during a matching right period. Notwithstanding a Calyxt Board Recommendation Change, unless Cibus or Calyxt terminates the Merger Agreement to consummate a Superior Offer and Calyxt pays Cibus the break fee discussed below, Calyxt will continue to be required to convene the meeting of its stockholders.

Termination Rights

The Merger Agreement may be terminated by either party (a) by mutual written consent, (b) if a governmental entity issues a non-appealable final order, decree or ruling or taken any other non-appealable final action, in each case having the effect of permanently restraining, enjoining or otherwise prohibiting any of the Mergers (c) if the Mergers are not consummated by May 15, 2023 (which date may be extended, at the election of Cibus, to June 30, 2023, provided that Cibus also increases the maximum Interim Funding to \$4,000,000) (as may be so extended, the "*Outside Date*") or (d) if Calyxt has held the meeting of the Calyxt stockholders to approve the Calyxt Stockholder Matters and failed to receive the Calyxt Stockholder Approval, in each case subject to customary carve outs and exceptions.

The Merger Agreement may be terminated by either party if there has been an uncured breach or failure to perform any representation, warranty, covenant or agreement on the part of the other party that would prevent the closing conditions from being satisfied, in each case, subject to customary carve outs and exceptions (an "Uncured Breach Event").

The Merger Agreement may also be terminated by Cibus if (a) the Calyxt Board effects a Calyxt Board Recommendation Change, or (b) if there is a material breach of Calyxt's obligations under the No-Shop Provision, in each case subject to customary carve outs and exceptions. If such a termination occurs, Calyxt shall be obligated to pay Cibus a fee equal to \$1,000,000 within three business days of such termination.

The Merger Agreement may also be terminated by Calyxt any time prior to the Calyxt Stockholder Meeting (as defined in the Merger Agreement) if (i) Calyxt receives a Superior Offer, (ii) Calyxt complies with its obligations under the Merger Agreement with respect to the Superior Offer, including making a Calyxt Board Recommended Change, and (iii) the Calyxt Board approves the Superior Offer and enters into a definitive agreement with respect thereto. If such termination occurs, Calyxt shall be obligated to pay Cibus a fee equal to \$1,000,000 within three business days of such termination.

If the Merger Agreement is terminated (i) by either party after the Outside Date, (ii) by Cibus due to an Uncured Breach Event by Calyxt, or (iii) by either party if Calyxt fails to obtain the Calyxt Stockholder Approval, and, in each case, an Acquisition Proposal (as defined in the Merger Agreement) has been publicly announced or disclosed to Calyxt or the Calyxt Board after the date of the Merger Agreement (but prior to such termination) and Calyxt enters into a definitive agreement with respect to or consummates an Acquisition Proposal within 12 months after the date of such determination, then Calyxt shall be obligated to pay Cibus a fee equal to \$1,000,000 within three business days of such termination.

The Merger Agreement may be terminated by Calyxt if (a) Cibus does not deliver written consents from Cibus and the Blockers approving the Merger Agreement and the applicable Mergers prior to 5:00 p.m., New York time, on the date that is 10 business days following the date the Form S-4 is made effective by the SEC or (b) there is a material breach of Cibus's obligations under the No-Shop Provision. If such a termination occurs, Cibus shall be obligated to pay Calyxt a fee equal to \$1,000,000 plus up to \$500,000 of Calyxt's reasonable and documented out-of-pocket transaction expenses.

The foregoing summary of the Merger Agreement is not complete and is qualified in its entirety by the full text of the Merger Agreement, which is attached hereto as Exhibit 2.1 and is incorporated by reference herein.

The Merger Agreement has been included to provide investors with information regarding its terms. It is not intended to provide any other factual information about Calyxt, Cibus, the Blockers or their respective subsidiaries or affiliates. The representations, warranties and covenants contained in the Merger Agreement were made only for purposes of the Merger Agreement and as of specific dates, were solely for the benefit of the parties to the Merger Agreement, may be subject to limitations agreed upon by the contracting parties, including being qualified by confidential disclosures made for the purposes of allocating contractual risk between the parties to the Merger Agreement instead of establishing these matters as facts, and may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. Accordingly, the representations, warranties and covenants, and any descriptions thereof, may not be accurate characterizations of the actual state of facts or condition of the parties to the Merger Agreement or any of their respective subsidiaries or affiliates. Moreover, information concerning the subject matter of representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in Calyxt's, Cibus's or any Blockers', or their respective affiliates', public disclosures.

Exchange Agreement

At the Closing, Resulting Public Company, Cibus, and the Electing Members will enter into an exchange agreement (the "*Exchange Agreement*"), in a form to be agreed by Cibus and Calyxt. Pursuant to the Exchange Agreement, the Electing Members shall generally have the right (on the terms and subject to the conditions to be agreed therein) to exchange (each, an "*Exchange*") their Up-C Units for shares of Class A Common Stock on a one-for-one basis.

Tax Receivable Agreement

At the Closing, Resulting Public Company, Cibus, and the Electing Members will also enter into a tax receivable agreement (the "Tax Receivable Agreement"), in a form to be agreed by Cibus and Calyxt. Pursuant to the Tax Receivable Agreement, Resulting Public Company will agree to make payments to the Electing Members from time to time with respect to the effect of tax attributes resulting from Exchanges and the Blocker Mergers on the liability for taxes of the Resulting Public Company. The Tax Receivable Agreement shall contain such usual and customary terms including that such payments will be in amounts corresponding to 85% of the deemed tax savings of Resulting Public Company resulting from the Exchanges and the Blocker Mergers.

Registration Rights Agreement

At the Closing, Resulting Public Company and the Electing Members will enter into a registration rights agreement (the "*Registration Rights Agreement*"), in a form to be agreed by Calyxt and such Electing Members. Pursuant to the Registration Rights Agreement, Resulting Public Company will agree to register for resale under the Securities Act, the shares of Class A Common Stock received by Electing Members who, in the future, elect to Exchange their Up-C Units for shares of Class A Common Stock pursuant to the Exchange Agreement.

Transaction Support Agreements

Calyxt Support Agreements

On January 13, 2023, concurrently with the execution of the Merger Agreement and as a condition to Cibus entering into the Merger Agreement, Cibus entered into Voting Agreements (the "Calyxt Support Agreements"), with certain of Calyxt's stockholders, directors and officers (the "Supporting Stockholders") who as of January 13, 2023, collectively and in the aggregate, beneficially hold voting power of approximately 49.8% of Calyxt's outstanding common stock, par value \$0.0001 per share (the "Subject Shares"). Pursuant to the terms of the Calyxt Support Agreements, the Supporting Stockholders have agreed to take certain actions to support the Transactions (including not transferring the Subject Shares during the term of the Calyxt Support Agreement, except under specified circumstances) and voting the Subject Shares in favor of the Merger Agreement and the Transactions and against any alternative acquisition proposals.

Notwithstanding the voting obligations in the Calyxt Support Agreements, if the Calyxt Board effects a Public Company Recommendation Change, then the Calyxt Support Agreements will terminate and Supporting Stockholders will have no obligations in respect of how to vote their respective Subject Shares. The Calyxt Support Agreements also contain customary termination provisions, including termination with respect to a given Supporting Stockholder in the event there is a modification or amendment to the Merger Agreement in a manner that reduces the amount or changes the form of Merger Consideration payable to such Supporting Stockholder or otherwise adversely effects the Supporting Stockholder in any material respect without his, her or its prior written approval.

The foregoing summary of the Calyxt Support Agreements is not complete and is qualified in its entirety by the full text of the form of Calyxt Support Agreement, which is attached hereto as Exhibit 99.1 and is incorporated by reference herein.

Cibus Support Agreements

On January 13, 2023, concurrently with the execution of the Merger Agreement and as a condition to Calyxt entering into the Merger Agreement, Calyxt entered into Voting Agreements (the "Cibus Support Agreements"), with certain of Cibus's equity holders (including the Blockers), directors and officers (the "Cibus Supporting Unitholders") who as of January 13, 2023, collectively and in the aggregate, hold voting power of approximately 54.3% of the outstanding voting Cibus membership units (the "Subject Units"). Pursuant to the terms of the Cibus Support Agreements, the Cibus Supporting Unitholders have agreed to take certain actions to support the Transactions (including not transferring the Subject Units during the term of the Cibus Support Agreement except under specified circumstances) and voting or acting by written consent with respect to all of its Subject Units in favor of the Mergers and the other Transactions, against any alternative acquisition proposals, to agree to convert all outstanding pre-Closing Cibus preferred units into Cibus voting common units on a one-for-one basis.

The Cibus Support Agreements also contain customary termination provisions, including termination with respect to a given Supporting Stockholder in the event there is a modification or amendment to the Merger Agreement in a manner that reduces the amount or changes the form of Merger Consideration payable to such Supporting Stockholder or otherwise adversely effects the Supporting Stockholder in any material respect without his, her or its prior written approval.

The foregoing summary of the Cibus Support Agreements is not complete and is qualified in its entirety by the full text of the form of Cibus Support Agreement, which is attached hereto as Exhibit 99.2 and is incorporated by reference herein.

Item 2.03. Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information in Item 1.01 under the caption "Interim Funding" is incorporated herein by reference.

The description of the terms and conditions of the Interim Funding does not purport to be complete and is qualified by reference in its entirety by reference to the Terms and Conditions of Interim Funding, a copy of which is attached to this report as Exhibit 10.3 and is incorporated herein by reference.

Item 5.02 Departure of Directors or Certain Executive Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Calyxt previously adopted the form of the 2021 Executive Severance Plan (the "Executive Severance Plan") for certain key management employees who agree to participate (the "plan participants"). The Executive Severance Plan provides plan participants with certain severance benefits upon termination of the plan participant's employee with Calyxt. On January 13, 2023, Calyxt adopted an amendment to the Executive Severance Plan (the "Plan Amendment"). The Plan Amendment amends the time period for payment of Severance Benefits under the Executive Severance Plan in the event of a change of control. Prior to the Plan Amendment, in the event of a change in control, payments were to be made within 60 days of the plan participant's termination of employment. Under the Plan Amendment, payments are to be in substantially equal installments over the severance coverage period, subject to any portions that are not exempt from Section 409A of the tax code being paid within 60 days of the termination date. For example, if a plan participant is entitled to six months of severance benefit, the payments would be paid over six months.

The foregoing description of the Amendment to the Executive Severance Plan does not purport to be complete and is qualified by reference in its entirety by reference to the Amendment to the Executive Severance Plan, a copy of which is attached to this report as Exhibit 10.4 and is incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On January 17, 2023, Calyxt and Cibus issued a joint press release announcing the execution of the Merger Agreement. The press release is furnished as Exhibit 99.1.

The information furnished in this Item 7.01 and Exhibit 99.1 attached hereto shall not be deemed to be "filed" for the purposes of Section 18 of the Securities and Exchange Act of 1934 (the "Exchange Act"), or otherwise subject to the liabilities of such section, nor shall such information be deemed incorporated by reference into any filing under the Securities Act of 1933 (the "Securities Act") or the Exchange Act, except as shall be expressly set forth by specific reference in such a filing.

Item 8.01 Other Events.

In light of the ongoing discussions with Cibus, Calyxt has focused its current business activities on ensuring it has cash sufficient to achieve a closing of the Transactions. Accordingly, Calyxt has taken additional steps to reduce its operating expenses. Calyxt has focused its operations on: scaling production of its Plant Cell Matrix™ platform with its manufacturing partner, Evologic Technologies GmbH; licensing efforts with respect to its PlantSpring™ technology and plant traits, including the TALEN™ technology; and continuing to progress its current customer projects. Calyxt's three key customer projects are (1) its research collaboration with a leading global food ingredient manufacturer to develop a soybean trait to serve as an alternative to palm oil, (2) its plant-based chemistry pilot project for a major consumer packaged goods company, and (3) supporting late-stage development activities for Calyxt's improved digestibility alfalfa trait, which was developed with and licensed to S&W Seed Company.

As a result of this narrowed focus and in light of the available Interim Funding from Cibus in connection with the Transactions, Calyxt now believes it has sufficient cash to fund operations through the end of the second quarter of 2023.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits

Exhibit <u>Number</u>	<u>Description</u>
2.1*	Agreement and Plan of Merger, dated January 13, 2023, by and among Calyxt, Inc., Calypso Merger Subsidiary, LLC, Cibus Global, LLC and the other parties thereto
10.1*	Form of Calyxt Support Agreement (included as Exhibit A in Exhibit 2.1)
10.2*	Form of Cibus Support Agreement (included as Exhibit B in Exhibit 2.1)
10.3	Terms and Conditions of Interim Funding (included as Exhibit E in Exhibit 2.1)
10.4	Amendment to Calyxt's 2021 Executive Severance Plan
99.1	Press Release, dated January 17, 2023, jointly issued by Calyxt, Inc. and Cibus Global, LLC
99.2	Social Media Posts posted by Cibus Global, LLC on January 17, 2023
99.3	Social Media Posts posted by Calyxt, Inc. on January 17, 2023
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

^{*} Certain exhibits and/or schedules (and similar attachments) have been omitted pursuant to the provisions of Regulation S-K, Item 601(a)(5). The registrant hereby undertakes to furnish supplementally to the Securities and Exchange Commission (SEC) upon request by the SEC copies of any of the omitted exhibits and schedules (or similar attachments).

Cautionary Statement Regarding Forward-Looking Statements

The information included in this Current Report on Form 8-K includes "forward-looking statements" within the meaning of Section 27A of the Securities Act and Section 21E of the Exchange Act. All statements, other than statements of present or historical fact included herein, regarding the transactions, the ability of the parties to the merger agreement to consummate the transactions, the benefits of the transactions, Calyxt's future financial performance (including its liquidity and capital resources and cash runway), the combined company's future performance following the transactions, and the potential for global regulatory developments, as well as Calyxt's, Cibus's and the combined company's respective strategies, future operations, financial positions, prospects and plans as well as the objectives of management are forward-looking statements. Words such as "expects," "continues," "may," "will," "approximately," "intends," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words.

These forward-looking statements are based on the current expectations and assumptions of Cibus's and Calyxt's management about future events and are based on currently available information as to the outcome and timing of future events. Forward-looking statements are subject to risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of Calyxt and Cibus. These risks include, but are not limited to, (i) the risk that the conditions to the closing of the proposed Transactions are not satisfied, including the failure to obtain stockholder approval of matters related to the proposed Transactions in a timely manner or at all, (ii) uncertainties as

to the timing of the consummation of the proposed Transactions, (iii) risks related to Calyxt's capital resources and the ability of Calyxt and Cibus, respectively, to correctly estimate and manage their respective operating expenses and expenses associated with the proposed Transactions, (iv) risks related to Calyxt's continued listing on the Nasdaq Capital Market until closing of the proposed Transactions, (v) risks associated with the possible failure to realize certain anticipated benefits of the proposed Transactions, including with respect to future financial and operating results; (vi) uncertainties regarding the impact that any delay in the Closing would have on the anticipated cash resources of the combined company upon Closing and other events and unanticipated spending and costs that could reduce the combined company's cash resources; (vii) the potential for the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the Merger Agreement; (viii) the possible effect of the announcement, pendency or completion of the Merger on Calyxt's or Cibus's business relationships, operating results and business generally; (ix) risks related to unexpected costs related to the Merger; (x) the potential for, and uncertainty associated with the outcome of, any legal proceedings that may be instituted against Calyxt or Cibus or any of their respective directors or officers related to the Merger Agreement or the Transactions; (xi) risks associated with the ability of Calyxt and Cibus to protect their respective intellectual property rights; (xii) the potential impact of competitive responses to the proposed Transactions and changes in expected or existing competition; (xiii) the possibility that Calyxt, Cibus or the combined company may be adversely affected by other economic, business, or competitive factors; (xiv) risks associated with the loss of key employees of Calvxt or Cibus; (xy) risks associated with changes in applicable laws or regulations and the potential impact of such changes on Calvxt's, Cibus's or the combined company's ability to advance product development and commercialization; and (xvi) other risks and uncertainties identified from time to time in documents filed or to be filed with the SEC by Calyxt or the combined company, including those discussed in the "Risk Factors" section of Calyxt's Annual Report on Form 10-K, which was filed with the SEC on March 3, 2022, and Quarterly Reports on Form 10-Q, which were filed with the SEC on August 4, 2022 and November 3, 2022, respectively. Should one or more of the risks or uncertainties occur, or should underlying assumptions prove incorrect, actual results and plans could differ materially from those expressed in any forward-looking statements. In addition, the forward-looking statements included in this Current Report on Form 8-K represent Calyxt's and Cibus's views as of the date hereof. Calyxt and Cibus anticipate that subsequent events and developments will cause the respective company's views to change. Calyxt and Cibus specifically disclaim any obligation to update such forward-looking statements in the future, except as required under applicable law. These forward-looking statements should not be relied upon as representing Calyxt's or Cibus's views as of any date subsequent to the date hereof.

Important Additional Information

In connection with the proposed transactions, Calyxt intends to file materials with the SEC, including the Form S-4, which will include a proxy statement of Calyxt for the stockholders of Calyxt and that will serve as a prospectus of Calyxt and an information statement of Cibus, and other documents relating to the proposed transactions. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THESE MATERIALS, INCLUDING THE FORM S-4 AND THE PROXY STATEMENT/PROSPECTUS INCLUDED THEREIN, WHEN THEY BECOME AVAILABLE, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT CALYXT, CIBUS AND THE PROPOSED TRANSACTIONS. After the Form S-4 is declared effective, the definitive proxy statement to be included in the Form S-4 will be mailed to Calyxt stockholders as of a record date to be established for voting on the Calyxt Stockholder Matters to be considered at the Calyxt Stockholder Meeting. The Form S-4, the proxy statement/prospectus included therein, and other materials filed by Calyxt with the SEC may be obtained free of charge from the SEC's website (www.sec.gov) or from Calyxt by directing a request to: Calyxt, Inc., 2800 Mount Ridge Road, Roseville, MN 55113.

About Cibus

Cibus is a leading agricultural technology company that uses precision gene editing to breed plants with new genetic traits that make them healthier and better able to thrive in their changing environments. Cibus's primary focus is productivity traits that improve farming productivity by increasing crop yields and reducing inputs such as fungicides, herbicides, pesticides, and fertilizers.

Participants in the Solicitation

Calyxt, Cibus and their respective directors, executive officers and other members of management may be deemed to be participants in the solicitation of proxies with respect to the proposed Transactions under the rules of the SEC. Information about the directors and executive officers of Calyxt is set forth in Calyxt's Annual Report on Form 10-K for the fiscal year ended December 31, 2021, which was filed with the SEC on March 3, 2022, and its definitive proxy statement for its 2022 Annual Meeting of Shareholders, which was filed with the SEC on April 19, 2022.

Other information regarding persons who may, under the rules of the SEC, be deemed to be participants in the proxy solicitation and a description of their interests in the transaction, by security holdings or otherwise, will be included in the proxy statement/prospectus included in the Form S-4 and other relevant materials to be filed with the SEC regarding the proposed transaction when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from Calyxt or the SEC's website, as indicated above.

No Offer or Solicitation

This press release shall not constitute an offer to sell or a solicitation of an offer to buy the securities of Calyxt, Cibus or the combined company, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offering of securities shall be made in the United States except by means of a prospectus meeting the requirements of Section 10 of the Securities Act. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, a public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone or internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, Calyxt, Inc. has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: January 17, 2023

CALYXT, INC.

By: /s/ Michael A. Carr

Name: Michael A. Carr

Title: President and Chief Executive Officer

EXECUTION

AGREEMENT AND PLAN OF MERGER

by and among

CALYXT, INC.,

CALYPSO MERGER SUBSIDIARY, LLC,

CIBUS GLOBAL, LLC,

NEW VENTURES I HOLDINGS, LLC,

BCGF CB HOLDINGS LLC,

BCGFCP CB HOLDINGS LLC,

BCGFK CB HOLDINGS LLC,

FSBCGF CB HOLDINGS LLC,

PYLBCG CB HOLDINGS LLC,

FSGRWCO CB HOLDINGS LLC,

GROWTHCO CB HOLDINGS LLC,

GRTHCOCP CB HOLDINGS LLC

and

GRWTHCOK CB HOLDINGS LLC

Dated as of January 13, 2023

		Page
ARTICL	E I THE MERGERS	4
1.1	Mergers; Effective Time of the Mergers	4
1.2	Closing; Actions at the Closing	5
1.3	Effects of the Mergers	6
1.4	Directors and Officers	7
1.5	Additional Action	7
ARTICL	E II CONVERSION OF SECURITIES	7
2.1	Conversion of Securities	7
2.2	Payment Fund	10
2.3	Allocation Schedule	11
2.4	Withholding Rights	12
2.5	Merger Consideration Election Procedures	13
ARTICL	E III REPRESENTATIONS AND WARRANTIES OF MERGER PARTNER AND BLOCKERS	14
3.1	Organization, Standing and Power	14
3.2	Capitalization	15
3.3	Subsidiaries	16
3.4	Authority; No Conflict; Required Filings and Consents	16
3.5	Financial Statements; Information Provided	17
3.6	Absence of Certain Changes	19
3.7	Merger Partner Tax Matters	19
3.8	Assets	20
3.9	Owned and Leased Real Property	21
3.10	Intellectual Property	22
3.11	Contracts	23
3.12	Litigation	25
3.13	Environmental Matters	26
3.14	Labor and Employment	26
3.15	Employee Benefit Plans	28

29

3.16 Compliance with Laws

(continued)

		Page
3.17	Unlawful Payments	29
3.18	Permits	30
3.19	Insurance	30
3.20	Cybersecurity and Privacy	31
3.21	Brokers	32
3.22	Blocker Organization, Standing and Power	32
3.23	Blocker Capitalization	32
3.24	No Business	33
3.25	Blocker Authority; No Conflict; Required Filings and Consents	33
3.26	Blocker Compliance with Laws	34
3.27	Blocker Brokers	34
3.28	Blocker Tax Matters	34
3.29	No Other Representations and Warranties	35
3.30	Reliance	35
RTICLI	E IV REPRESENTATIONS AND WARRANTIES OF PUBLIC COMPANY AND MERGER SUBSIDIARY	36
4.1	Organization, Standing and Power	36
4.2	Capitalization	37
4.3	Subsidiaries	38
4.4	Authority; No Conflict; Required Filings and Consents	39
4.5	Operations of Merger Subsidiary	40
4.6	SEC Filings; Financial Statements	41
4.7	Absence of Certain Changes	43
4.8	Tax Matters	43
4.9	Assets	45
4.10	Owned and Leased Real Property	45
4.11	Intellectual Property	46
4.12	Contracts	47
4.13	Litigation	50
4.17	Compliance with Laws	54

(continued)

			Page
	4.18	Unlawful Payments	54
	4.19	Permits	54
	4.20	Insurance	55
	4.21	Cybersecurity and Privacy	55
	4.22	Brokers	56
	4.23	Transactions with Affiliates	56
	4.24	[Reserved]	56
	4.25	Nasdaq Market Quotation	56
	4.26	No Shareholder Rights Plan; No Antitakeover Law	57
	4.27	Valid Issuance	57
	4.28	No Other Representations and Warranties	57
	4.29	Reliance	57
AR	TICLE	V PRE-CLOSING COVENANTS	57
	5.1	Operation of Merger Partner's Business	57
	5.2	Operation of Public Company's Business	60
	5.3	Operation of Blockers	62
	5.4	Public Company No Solicitation	62
	5.5	Merger Partner Non-Solicitation	64
	5.6	Notification of Certain Matters	65
	5.7	Proxy Statement/Prospectus; Registration Statement	66
	5.8	Nasdaq Listing	67
	5.9	Public Company Stockholders' Meeting	68
	5.10	Merger Partner and Blocker Member Approval	71
	5.11	Access to Information	72
	5.12	Closing Efforts	72
	5.13	Public Disclosure	72
	5.14	Antitrust	73
	5.15	Warrant Restructuring	73
	5.16	Transaction Documentation	73
	5.17	Merger Partner Funding	73
	5.18	Section 16 Matters	73

(continued)

		Page
ARTICI	LE VI CONDITIONS TO CONSUMMATION OF THE MERGERS	74
6.1	Conditions to Obligations of Public Company and Merger Subsidiary	74
6.2	Conditions to Obligations of Merger Partner and the Blockers	75
ARTICI	LE VII ADDITIONAL AGREEMENTS	77
7.1	Employee Matters	77
7.2	D&O Indemnification	78
7.3	Tax Matters	80
7.4	Transaction Litigation	82
7.5	Asset Transfers	82
ARTICI	LE VIII TERMINATION AND AMENDMENT	82
8.1	Termination	82
8.2	Effect of Termination	84
8.3	Fees and Expenses	85
8.4	Amendment	86
8.5	Extension; Waiver	86
ARTICI	LE IX DEFINITIONS	87
ARTICI	LE X MISCELLANEOUS	103
10.1	1 Notices	103
10.2	2 Entire Agreement	104
10.3	3 Third-Party Beneficiaries	104
10.4	4 Assignment	104
10.5	5 Severability	104
10.6	6 Counterparts and Signature	104
10.7	7 Interpretation	104
10.8	8 Governing Law	105
10.9	9 Remedies	105
10.1	10 Confidentiality	106
10.1	11 Submission to Jurisdiction	106
10.1	12 Non-Survival of Representations, Warranties and Covenants	106

(continued)

Page

EXHIBITS

Exhibit A Public Company Support Agreements
Exhibit B Merger Partner Support Agreement
Exhibit C-1 First Blocker Merger Certificate of Merger
Exhibit C-2 Form of Blocker Certificate of Merger
Exhibit C-3 Merger Partner Certificate of Merger

Exhibit D-1 Public Company Directors
Exhibit D-2 Public Company Officers
Exhibit E Interim Funding Term Sheet
Exhibit F Amendment to License Agreement

Exhibit G Allocation Schedule

Exhibit H Term Sheet for Tax Receivables Agreement

Exhibit I Form of Unitholder Representative Engagement Agreement

AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") is entered into as of January 13, 2023, by and among: Calyxt, Inc., a Delaware corporation ("Public Company"); Calypso Merger Subsidiary, LLC, a Delaware limited liability company and wholly-owned subsidiary of Public Company ("Merger Subsidiary"); Cibus Global, LLC, a Delaware limited liability company ("Merger Partner"); New Ventures I Holdings, LLC, a Delaware limited liability company ("Blocker 1"); BCGF CB Holdings LLC, a Delaware limited liability company ("Blocker 2"); BCGFCP CB Holdings LLC, a Delaware limited liability company ("Blocker 3"); PYLBCG CB Holdings LLC, a Delaware limited liability company ("Blocker 5"); PYLBCG CB Holdings LLC, a Delaware limited liability company ("Blocker 6"); FSGRWCO CB Holdings LLC, a Delaware limited liability company ("Blocker 7"); GROWTHCO CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); GRTHCOCP CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company ("Blocker 9"); and GRWTHCOK CB Holdings LLC, a Delaware limited liability company

RECITALS

WHEREAS, Merger Subsidiary is a newly-formed, wholly-owned Subsidiary of Public Company, and was formed for the sole purpose of the Merger Partner Merger;

WHEREAS, (a) Public Company and Blocker 1 intend to effect a merger (the "First Blocker Merger") in which Blocker 1 will merge with and into Public Company with Public Company being the surviving entity in such merger; (b) Public Company and Blocker 2 intend to effect a merger (the "Second Blocker Merger") in which Blocker 2 will merge with and into Public Company with Public Company being the surviving entity in such merger; (c) Public Company and Blocker 3 intend to effect a merger (the "Third Blocker Merger") in which Blocker 3 will merge with and into Public Company with Public Company being the surviving entity in such merger; (d) Public Company and Blocker 4 intend to effect a merger (the "Fourth Blocker Merger") in which Blocker 4 will merge with and into Public Company with Public Company being the surviving entity in such merger; (e) Public Company and Blocker 5 intend to effect a merger (the "Fifth Blocker Merger") in which Blocker 5 will merge with and into Public Company with Public Company being the surviving entity in such merger; (f) Public Company and Blocker 6 intend to effect a merger (the "Sixth Blocker Merger") in which Blocker 6 will merge with and into Public Company with Public Company being the surviving entity in such merger; (g) Public Company and Blocker 7 intend to effect a merger (the "Seventh Blocker Merger") in which Blocker 7 will merge with and into Public Company with Public Company being the surviving entity in such merger; (h) Public Company and Blocker 8 intend to effect a merger (the "Eighth Blocker Merger") in which Blocker 8 will merge with and into Public Company with Public Company being the surviving entity in such merger; (i) Public Company and Blocker 9 intend to effect a merger (the "Ninth Blocker Merger") in which Blocker 9 will merge with and into Public Company with Public Company being the surviving entity in such merger; and (j) Public Company and Blocker 10 intend to effect a merger (the "Tenth Blocker Merger" and, collectively with the First Blocker Merger, Second Blocker Merger, Third Blocker Merger, Fourth Blocker Merger, Fifth Blocker Merger, Sixth Blocker Merger, Seventh Blocker Merger, Eighth Blocker Merger and Ninth Blocker Merger, the "Blocker Mergers") in which Blocker 10 will merge with and into Public Company with Public Company being the surviving entity in such merger;

WHEREAS, following the Blocker Mergers, Public Company, Merger Subsidiary and Merger Partner intend to effect a merger (the "Merger Partner Merger" and, collectively with the Blocker Mergers, the "Mergers"), pursuant to which Merger Subsidiary will merge with and into Merger Partner, Merger Subsidiary will cease to exist, and Merger Partner will survive, with the former security holders of Merger Partner receiving the Merger Consideration set forth in Section 2.1(b);

WHEREAS, in connection with the consummation of the Merger Partner Merger, Public Company, the Electing Members and Merger Partner will enter into a Tax Receivable Agreement, in a form to be agreed by Merger Partner and Public Company consistent with the terms set forth on Exhibit H promptly following the date of this Agreement, pursuant to which the Public Company will agree to make payments to the post-Closing holders of Merger Partner Common Units (other than the Public Company) from time to time with respect to the effect of tax attributes resulting from Exchanges and the Blocker Mergers on the liability for Taxes of the Public Company (the "Tax Receivable Agreement");

WHEREAS, in connection with the consummation of the Merger Partner Merger, Public Company, the Electing Members and Merger Partner will enter into an Exchange Agreement, in a form to be agreed by Merger Partner and Public Company promptly following the date of this Agreement, pursuant to which the post-Closing holders of Merger Partner Common Units (other than the Public Company) shall have the right (on the terms and subject to the conditions to be agreed therein) to exchange their Merger Partner Common Units and shares of Public Company Class B Common Stock for shares of Public Company Class A Common Stock (the "Exchange Agreement", and any such exchanges, "Exchanges");

WHEREAS, in connection with the consummation of the Merger Partner Merger, Public Company and the Electing Members will enter into a Registration Rights Agreement, in a form to be agreed by Merger Partner and Public Company promptly following the date of this Agreement, providing for, among other things, the registration for resale of any shares Public Company Class A Common Stock received by post-Closing holders of Merger Partner Common Units (other than the Public Company) in connection with any Exchanges (the "Registration Rights Agreement");

WHEREAS, each recipient of shares of Public Company Class B Common Stock shall also be a Participating Member in respect of the shares of Public Company Class A Stock such Person shall receive in the Merger Partner Merger, as contemplated herein;

WHEREAS, the Parties intend for the Transactions to be treated in accordance with the Intended Tax Treatment set forth in Section 7.3(b);

WHEREAS, immediately prior to the Mergers, Public Company intends to repay the outstanding balance owed to Merger Partner in connection with the Interim Funding through the transfer of a portion of its assets to Merger Partner and immediately thereafter, Public Company intends to contribute the residual of its assets and liabilities to Merger Partner, as a contribution to the capital of Merger Partner in exchange for Merger Partner Common Units, in each case, pursuant to Section 7.5 (the "Asset Transfers");

WHEREAS, the Boards of Directors (or equivalent governing bodies) of Public Company, Merger Subsidiary and Merger Partner deem it advisable and in the best interests of each company and their respective equity holders to consummate the Mergers and the other Transactions in accordance with and on the terms contemplated by this Agreement;

WHEREAS, the managing member of Blocker 1 executed and delivered to Public Company on or prior to the date hereof, a written consent wherein the managing member (i) determined that the Transactions are fair to, advisable, and in the best interests of Blocker 1 and its sole member and (ii) adopted this Agreement and approved the Transactions, including the First Blocker Merger, in each case, consistent with the applicable provisions of Blocker 1's Organizational Documents and Delaware Law;

WHEREAS, concurrently with the execution and delivery of this Agreement and as a condition and inducement to Merger Partner's willingness to enter into this Agreement, the Persons set forth on Schedule A attached hereto (the "Public Company Support Stockholders") have entered into Voting Agreements, dated as of the date of this Agreement, in the form attached hereto as Exhibit A (the "Public Company Support Agreements"), pursuant to which the Public Company Support Stockholders have, subject to the terms and conditions set forth therein, agreed to vote all of their respective shares of capital stock of Public Company in favor of Public Company Stockholder Matters and the Other Public Company Stockholder Matters at the Public Company Stockholders' Meeting to be convened to approve the Mergers and the other Transactions; and

WHEREAS, concurrently with the execution and delivery of this Agreement and as a condition and inducement to Public Company's willingness to enter into this Agreement, the Persons set forth on Schedule B attached hereto (the "Merger Partner Support Holders") have each entered into a Voting Agreement, dated as of the date of this Agreement, in the form attached hereto as Exhibit B (the "Merger Partner Support Agreement"), pursuant to which the Merger Partner Support Holders have, subject to the terms and conditions set forth therein, agreed to vote all of their respective shares of membership interest in Merger Partner in favor of the Merger Partner Member Approvals by executing and delivering an action by written consent, representing the Merger Partner Member Approvals (together with the actions by written consent representing the Blocker Approvals, the "Written Consents").

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth below, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Public Company, Merger Subsidiary, Merger Partner and each Blocker agree as follows:

ARTICLE I THE MERGERS

1.1 Mergers; Effective Time of the Mergers.

(a) Blocker Mergers.

- (i) Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the applicable provisions of Delaware Law, Public Company and each Blocker shall cause the corresponding Blocker Merger to be consummated (and for the avoidance of doubt, each Blocker shall only be responsible for causing the merger of itself with and into Public Company pursuant to its corresponding Blocker Merger). Each Blocker Merger shall be consummated at the applicable Blocker Merger Effective Time in accordance with this Agreement upon the filing and effectiveness of a certificate of merger relating to the corresponding Blocker Merger in substantially the form of Exhibit C-1 (with respect to the First Blocker Merger) or Exhibit C-2 (with respect to all subsequent Blocker Mergers other than the First Blocker Merger) (the applicable certificate of merger corresponding to each Blocker Merger being such Blocker Merger's applicable "Blocker Certificate of Merger").
- (ii) Upon the applicable Blocker Merger Effective Time, the separate company existence of each Blocker being merged shall cease and Public Company, as the surviving corporation of such Blocker Merger, shall continue its corporate existence under Delaware Law.
- (iii) Subject to the provisions of this Agreement, Public Company and each Blocker shall cause the corresponding Blocker Certificate of Merger to be executed, acknowledged and filed with the Secretary of State of the State of Delaware (and for the avoidance of doubt, each Blocker shall only be responsible for causing its applicable Blocker Certificate of Merger to be executed, acknowledged and filed). The applicable Blocker Merger shall become effective at the time when the corresponding Blocker Certificate of Merger has been duly filed with the Secretary of State of the State of Delaware or at such later time as may be agreed by Public Company, Merger Partner and the applicable Blocker in writing and specified in the corresponding Blocker Certificate of Merger (the effective time of each Blocker Merger being such Blocker Merger's applicable "Blocker Merger Effective Time").

(b) Merger Partner Merger.

- (i) Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the applicable provisions of Delaware Law, Public Company, Merger Subsidiary and Merger Partner shall cause the Merger Partner Merger to be consummated. The Merger Partner Merger shall be consummated at the Merger Partner Merger Effective Time immediately following the Blocker Mergers in accordance with this Agreement upon the filing and effectiveness of a certificate of merger relating to the Merger Partner Merger in substantially the form of Exhibit C-3 (the "Merger Partner Certificate of Merger").
- (ii) Upon the Merger Partner Merger Effective Time, the separate company existence of Merger Subsidiary shall cease and Merger Partner, as the surviving limited liability company in the Merger Partner Merger, shall continue its existence as a limited liability company under Delaware Law.

(iii) Subject to the provisions of this Agreement, Public Company, Merger Subsidiary and Merger Partner and shall cause the Merger Partner Certificate of Merger to be executed, acknowledged and filed with the Secretary of State of the State of Delaware. The Merger Partner Merger shall become effective at the time when the Merger Partner Certificate of Merger has been duly filed with the Secretary of State of the State of Delaware or at such later time as may be agreed by Public Company and Merger Partner in writing and specified in the Merger Partner Certificate of Merger (the "Merger Partner Merger Effective Time").

1.2 Closing; Actions at the Closing.

(a) The Closing shall take place at 10:00 a.m., Eastern time, on the Closing Date remotely by electronic exchange of documents and/or at the offices of Jones Day, 250 Vesey Street, New York, NY 10281, unless another date, place or time is agreed to in writing by Public Company and Merger Partner.

(b) At the Closing:

- (i) Merger Partner and each Blocker shall deliver to Public Company and Merger Subsidiary the various certificates, instruments and documents referred to in Section 6.1 that are required to be delivered by it;
- (ii) Public Company and Merger Subsidiary shall deliver to Merger Partner the various certificates, instruments and documents referred to in Section 6.2;
- (iii) Public Company and Merger Partner shall file with the Secretary of State of the State of Delaware the Blocker Certificate of Merger for each of the Blocker Mergers and the Merger Partner Certificate of Merger, respectively;
- (iv) Public Company, Merger Subsidiary, Merger Partner and each Blocker shall take, or cause to be taken, the actions set forth in Sections 1.1(a) and 1.1(b) that are required to be taken by it;
- (v) Public Company shall issue the shares of Public Company Class A Common Stock and Public Company Class B Common Stock constituting the Merger Consideration and deposit them with the Exchange and Paying Agent pursuant to Section 2.2(a);
- (vi) Merger Partner shall issue the Merger Partner Common Units constituting the Merger Consideration and deposit them with the Exchange and Paying Agent pursuant to Section 2.2(a);
- (vii) Merger Partner shall issue the Merger Partner Common Units to be issued to Public Company pursuant to Section 2.1(b)(ii) and 7.5; and

(viii) Each Blocker shall deliver a properly executed certification that such entity's equity interests are not "U.S. real property interests" in accordance with the Treasury Regulations under Sections 897 and 1445 of the Code, together with a notice to the IRS in accordance with the provisions of Treasury Regulations section 1.897-2(h)(2) and Merger Partner shall deliver a properly executed certification satisfying the requirements of Temporary Treasury Regulations section 1.1445-11T to the effect that fifty percent (50%) or more of the value of the gross assets of Merger Partner does not consist of "U.S. real property interests", or that ninety percent (90%) or more of the value of the gross assets of Merger Partner does not consist of "U.S. real property interests" plus cash or cash equivalents.

1.3 Effects of the Mergers.

- (a) At and after the Blocker Merger Effective Time corresponding to the First Blocker Merger, without any action on the part of Blocker 1 or Public Company or the holders of securities of the foregoing, (i) Blocker 1 shall merge with and into Public Company, the separate existence of Blocker 1 shall cease, and Public Company shall survive the First Blocker Merger and continue its corporate existence, (ii) the effect of the First Blocker Merger shall be as provided in this Agreement and the applicable provisions of Delaware Law, (iii) the certificate of incorporation of Public Company shall be amended and restated in its entirety in a manner to be mutually agreed as soon as reasonably practicable following the date hereof by Merger Partner and Public Company (the "Amended Certificate") and (iv) the by-laws of Public Company shall be amended and restated in their entirety in a manner to be mutually agreed as soon as reasonably practicable following the date hereof by Merger Partner and Public Company (the "Amended By-Laws"). Without limiting the foregoing, Public Company, as the surviving corporation in the First Blocker Merger, shall thereupon and thereafter possess all of the rights, property, privileges, powers and franchises, of a public as well as a private nature, of each of Public Company and Blocker 1, and shall become subject to all the restrictions, disabilities and duties of each of Public Company and Blocker 1.
- (b) At and after the applicable Blocker Merger Effective Time of each Blocker Merger (other than the First Blocker Merger), without any action on the part of applicable Blocker being merged (other than Blocker 1) or Public Company or the holders of securities of the foregoing, (i) such Blocker shall merge with and into Public Company, the separate existence of such Blocker shall cease, and Public Company shall survive such Blocker Merger and continue its corporate existence and (ii) the effect of such Blocker Merger shall be as provided in this Agreement and the applicable provisions of Delaware Law. Without limiting the foregoing, Public Company, as the surviving corporation in each Blocker Merger (other than the First Blocker Merger), shall thereupon and thereafter possess all of the rights, property, privileges, powers and franchises, of a public as well as a private nature, of each of Public Company and the Blocker being merged in such Blocker Merger, and shall become subject to all the restrictions, disabilities and duties of each of Public Company and the Blocker being merged in such Blocker Merger.
- (c) At and after the Merger Partner Merger Effective Time without any action on the part of Merger Partner, Merger Subsidiary or Public Company or the holders of securities of the foregoing, (i) Merger Subsidiary shall merge with and into Merger Partner, the separate existence of Merger Subsidiary shall cease, and Merger Partner shall survive the Merger and continue its existence as a limited liability company, (ii) the effect of the Merger Partner Merger shall be as provided in this Agreement and the applicable provisions of Delaware Law and (iii)

the limited liability company agreement of Merger Partner shall be amended and restated in its entirety in a manner to be mutually agreed as soon as reasonably practicable following the date hereof by Merger Partner, Public Company and the Electing Members (the "Amended Operating Agreement"). Without limiting the foregoing, Merger Partner, as the surviving limited liability company of the Merger Partner Merger, shall thereupon and thereafter possess all of the rights, property, privileges, powers and franchises, of a public as well as a private nature, of the Merger Partner and Merger Subsidiary, and shall become subject to all the restrictions, disabilities and duties of each of Merger Partner and Merger Subsidiary.

1.4 Directors and Officers.

- (a) At and after the Blocker Merger Effective Time corresponding to the First Blocker Merger, without any action on the part of Blocker 1 or Public Company or the holders of securities of the foregoing, the individuals set forth on Exhibit D-1 shall be the directors of Public Company, each to hold office in accordance with the Amended Certificate and Amended By-Laws of Public Company and until their respective successors are duly elected and qualified or until such director's earlier death, resignation or removal.
- (b) At and after the Blocker Merger Effective Time corresponding to the First Blocker Merger, without any action on the part of Blocker 1 or Public Company or the holders of securities of the foregoing, the individuals set forth on Exhibit D-2 shall be the officers of Public Company, each to hold office in accordance with the Amended Certificate and Amended By-Laws of Public Company and until their respective successors are duly appointed or until such officer's earlier death, resignation or removal.
- 1.5 <u>Additional Action</u>. Public Company, at any time from and after the effective time of the Blocker Mergers and Merger Partner Merger, may take any lawful action, including executing and delivering any document, in the name and on behalf of either Merger Subsidiary, any Blocker or Merger Partner necessary or desirable in order to consummate and give effect to the Transactions.

ARTICLE II CONVERSION OF SECURITIES

2.1 Conversion of Securities.

(a) Blocker Mergers.

(i) At the applicable Blocker Merger Effective Time for each Blocker Merger, by virtue of such Blocker Merger and without any action on the part of Public Company, the Blocker being merged, any holder of limited liability company interests or capital stock, as applicable, in such Blocker or any other Person, (A) each limited liability company interest or share of capital stock, as applicable, of the Blocker being merged that is owned by such Blocker in treasury or by Public Company or any direct or indirect wholly-owned Subsidiary of Public Company as of immediately prior to the applicable Blocker Merger Effective Time shall be cancelled and cease to exist and no payment of consideration shall be delivered in exchange therefor and (B) each limited liability company interest or share of capital stock, as applicable, of the Blocker being merged issued and outstanding immediately prior to the applicable Blocker

Merger Effective Time (other than the limited liability company interests or share of capital stock, as applicable, referenced in clause (A) above) shall be automatically cancelled and converted into the right to receive the applicable portion of the Merger Consideration (in the form of shares of Public Company Class A Common Stock) corresponding to that Blocker Merger as set forth on the Allocation Schedule.

(ii) The shares of capital stock of Public Company outstanding immediately prior to each applicable Blocker Merger Effective Time shall remain outstanding.

(b) Merger Partner Merger.

- (i) At the Merger Partner Merger Effective Time, by virtue of the Merger Partner Merger and without any action on the part of Public Company, Merger Subsidiary, Merger Partner, any holder of limited liability company interests in Merger Partner or any other Person:
- (A) each Merger Partner Unit that is owned by Merger Partner in treasury as of immediately prior to the Merger Partner Merger Effective Time shall be cancelled and cease to exist and no payment of consideration shall be delivered in exchange therefor;
- (B) each Merger Partner Unit (other than a Merger Partner PIU) held by an Electing Member (as determined pursuant to Section 2.5(b)) issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall automatically be cancelled and converted into the right to receive in exchange therefor, at the election by such Electing Member pursuant to Section 2.5, either (I) the number of shares of Public Company Class A Common Stock set forth on the Allocation Schedule, or (II) the number of Merger Partner Common Units set forth on the Allocation Schedule and an equal number of shares of Public Company Class B Common Stock (a Merger Partner Common Unit and a share of Public Company Class B Common Stock, collectively, an "Up-C Unit");
- (C) each Merger Partner Warrant issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall automatically be exchanged for the right to receive the number of shares of Public Company Class A Common Stock set forth on the Allocation Schedule;
- (D) each Merger Partner Unit (other than a Merger Partner PIU) held by a Participating Member (or an Electing Member with respect to any Merger Partner Units held by such Electing Member that are not Excluded Units) issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall (without duplication of any conversions pursuant to Section 2.1(b)(i)(B) above) automatically be cancelled and converted into the right to receive the number of shares of Public Company Class A Common Stock set forth on the Allocation Schedule; and
- (E) each Merger Partner PIU issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall, (I) to the extent not a "restricted unit" under the award agreements and governing documents applicable to such Merger Partner PIU as of such time, be automatically cancelled and converted into the right to

receive the number of shares of Public Company Class A Common Stock set forth on the Allocation Schedule and (II) to the extent a "restricted unit" (or unvested) under the award agreements and other governing documents applicable to such Merger Partner PIU as of such time, be automatically cancelled and converted into the right to receive the number of restricted shares of Public Company Class A Common Stock pursuant to a Public Company Stock Plan, as set forth on the Allocation Schedule and subject to the same vesting schedule as was applicable to such Merger Partner PIU.

- (ii) Immediately following the consummation of the Merger Partner Merger, and without any action on the part of Public Company, Merger Subsidiary, Merger Partner, any other holder of limited liability company interests in Merger Partner or any other Person, (A) each limited liability company interest of Merger Subsidiary issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall be automatically be cancelled and cease to exist, (B) each Merger Partner Unit or Merger Partner Warrant held by the Public Company by operation of the Blocker Mergers issued and outstanding immediately prior to the Merger Partner Merger Effective Time shall automatically be converted or exchanged for the number of Merger Partner Common Units set forth in the Allocation Schedule, (C) each Merger Partner Warrant held by Public Company as a result of the exchanges set forth in Section 2.1(b)(i)(C) shall automatically be converted or exchanged for the number of Merger Partner Common Units set forth in the Allocation Schedule and (D) Merger Partner shall issue an aggregate number of Merger Partner Common Units to Public Company in an equal amount to the number shares of Public Company Class A Common Stock issued pursuant to Sections 2.1(b)(i)(B)(I), 2.1(b)(i)(D), and 2.1(b)(i)(E) (without duplication of Merger Partner Common Units issued pursuant to the foregoing clauses (B) and (C)) as set forth in the Allocation Schedule.
- (c) <u>Certain Adjustments to Per Share Amounts</u>. All per-share or per-unit amounts of the Merger Consideration shall be equitably adjusted, as applicable and appropriate, to reflect fully the effect of, and to provide the holders of equity securities of Public Company, Blockers or Merger Partner the same economic effect as contemplated by this Agreement prior to the occurrence of, any reclassification, stock split, reverse split, stock dividend (including any dividend or distribution of securities convertible into Company Stock), reorganization, recapitalization or other like change with respect to the Public Company Common Stock or the Merger Partner Units or any securities convertible or exchangeable into or exercisable for shares of Public Company Common Stock or Merger Partner Units occurring (or for which a record date is established) after the date of this Agreement and prior to the consummation of the First Blocker Merger. Nothing in this Section 2.1(c) will be construed to permit Public Company, Merger Partner, Merger Sub, or Blockers to take any action prohibited by the terms of this Agreement.
- (d) No Fractional Shares. Notwithstanding any other provision of this Agreement, no fractional shares of Public Company Class A Common Stock or Public Company Class B Common Stock shall be issued in exchange for any Merger Partner Units or equity securities in a Blocker, and no holder of any of the foregoing shall be entitled to receive a fractional share of Public Company Class A Common Stock or Public Company Class B Common Stock. Any fractional share of Public Company Class A Common Stock or Public Company Class B Common Stock issuable pursuant to Sections 2.1(a) or 2.1(b), shall be rounded up to the nearest whole share.

- 2.2 <u>Payment Fund</u>. The procedures for exchanging outstanding Merger Partner Units, Merger Partner Warrants or equity securities of any Blocker for the shares of Public Company Class A Common Stock, shares of Public Company Class B Common Stock, or Merger Partner Common Units to be issued to such holders in connection with the Mergers are as follows:
- (a) <u>Deposit at Closing</u>. At the Closing, (i) Public Company shall deliver to the Exchange and Paying Agent shares of Public Company Class A Common Stock and Public Company Class B Common Stock issued in book entry form equal to the amount of the Merger Consideration payable in shares of Public Company Class A Common Stock and Public Company Class B Common Stock and (ii) Merger Partner shall deliver to the Exchange and Paying Agent Merger Partner Common Units in book entry form equal to the amount of the Merger Consideration payable in Merger Partner Common Units (the "<u>Payment Fund</u>"). For the avoidance of doubt, any Merger Partner Common Units to be issued to Public Company pursuant to Section 2.1(b)(ii), shall be issued by Merger Partner directly to Public Company immediately following the consummation of the Merger Partner Merger.
- (b) Exchange and Paying Agent. The Exchange and Paying Agent shall, pursuant to instructions from Public Company and Merger Partner in accordance with the Exchange and Paying Agent Agreement and the Allocation Schedule, deliver (i) the applicable portion of the Merger Consideration to be paid in shares of Public Company Class A Common Stock, shares of Public Company Class B Common Stock and Merger Partner Common Units to the equity holders of Merger Partner and (ii) the applicable portion of the Merger Consideration for each Blocker Merger to be paid in shares of Public Company Class A Common Stock to the equity holders of the corresponding Blocker being merged in such Blocker Merger. The Payment Fund shall not be used for any purpose other than as specified in this Section 2.2(a).
- (c) Exchange Procedures. Promptly, but in no event more than three (3) Business Days, following the date the Registration Statement is declared effective by the SEC, Public Company shall cause the Exchange and Paying Agent to mail to each Person who was a member of Merger Partner or any Blocker as of immediately prior to the effective time of each respective Mergers, (i) a Letter of Transmittal and (ii) instructions for effecting the issuance of the portion of the Merger Consideration, as applicable, payable with respect thereto pursuant to the terms of this Agreement and the Allocation Schedule. Upon proper delivery of a duly completed and executed Letter of Transmittal, each such Person shall be entitled to receive the number of shares of Public Company Class A Common Stock, shares of Public Company Class B Common Stock and Merger Partner Common Units as determined in accordance with Section 2.1 and reflected on the Allocation Schedule. If payment is to be made to a Person other than the Person who is reflected in the books and records of the applicable company as the applicable member, it shall be a condition of payment that the Person requesting such payment shall have established to the satisfaction of Public Company and the Exchange and Paying Agent that any transfer and other Taxes required by reason of such payment to a Person other than the applicable member have been paid or are not applicable. From and after the Merger Partner Merger Effective Time, or each applicable Blocker Merger Effective Time, as applicable, other than Electing Members (with respect to Excluded Units) each Person that was a member of

Merger Partner or the corresponding Blocker being merged, respectively (or the holder of an economic interest therein) shall have no rights as a member or holder of any such economic interests and, until the procedures contemplated by this Section 2.2 have been complied with, any such former limited liability company or economic interest shall represent only the right to receive the applicable portion of the Merger Consideration, respectively, on the terms and subject to the conditions of this Agreement.

- (d) No Further Ownership Rights. All Merger Consideration paid in accordance with the terms hereof shall be deemed to have been paid in satisfaction of all rights pertaining to the applicable limited liability company or other interests in Merger Partner and each Blocker, respectively, and from and after the Merger Partner Merger Effective Time and the applicable Blocker Merger Effective Time, as applicable, (except as necessary to reflect the issuance of Merger Partner Common Units as Merger Consideration pursuant to Section 2.1) there shall be no further registration of transfers on the register or members or other books of Merger Partner or the corresponding Blocker which were outstanding as of immediately prior to the Merger Effective Time and the applicable Blocker Merger Effective Time, as applicable. If, after the Merger Partner Merger Effective Time and the applicable Blocker Merger Effective Time, as applicable, claims of ownership are presented to Public Company, Merger Partner or the Exchange and Paying Agent for any reason, they shall be cancelled and exchanged as provided in this Article II, subject to Section 2.2(f).
- (e) <u>Termination of Payment Fund</u>. Any portion of the Payment Fund deposited with the Exchange and Paying Agent (including shares of Public Company Class A Common Stock or Public Company Class B Common Stock issued to a Public Company reserve account) that remains undistributed as of one year after the Closing shall be delivered to Public Company (subject to abandoned property, escheat or similar Law), upon demand, and any Person who is entitled to such amount under this Section 2.2 shall (subject to Section 2.2(f)) be entitled to seek payment of such amount from Public Company only as a general creditor thereof.
- (f) No Liability. To the extent permitted by applicable Law, none of Public Company, Merger Subsidiary, Merger Partner, any Blocker or the Exchange and Paying Agent shall be liable to any Person for any amount delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law. If the applicable exchange procedures shall not have been completed prior to the first (1st) anniversary of the Closing Date (or immediately prior to such earlier date on which the related consideration payable pursuant to this Article II would otherwise escheat to or become the property of any Governmental Entity), any such consideration in respect thereof shall, to the extent permitted by applicable Law, become the property of Public Company, free and clear of all claims or interest of any Person previously entitled thereto.
- 2.3 <u>Allocation Schedule</u>. The Allocation Schedule sets forth a true, correct and complete allocation of the amounts payable to the equity holders of Merger Partner and the Blockers pursuant to this Agreement as of the date of this Agreement and the pro forma equity capitalization of Public Company and Merger Partner giving effect to the Mergers. Merger Partner shall deliver an updated Allocation Schedule reflecting any transfers of equity interests in Merger Partner or any Blocker or issuances of equity or equity linked securities of Merger Partner after the date hereof (which issuances for the avoidance of doubt shall only be made in

compliance with the terms of this Agreement), as well as the elections contemplated by Section 2.5, to Public Company at least three (3) Business Days prior to the Closing, which updated Allocation Schedule shall not, for the avoidance of doubt, change the aggregate number of shares of Public Company Class A Common Stock and Public Company Class B Common Stock comprising the Merger Consideration or the aggregate number of Merger Partner Common Units to be issued in connection with the Mergers (subject to the election rights of the Electing Members pursuant to Section 2.5); provided, however, that in the event Merger Partner sells additional Merger Partner Series F Preferred Units after the date of this Agreement at a cash price per Series F Preferred Unit of \$2.00 and otherwise on terms substantially the same as issuances of Series F Preferred Units by Merger Partner prior to the date hereof (each, an "Additional Merger Partner Unit") the Allocation Schedule shall be updated to reflect such issuances of Additional Series F Preferred Unit(s) of Merger Partner and the number of shares of Public Company Class A Common Stock, Public Company Class B Common Stock and Member Partner Class B Units comprising the Merger Consideration in exchange therefor and any additional Member Partner Class A Units to be issued as a result pursuant to Section 2.1(b)(ii). Public Company, Merger Partner and the Exchange and Paying Agent shall be entitled to rely conclusively on the Allocation Schedule as in effect from time to time and any amounts delivered by Public Company and Merger Partner to any security holder of Merger Partner or a Blocker (or delivered by Public Company to the Exchange and Paying Agent for further delivery to such Persons) in accordance with the Allocation Schedule, shall be deemed for all purposes to have been delivered to the applicable equity holder in full satisfaction of the obligations of Public Company and Merger Partner under this Article II.

2.4 Withholding Rights. Public Company, Merger Partner and the Exchange and Paying Agent will be entitled to deduct and withhold from the amounts otherwise payable by it pursuant to this Agreement to any Person, such amounts as it reasonably determines that it is required to deduct and withhold with respect to the making of such payment under the Code, or any provision of state, local or foreign Tax Law, and to collect any necessary Tax forms, including Forms W-8 or W-9, as applicable, or any similar information, from any equity holders of Merger Partner or a Blocker and any other recipients of payments hereunder, provided, that if any payment to the equity holders of Merger Partner or a Blocker is subject to deduction or withholding (other than any payment to a Merger Partner employee in its capacity as such or a requirement to deduct or withhold that is attributable to a failure to provide a duly completed Form W-8 or W-9, as applicable, to the Exchange and Paying Agent), then the Public Company or the Exchange and Paying Agent shall (i) provide notice to Merger Partner or the applicable Blocker as soon as reasonably practicable after such determination and prior to such payment and (ii) reasonably cooperate with Merger Partner or the applicable Blocker to reduce or eliminate any such deduction or withholding to the extent permitted by applicable Law. In the event that any amount is so deducted and withheld, and properly remitted, such amount will be treated for all purposes of this Agreement as having been paid to the Person to whom the payment from which such amount was withheld was made.

2.5 Merger Consideration Election Procedures.

- (a) Promptly, but in no event more than three (3) Business Days, following the date the Registration Statement declared effective by the SEC (the "Mailing Date"), Merger Partner will mail, or cause the Exchange and Paying Agent to mail, to each holder of Merger Partner Units (other than Merger Partner PIUs) that is a Top 99 Merger Partner Holder as of the Mailing Date (or another date selected by Merger Partner which is reasonably acceptable to Public Company) a merger consideration election form in a form mutually satisfactory to Merger Partner and Public Company (the "Election Form"). The Election Form will accompany the Disclosure Statement and copies of the form of Tax Receivable Agreement, Amended Operating Agreement, Exchange Agreement and Registration Rights Agreement and the Election Form will include counterpart signatures to each of the foregoing agreements.
- (b) Each Election Form will permit each holder (other than Blockers) of Merger Partner Units that is a Top 99 Merger Partner Holder, to elect on a per-Merger Partner Unit basis (other than in respect of Merger Partner PIUs), to receive as Merger Consideration in the Merger Partner Merger in the form of either (A) shares of Public Company Class A Common Stock or (B) Up-C Units (subject to the limitations in this Section 2.5(b), any such holder validly electing to receive Up-C Units in exchange for any one or more of its Merger Partner Units, an "Electing Member", and the Merger Partner Units that such Electing Member will exchange for Up-C Units, the "Excluded Units"); provided, however, that no Electing Member will be entitled to receive Up-C Units to the extent such holder's Excluded Units exchanged therefor would represent more than ninety-five percent (95%), or such lower percentage as may be determined by Merger Partner and set forth on the Election Form, of the total number of Merger Partner Units held by such Person as of immediately prior to the Merger Partner Merger Effective Time (for the avoidance of doubt, any Merger Partner Units held by such Person in excess of such amount will not constitute Excluded Units and will instead receive shares of Public Company Class A Common Stock in exchange for the Merger Partner Units representing such excess, notwithstanding such Person's election).
- (c) Any Merger Partner Units with respect to which an election may be made and the Exchange and Paying Agent does not receive a properly completed Election Form during the period (the "Election Period") from the Mailing Date to 5:00 p.m., New York Time, on the Business Day that is five (5) Business Days prior to the Closing Date or such other date as Merger Partner and Public Company will, prior to the Closing, mutually agree (the "Election Deadline") will be deemed to be held by a Participating Member and will receive the form of Merger Consideration set forth in Section 2.1(b)(i)(D). If the Closing Date is delayed to a subsequent date, the Election Deadline shall be similarly delayed to a subsequent date, and Merger Partner and Public Company shall promptly announce any such delay and, when determined, the rescheduled Election Deadline.
- (d) Any election made pursuant to this Section 2.5 will have been properly made only if the Exchange and Paying Agent has actually received a properly completed Election Form (including duly executed counterpart signature pages to the Tax Receivable Agreement, the Amended Operating Agreement, the Exchange Agreement and the Registration Rights Agreement) during the Election Period. Any Election Form may be revoked or changed by the Person submitting it, by written notice received by the Exchange and Paying Agent during the Election Period. In the event an Election Form is revoked during the Election Period, the Merger Partner Units represented by such Election Form will be subject to Section 2.5(c) except to the extent a subsequent election is properly made during the Election Period. Any Election Form indicating Electing Member status must be accompanied by duly executed signature pages to the Tax Receivable Agreement, the Amended Operating Agreement, the Exchange Agreement

and the Registration Rights Agreement for such Election Form to be validly received. Subject to the terms of this Agreement and of the Election Form, the Exchange and Paying Agent will have reasonable discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in the Election Forms, and any good faith decisions of the Exchange and Paying Agent regarding such matters will be binding and conclusive. None of Merger Partner, Public Company or the Exchange and Paying Agent will be under any obligation to notify any Person of any defect in an Election Form.

ARTICLE III REPRESENTATIONS AND WARRANTIES OF MERGER PARTNER AND BLOCKERS

Merger Partner represents and warrants to Public Company that, except as set forth in Merger Partner Disclosure Schedule, the statements contained in Sections 3.1 through 3.21, and 3.29 and 3.30 (with respect to Merger Partner) are true and correct as of the date of this Agreement and as of the Closing Date, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties will be true and correct as of such date). Each Blocker (solely with respect to itself and not to any other Blocker or Merger Partner) represents and warrants to Public Company that, except as set forth in Merger Partner Disclosure Schedule, the statements contained in Sections 3.22 through 3.28, and 3.29 and 3.30 (solely with respect to such Blocker) are true and correct as of the date of this Agreement and as of the Closing Date, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties will be true and correct as of such date). Merger Partner Disclosure Schedule shall be arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs contained in this Article III; provided, however, that the disclosures in any section or paragraph of Merger Partner Disclosure Schedule shall qualify (a) the corresponding section or paragraph in this Article III and (b) such other sections or paragraphs in this Article III (whether or not there is a specific cross reference) to the extent that it is reasonably apparent on the face of the disclosure that such disclosure also qualifies or applies to such other section or paragraph.

3.1 <u>Organization, Standing and Power</u>. Merger Partner is a limited liability company duly organized, validly existing and in good standing under the Laws of the State of Delaware. Except as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect, Merger Partner is duly qualified to conduct business and is in good standing under the Laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification. Merger Partner has all requisite power and authority to (i) carry on the businesses in which it is engaged, (ii) own and use the properties owned and used by it and (iii) perform its obligations under all Merger Partner Contracts. Merger Partner has made available to Public Company complete and accurate copies of the Organizational Documents of Merger Partner as in effect on the date of this Agreement. Merger Partner is not in material default under or in material violation of any provision of its Organizational Documents.

3.2 Capitalization.

- (a) Section 3.2(a) of the Merger Partner Disclosure Schedule sets forth the authorized capital of Merger Partner and the issued and outstanding equity interests of Merger Partner, in each case as of the date of this Agreement.
- (b) Section 3.2(b) of Merger Partner Disclosure Schedule sets forth a complete and accurate list, as of the date of the Agreement, of (i) the members and holders of limited liability company interests of each of Merger Partner, showing the number and class(es) held by each holder and (ii) the holders of warrants to purchase limited liability company interest of the Merger Partner. Section 3.2(b) of Merger Partner Disclosure Schedule also indicates all limited liability company interests that are subject to a threshold value, vesting or otherwise subject to a repurchase or redemption right, indicating the name of the applicable member or interest holder, the threshold value, vesting schedule (including any acceleration provisions with respect thereto), and the repurchase price payable by Merger Partner. All of the limited liability company interests of Merger Partner have been duly authorized, validly issued, and free of all preemptive rights. To the Knowledge of Merger Partner, all of the limited liability company interests of Merger Partner have been offered, issued and sold by Merger Partner in compliance in all material respects with all applicable federal and state securities Laws.
- (c) Except as set forth in Section 3.2(b), (A) there are no limited liability company interests of any class of Merger Partner, or any security exchangeable into or exercisable for such membership interests, issued, reserved for issuance or outstanding, (B) there are no options, warrants, securities, calls, rights, commitments or agreements to which Merger Partner is a party or by which Merger Partner is bound obligating Merger Partner to issue, exchange, transfer, deliver or sell, or cause to be issued, exchanged, transferred, delivered or sold, additional limited liability company interests of Merger Partner or any security or rights convertible into or exchangeable or exercisable for any such membership interests, or obligating Merger Partner to grant, extend, accelerate the vesting of, otherwise modify or amend or enter into any such option, warrant, equity security, call, right, commitment or agreement, and (C) Merger Partner have no obligation (contingent or otherwise) to purchase, redeem or otherwise acquire any of their limited liability company interests or to make any distribution in respect thereof.
- (d) Except for the Organizational Documents of Merger Partner or as set forth on Section 3.2(d) of the Merger Partner Disclosure Schedule, there is no agreement, written or, to the Knowledge of Merger Partner, oral, between Merger Partner and any holder of its limited liability company interests or, to the Knowledge of Merger Partner, among any holders of the limited liability company interests of Merger Partner, relating to the sale or transfer (including agreements relating to rights of first refusal, co-sale rights or "drag along" rights), registration under the Securities Act or the securities Laws of any other jurisdiction, or voting, of the limited liability company interests of Merger Partner.
- (e) The Allocation Schedule sets forth as of the date of this Agreement a true, correct and complete summary of the allocation of the amounts payable to the equity holders of Merger Partner and the Blockers pursuant to this Agreement and the pro forma equity capitalization of each of Public Company and Merger Partner giving effect to the Closing. The allocation of payments set forth on the Allocation Schedule complies with the terms of Organizational Documents of Merger Partner.

3.3 Subsidiaries.

- (a) Section 3.3(a) of Merger Partner Disclosure Schedule sets forth each Subsidiary of Merger Partner, all of which are wholly-owned by Merger Partner. Merger Partner does not own or control directly or indirectly or have any direct or indirect equity participation or similar interest in, or any obligation to provide funding to, any corporation, partnership, limited liability company, joint venture, trust or other business association or entity.
- (b) Each Subsidiary of Merger Partner (i) is duly organized, validly existing and in standing under the Laws of the jurisdiction of its formation, (ii) is duly qualified to conduct business and is in good standing under the Laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification except as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect, and (iii) has all requisite power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it. Merger Partner has delivered to Public Company complete and accurate copies of the Organizational Documents of each Subsidiary of Merger Partner. There are no outstanding or authorized options, warrants, rights, agreements or commitments to which Merger Partner or any Subsidiary of Merger Partner is a party or which are binding on any of them providing for the issuance, disposition or acquisition of any equity interests of any such Subsidiary other than to Merger Partner or another Subsidiary.
- (c) All of the outstanding capital stock of or other voting securities of or ownership interests in each Subsidiary of Merger Partner, is owned by Merger Partner, directly or indirectly, free and clear of any Lien and free of any transfer restriction (other than transfer restrictions of general applicability as may be provided under the Securities Act or other applicable securities Laws), including any restriction on the right to vote, sell or otherwise dispose of such capital stock or other voting securities or ownership interests.
- (d) Except for its interests in each Subsidiary of Merger Partner, Merger Partner does not own, directly or indirectly, any capital stock, membership interest, partnership interest, joint venture interest or other equity interest in any other Person. There are no outstanding contractual obligations of Merger Partner or any of the Subsidiaries of Merger Partner in any Person.

3.4 Authority; No Conflict; Required Filings and Consents.

(a) Merger Partner has all requisite power and authority to execute and deliver this Agreement and the other agreements contemplated hereby and to perform its respective obligations hereunder and thereunder. The execution and delivery by Merger Partner of this Agreement and the other agreements contemplated hereby and, subject to obtaining the Merger Partner Member Approval, which are the only approval required from the members of Merger Partner, the performance by Merger Partner of this Agreement and the consummation by Merger Partner of the Transactions have been duly and validly authorized by all necessary company or

other corporate action on the part of Merger Partner. This Agreement and all other agreements contemplated hereby have been duly and validly executed and delivered by each of Merger Partner and, assuming the due execution and delivery hereof and thereof by the other parties hereto and thereto, constitutes or will constitute a valid and binding obligation of Merger Partner, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, and other similar Laws affecting creditors' rights generally and by general principles of equity.

- (b) The Merger Partner Board (by unanimous written consent) has: (i) determined that the Transactions are fair to, advisable, and in the best interests of Merger Partner and its Subsidiaries; (ii) authorized, approved and declared advisable this Agreement and the Transactions, including the Merger Partner Merger; and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the Merger Partner Board has unanimously voted to adopt this Agreement and thereby approve the Transactions.
- (c) Subject to the filing of the Merger Partner Certificate of Merger as required by Delaware Law, neither the execution and delivery by Merger Partner of this Agreement or any other agreement contemplated hereby, nor the performance by Merger Partner of its obligations hereunder or thereunder, nor the consummation by Merger Partner of the Transactions, will (i) conflict with or violate any provision of the Organizational Documents of Merger Partner, each as amended or restated to date, (ii) require on the part of Merger Partner or any of its Subsidiaries any notice to or filing with, or any permit, authorization, consent or approval of, any Governmental Entity, (iii) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, any contract, lease, sublease, license, sublicense, franchise, permit, indenture, agreement or mortgage for borrowed money, instrument of Indebtedness, Lien (other than Permitted Liens) or other arrangement to which Merger Partner or any of its Subsidiaries is a party or by which Merger Partner or any of its Subsidiaries is bound or to which any of the assets of Merger Partner or any of its Subsidiaries or (v) violate any order, writ, injunction, decree, statute, rule or regulation applicable to Merger Partner or any of its Subsidiaries or any properties or assets of any of the foregoing, except in the case of clauses (ii) through (v) as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect.

3.5 Financial Statements; Information Provided.

(a) Merger Partner has made available to Public Company the Merger Partner Financial Statements. The Merger Partner Financial Statements (i) comply as to form with all applicable accounting requirements and (ii) were prepared in accordance with GAAP applied on a consistent basis throughout the periods covered thereby; <u>provided</u>, <u>however</u>, that the Merger Partner Financial Statements referred to in clause (b) of the definition of such term are subject to normal recurring year-end adjustments (which, individually and in the aggregate, will not be material) and do not include footnotes.

- (b) Each of the Merger Partner Financial Statements fairly presents the consolidated assets, liabilities, business, financial condition, results of operations and cash flows of Merger Partner and its Subsidiaries as of the date thereof and for the period referred to therein, and is consistent with the books and records of Merger Partner and its Subsidiaries.
- (c) Merger Partner maintains accurate books and records reflecting its assets and liabilities and maintains proper and a system of internal accounting control over financial reporting designed to provide reasonable assurance that (i) transactions are executed with management's authorization, (ii) transactions are recorded as necessary to permit preparation of the financial statements of Merger Partner and to maintain accountability for Merger Partner's assets, (iii) access to assets of Merger Partner is permitted only in accordance with management's authorization, (iv) the reporting of assets of Merger Partner is compared with existing assets at regular intervals, and (v) accounts, notes and other receivables and inventory were recorded accurately, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis. Merger Partner has disclosed to Merger Partner's auditors and audit committee (and made available to Public Company a summary of the significant aspects of such disclosure) (A) all significant deficiencies and material weaknesses, if any, in the design or operation of internal control over financial reporting that involves management or other employees who have a significant role in Merger Partner's or its Subsidiaries' internal control over financial reporting. Merger Partner has not identified any material weaknesses in the design or operation of Merger Partner's internal control over financial reporting.
- (d) Neither Merger Partner nor any of its Subsidiaries has any Liability, except for: (i) Liabilities disclosed, reflected or reserved against in Public Company's most recent interim consolidated unaudited balance sheet for the period ended September 30, 2022; (ii) Liabilities that have been incurred by Merger Partner or its Subsidiaries since September 30, 2022 in the Ordinary Course of Business; (iii) Liabilities for performance of obligations of Merger Partner or any of its Subsidiaries under Merger Partner Contracts (other than those arising as a result of a breach or default thereunder or as a result of failure to comply with applicable Law); (iv) Liabilities for payment of fees and expenses incurred in connection with the Transactions; (v) Liabilities which would not, individually or in the aggregate, reasonably be expected to be material to Merger Partner and its Subsidiaries, taken as a whole; and (vi) Liabilities described in Section 3.5(d) of the Merger Partner Disclosure Schedule.
- (e) The information (including information relating to the Blockers) to be supplied by or on behalf of Merger Partner for inclusion or incorporation by reference in the registration statement on Form S-4 to be filed by Public Company pursuant to which shares of Public Company Class A Common Stock and Public Company Class B Common Stock issued in connection with the Mergers shall be registered under the Securities Act (the "Registration Statement"), or supplied by or on behalf of Merger Partner for inclusion in any filing pursuant to Rule 165 and Rule 425 under the Securities Act or Rule 14a-12 under the Exchange Act (each a "Regulation M-A Filing"), shall not at the time the Registration Statement or any such Regulation M-A Filing is filed with the SEC, at any time it is amended or supplemented or at the time the Registration Statement is declared effective by the SEC, as applicable, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein

or necessary in order to make the statements therein not misleading, in light of the circumstances under which they were made. The information (including information relating to the Blockers) to be supplied by or on behalf of Merger Partner for inclusion or incorporation by reference in the proxy statement/prospectus (the "<u>Proxy Statement/Prospectus</u>") to be sent to the stockholders of Public Company in connection with the meeting of Public Company's Stockholders' Meeting shall not, on the date the Proxy Statement/Prospectus is first mailed to stockholders of Public Company, or at the time of the Public Company Meeting or as of the Effective Time, contain any statement that, at such time and in light of the circumstances under which it shall be made, is false or misleading with respect to any material fact, or omit to state any material fact necessary in order to make the statements made in the Proxy Statement/Prospectus not false or misleading; or omit to state any material fact necessary to correct any statement in any earlier communication with respect to the solicitation of proxies for the Public Company Meeting that has become, in light of the circumstances in which they were made, false or misleading.

- (f) Between January 1, 2020 and the date of this Agreement, there have been no formal investigations regarding financial reporting or accounting policies and practices discussed with, reviewed by or initiated at the direction of the chief executive officer, chief financial officer, principal accounting officer or general counsel of Merger Partner, the Merger Partner Board or any committee thereof, other than ordinary course audits or reviews of accounting policies and practices or internal controls.
- 3.6 <u>Absence of Certain Changes</u>. Since December 31, 2021, except for the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto, (a) there has occurred no event or development which, individually or in the aggregate, has had, or could reasonably be expected to have in the future, a Merger Partner Material Adverse Effect, (b) Merger Partner and its Subsidiaries have conducted their businesses in the Ordinary Course of Business and (c) neither Merger Partner nor any Subsidiary has taken any of the actions set forth in clauses (a) through (o) of Section 5.1.

3.7 Merger Partner Tax Matters.

- (a) Each of Merger Partner and its Subsidiaries has properly filed on a timely basis all income and other material Tax Returns that it was required to file, and all such Tax Returns were true, correct and complete in all material respects. Each of Merger Partner and its Subsidiaries has paid on a timely basis all income or other material Taxes, whether or not shown on any Tax Return, that were due and payable.
- (b) All material Taxes that Merger Partner or any of its Subsidiaries was required by Law to withhold or collect have been duly withheld or collected and, to the extent required, have been properly paid to the appropriate Governmental Entity, in each case in compliance in all material respects with applicable law.
- (c) No examination or audit of any Tax Return of Merger Partner or any of its Subsidiaries by any Governmental Entity is currently in progress or, to the Knowledge of Merger Partner, has been threatened by any Governmental Entity. No deficiencies for Taxes of Merger Partner or any of its Subsidiaries have been claimed, proposed or assessed by any Governmental Entity in writing. Neither Merger Partner nor any of its Subsidiaries has been informed in writing

by any jurisdiction in which Merger Partner or any of its Subsidiaries does not file a Tax Return that the jurisdiction believes that Merger Partner or any of its Subsidiaries was required to file any Tax Return that was not filed or is subject to Tax in such jurisdiction. Neither Merger Partner nor any of its Subsidiaries has (i) waived any statute of limitations with respect to Taxes or agreed to extend the period for assessment or collection of any Taxes, which waiver or extension is still in effect, (ii) requested any extension of time within which to file any Tax Return (other than any automatic extension granted in the ordinary course of business and consistent with past custom and practice of Merger Partner), or (iii) executed or filed any power of attorney with any taxing authority, which is still in effect.

- (d) Neither Merger Partner nor any of its Subsidiaries has participated in any "reportable transaction" as defined in Treasury Regulations Section 1.6011-4(b) or a "listed transaction" within the meaning of Treasury Regulations Section 1.6011-4(b)(2) or any analogous provision of state or local Law.
 - (e) There are no Liens with respect to Taxes on any of the assets of Merger Partner or its Subsidiaries, other than Permitted Liens.
- (f) With the exception of customary commercial leases or contracts that are not primarily related to Taxes entered into in the Ordinary Course of Business and liabilities thereunder, neither Merger Partner nor any of its Subsidiaries (i) has any liability under Treasury Regulations Section 1.1502-6 (or any comparable or similar provision of state, local or non-U.S. law), as a transferee or successor, or pursuant to any contractual obligation for any Taxes of any Person other than Merger Partner or any of its Subsidiaries, or (ii) is a party to or bound by any Tax indemnity, Tax sharing, Tax allocation or similar agreement.
- (g) During the two-year period ending on the date hereof, neither Merger Partner (including any predecessor thereof) nor any of its Subsidiaries (including any predecessor thereof) has constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).
- (h) Other than Section 3.15, this Section 3.7 provides the sole and exclusive representations and warranties of Merger Partner in respect of Tax matters.
- 3.8 <u>Assets</u>. Except as would not be, individually or in the aggregate, material to Merger Partner and its Subsidiaries, taken as a whole, Merger Partner or its applicable Subsidiary is the true and lawful owner of, and has good title to, all of the assets (tangible or intangible) purported to be owned by Merger Partner or such Subsidiary, free and clear of all Liens other than Permitted Liens. Merger Partner and each Subsidiary owns or leases all material tangible assets sufficient for the conduct of its businesses as presently conducted, which tangible assets are reflected in the Merger Partner Financial Statements (other than to the extent disposed of in the Ordinary Course of Business). Except as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect, each such tangible asset is free from defects, has been maintained in accordance with normal industry practice, is in good operating condition and repair (subject to normal wear and tear) and is suitable for the purposes for which it presently is used.

3.9 Owned and Leased Real Property.

- (a) Neither Merger Partner nor any of its Subsidiaries owns any real property.
- (b) Merger Partner has delivered to Public Company complete and accurate copies of all material Leases of Merger Partner and its Subsidiaries. Neither Merger Partner nor any Subsidiary occupies any space other than pursuant to a Lease. With respect to each such Lease, except as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect:
- (i) such Lease is legal, valid, binding, enforceable and in full force and effect against Merger Partner or its Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Merger Partner, against each other party thereto;
- (ii) such Lease will continue to be legal, valid, binding, enforceable and in full force and effect against Merger Partner or the Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Merger Partner, against each other party thereto immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing;
- (iii) none of Merger Partner, any Subsidiary of Merger Partner or, to the Knowledge of Merger Partner, any other party, is in breach or violation of, or default under, any such Lease, and no event has occurred, is pending or, to the Knowledge of Merger Partner, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute any such breach or default by Merger Partner or any Subsidiary of Merger Partner or, to the Knowledge of Merger Partner, any other party under such Lease; and no event has occurred that would give rise to a termination right under such Lease;
- (iv) neither Merger Partner nor any Subsidiary of Merger Partner has assigned, transferred, conveyed, mortgaged, subleased, licensed, deeded in trust or encumbered any interest in the leasehold or subleasehold; and
- (v) to the Knowledge of Merger Partner, there are no Liens (other than Permitted Liens) applicable to the real property subject to such Lease which would reasonably be expected to impair the current uses or the occupancy by Merger Partner or any Subsidiary of Merger Partner of the property subject thereto.
- (c) Each of Merger Partners and its Subsidiaries owns, and has good and valid title to, or, in the case of leased properties and assets, valid leasehold interests in, all tangible properties or tangible assets and equipment used or held for use in its business or operations or purported to be owned by it, in each case, that are material to Merger Partner and its Subsidiaries, taken as a whole, including: (A) all such tangible assets reflected on the Merger Partner Financial Statements; and (B) all other such tangible assets reflected in the books and records of Merger Partner or any of its Subsidiaries as being owned by Merger Partner or such Subsidiary. All of such assets are owned or, in the case of leased assets, leased by Merger Partner or any of its Subsidiaries free of any Liens other than Permitted Liens.

3.10 Intellectual Property.

(a) Section 3.10(a) of the Merger Partner Disclosure Schedule sets forth a true, correct and complete list of all Patents, Copyright registrations, Trademark registrations for Trademark registration, material unregistered Trademarks, Design registrations and applications for Design registration, and Internet domain names owned (in whole or in part) by or exclusively licensed to Merger Partner or any of its Subsidiaries.

(b) (i) Merger Partner and its Subsidiaries (A) solely own all Intellectual Property owned or purported to be owned by Merger Partner or any of its Subsidiaries and (B) possess, hold or have obtained valid and enforceable licenses or other rights under all other Intellectual Property necessary to operate the business of Merger Partner in the manner now operated and as currently proposed to be operated by it as disclosed in the Registration Statement and Prospectus, except, in each case of (A) and (B), as such failure to own or obtain such licenses would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect; (ii) none of the Intellectual Property owned by, exclusively licensed to, or to the Knowledge of Merger Partner, otherwise licensed to or used by, the Merger Partner and its Subsidiaries (collectively, the "Merger Partner Intellectual Property") has been adjudged invalid or unenforceable, in whole or in part; (iii) to the Knowledge of Merger Partner, no third party is infringing, misappropriating, or otherwise violating any Merger Partner Intellectual Property, except, where such infringement, misappropriation or other violation would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect; (iv) since January 1, 2020 to the date of this Agreement, to the Knowledge of Merger Partner, there has been no actual or threatened action, suit, proceeding or claim by others challenging the validity, enforceability or scope of, or challenging Merger Partner's ownership of or rights in or to, any Merger Partner Intellectual Property, and Merger Partner is not aware of any facts or circumstances that would render any Merger Partner Intellectual Property invalid or unenforceable by Merger Partner or one of its Subsidiaries, or of inadequate scope to protect the interests of Merger Partner in conducting its business, except in each case, as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect; (v) since January 1, 2020 to the date of this Agreement, to the Knowledge of Merger Partner, there has been no actual, or threatened action, suit, proceeding or claim by a third party alleging that Merger Partner or any of its Subsidiaries infringes, misappropriates, or otherwise violates, any Intellectual Property of third parties, and Merger Partner has not received any notice alleging, or is otherwise aware of, any facts or circumstances that would give rise to such an action, proceeding or claim, except, in each case, where such infringement, misappropriation or other violation would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect; (vi) Merger Partner has taken all reasonable steps to protect, maintain and safeguard the Merger Partner Intellectual Property, including the execution of appropriate nondisclosure, confidentiality agreements and invention assignment agreements and invention assignments with their directors, officers, employees, contractors and other agents and since January 1, 2020 to the date of this Agreement, to the Knowledge of Merger Partner, no such individual is in or has been in violation of any term of such nondisclosure, confidentiality agreements and invention assignment agreements; and (vii) since January 1, 2020 to the date of this Agreement, to the Knowledge of Merger Partner, no material technology employed by Merger Partner or its Subsidiaries has been obtained or is being used by Merger Partner or its Subsidiaries in violation

of any contractual or legal obligation binding on Merger Partner, its Subsidiaries or any of its officers, directors or employees, which violation relates to the breach of a confidentiality obligation, obligation to assign Intellectual Property to a previous employer or obligation otherwise not to use the Intellectual Property of a third party.

(c) To the Knowledge of Merger Partner, the material information technology systems of the Merger Partner and its Subsidiaries are adequate for their respective businesses as currently conducted in all material respects.

3.11 Contracts.

- (a) Section 3.11(a) of Merger Partner Disclosure Schedule lists the following agreements (each a "Merger Partner Contract") to which Merger Partner or any Subsidiary of Merger Partner is a party as of the date of this Agreement which contains any remaining material rights or obligations:
- (i) any agreement (or group of related agreements) (A) which involves expected payments in an amount in excess of \$500,000 in the aggregate, (B) in which Merger Partner or any of its Subsidiaries has granted "most favored nation" pricing provisions or exclusive marketing or distribution rights or has agreed to purchase goods or services exclusively from a certain party, (C) which limits the freedom of Merger Partner or its Subsidiaries to engage in any line of business or compete with any Person or (D) which contains any non-solicitation provision, in each case, except for restrictions that would not materially affect the ability of Merger Partner and its Subsidiaries to conduct its business, in each case, that is not terminable by Merger Partner or its Subsidiaries, as applicable, on 90 days' notice or less;
- (ii) any agreement relating to the disposition or acquisition of material assets or any ownership interests in any Person, in each case, involving payments in excess of \$500,000, other than agreements in which the applicable acquisition or disposition has been consummated and there are no material ongoing liabilities or obligations;
 - (iii) any agreement providing for any material royalty, milestone or similar payments by Merger Partner or its Subsidiaries;
 - (iv) any agreement concerning the establishment or operation of a joint venture with a third party;
- (v) any agreement (or group of related agreements) under which Merger Partner or any Subsidiary has created, incurred, assumed or guaranteed (or may create, incur, assume or guarantee) material Indebtedness (including capitalized lease obligations) or under which it has imposed (or may impose) a Lien (other than Permitted Liens) on any of its assets, tangible or intangible;
- (vi) any agreement for the disposition of any material assets or business of Merger Partner or any Subsidiary or any agreement for the acquisition of material assets or business of any other Person (other than purchases of inventory or components in the Ordinary Course of Business);

- (vii) any noncompetition agreement restricting Merger Partner or any of its Subsidiaries from competing in any line of business or jurisdiction;
- (viii) any written employment agreement or consulting agreement that is not on Merger Partner's standard form of employment agreement or consulting agreement, each of which has been previously made available to Public Company (excluding employment agreements that are terminable "at will" without the payment of severance or other amounts upon termination, and consulting agreements which are terminable on 30 days or less notice without the payment of additional consideration);
 - (ix) any agreement providing for severance, retention, change in control payments, or transaction-based bonuses or incentives;
- (x) any material settlement agreement or agreement that includes the admission of wrongdoing by Merger Partner or any of its Subsidiaries or any of their respective officers or directors;
- (xi) any agreement with respect to an interest rate, currency or other swap or derivative transaction (other than those solely between Merger Partner or any of its Subsidiaries);
- (xii) any agreement between Merger Partner or any of its Subsidiaries, on the one hand, and any director, officer or employee of Merger Partner or any of its Subsidiaries or any Person beneficially owning five percent or more of the outstanding equity interests of Merger Partner or any of their respective Affiliates, on the other hand;
- (xiii) except as set forth in the Organizational Documents of Merger Partner or in Merger Partner's standard form of restricted unit grant agreement, any agreement that contains a put, call or similar right pursuant to which Merger Partner or any of its Subsidiaries or after the Merger Partner Merger Effective Time, Public Company or any of its Subsidiaries or Affiliates, could be required to purchase or sell, as applicable, in a single transaction or series of transactions, (A) any equity interests of any Person or (B) any asset or group of related assets that have a fair market value or purchase price of more than \$500,000 individually or in the aggregate;
- (xiv) any agreement relating to royalty, dividend or similar arrangement to be paid, or received, by Merger Partner or any of its Subsidiaries that is based on the revenue or profits of Merger Partner or any of its Subsidiaries or any material agreement involving fixed price or fixed volume arrangements;
- (xv) any agreement under which the consequences of a default or termination would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect;
- (xvi) any agreement with expected annual receipts or payables of more than \$500,000 annually, under which Merger Partner or any of its Affiliates or Subsidiaries has (A) granted a license, sublicense, covenant not to sue or assert, immunity from suit, right to coexist or other right under any Intellectual Property to any other Person; (B) received from any

other Person a grant of a license, sublicense, covenant not to sue or assert, immunity from suit, right to coexist or other right under any Intellectual Property (other than licenses to any shrink wrap, click wrap or other software that is generally commercially available and not customized costing less than \$50,000 per year or as a one-time fee); (C) assigned any rights under any Intellectual Property to any other Person; or (D) received an assignment of any rights under Intellectual Property from any Person (other than assignments from employees on the form assignment agreement used by Merger Partner with such employees);

(xvii) any agreement the primary purpose of which is to provide for the sharing, indemnification or allocation of Taxes between or among the Merger Partner and its Subsidiaries, or between or among the Subsidiaries of Merger Partner; and

(xviii) any other agreement (or group of related agreements) that is material to Merger Partner and its Subsidiaries, taken as a whole, including any material contacts (as defined in Item 601(b)(10) of Regulation S-K as promulgated under the Securities Act).

- (b) Merger Partner has made available to Public Company a complete and accurate copy of each Merger Partner Contract (as amended to date) as of the date of this Agreement. With respect to each Merger Partner Contract: (i) the Merger Partner Contract is legal, valid, binding and enforceable and in full force and effect against Merger Partner or the Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Merger Partner, against each other party thereto; (ii) the Merger Partner Contract will continue to be legal, valid, binding and enforceable and in full force and effect against Merger Partner or the Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Merger Partner, against each other party thereto immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing; and (iii) neither Merger Partner, any Affiliate or Subsidiary nor, to the Knowledge of Merger Partner, any other party, is in breach or violation of, or default under, any such Merger Partner Contract, and no event has occurred, is pending or, to the Knowledge of Merger Partner, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute any such breach or default by Merger Partner, any Subsidiary or, to the Knowledge of Merger Partner, any other party under such Merger Partner Contract, in each case, where any such breach or violation would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect.
- 3.12 Litigation. As of the date hereof, there is no Legal Proceeding pending or, to the Knowledge of Merger Partner, threatened with respect to, (A) against or affecting (i) Merger Partner or any Subsidiary of Merger Partner, (ii) any current or former officer, director, employee, consultant, agent or member of any of the foregoing in its, his or her capacity as such, (iii) any of the material assets owned or used by Merger Partner and its Subsidiaries or (iv) to the Knowledge of Merger Partner, any Blocker and (B) no notice of any Legal Proceeding involving or relating to Merger Partner, any of its Subsidiaries, or any Blocker, whether pending or threatened, has been received by Merger Partner, any of its Subsidiaries, or to the Knowledge of Merger Partner any Blocker, in each case as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect, a Public Company Material Adverse Effect (after giving effect to the Closing) or that would have the effect of preventing, delaying beyond the Outside Date, or making illegal, the Transactions (including the Mergers). As of the date of this Agreement, there are no material judgments, orders, injunctions,

decrees, stipulations or awards (whether rendered by a court, administrative agency or other Governmental Entity, by arbitration or otherwise) against or involving Merger Partner or any Subsidiary of Merger Partner or, to the Knowledge of Merger Partner, any Blocker. As of the date of this Agreement, there is no material Legal Proceeding by Merger Partner or any Subsidiary of Merger Partner pending, or which Merger Partner, any Subsidiary of Merger Partner or, to the Knowledge of Merger Partner, any Blocker has commenced preparations to initiate, against any other Person.

3.13 Environmental Matters.

- (a) Since January 1, 2020, Merger Partner and its Subsidiaries have complied with all applicable Environmental Laws except as would not, individually or in the aggregate, reasonably be expected to result in a Merger Partner Material Adverse Effect. There is no pending or, to the Knowledge of Merger Partner, threatened material Legal Proceeding relating to any Environmental Law involving Merger Partner or any Subsidiary of Merger Partner.
- (b) To the Knowledge of Merger Partner, neither Merger Partner nor any Subsidiary of Merger Partner has any material liabilities or material obligations arising from the release or threatened release of any Materials of Environmental Concern into the environment.
- (c) Neither Merger Partner nor any Subsidiary of Merger Partner is a party to or bound by any court order, administrative order, consent order or other agreement between Merger Partner or any Subsidiary and any Governmental Entity entered into in connection with any material legal obligation or material liability arising under any Environmental Law.
- (d) Merger Partner has no Knowledge of any material environmental liability relating to any solid or hazardous waste transporter or treatment, storage or disposal facility that has been used by Merger Partner or any Subsidiary of Merger Partner.

3.14 Labor and Employment.

- (a) Each such current Merger Partner Employee is retained at-will and none of such current Merger Partner Employees is a party to a written employment agreement or contract with Merger Partner or any Subsidiary contradicting their at-will employment. Each current Merger Partner Employee has entered into Merger Partner's or such Subsidiary's standard form of confidentiality, non-competition and assignment of inventions agreement, a copy of which has previously been delivered to Public Company. All of the agreements referenced in the preceding sentence will continue to be in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing.
- (b) Since January 1, 2020, neither Merger Partner nor any Subsidiary of Merger Partner has breached or violated any (i) applicable Law respecting employment and employment practices, terms and conditions of employment and wages and hours, including any such Law respecting employment discrimination, employee classification (for overtime purposes or as employee versus independent contractor), workers' compensation, family and medical leave, the Immigration Reform and Control Act and occupational safety and health requirements, or (ii) employment or other individual service provider agreement, in each case except as would

not, individually or in the aggregate, reasonably be expected to be material to Merger Partner and its Subsidiaries, taken as a whole. As of the date hereof, no material claims, controversies, investigations, audits or other Legal Proceedings are pending or, to the Knowledge of Merger Partner, threatened, with respect to such Laws or agreements, either by private Persons or by Governmental Entities.

- (c) Neither Merger Partner nor any Subsidiary of Merger Partner is a party to or bound by any collective bargaining agreement, nor has either of them experienced any actual or, to the Knowledge of Merger Partner, threatened strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. Merger Partner has no Knowledge of any material organizational effort made or threatened (including the filing of a petition for certification) either currently or within the past two (2) years, by or on behalf of any labor union with respect to Merger Partner Employees.
- (d) Each such consultant or independent contractor is or was a party to a written agreement or contract with Merger Partner or its Subsidiaries. Merger Partner and its Subsidiaries have not incurred, and, to the Knowledge of Merger Partner, no circumstances exist under which, Merger Partner and its Subsidiaries could reasonably be expected to incur, any material liability arising from the misclassification of employees as consultants or independent contractors, or from the misclassification of consultants or independent contractor has entered a confidentiality and assignment of inventions agreement with Merger Partner, a copy of which has previously been made available to Public Company.
- (f) As of the date hereof, no charges or complaints are open and pending against Merger Partner or any Subsidiary with the Equal Employment Opportunity Commission, the Office of Federal Contract Compliance Programs (the "OFCCP"), or similar Governmental Entity or pursuant to internal complaint procedures, and, to the Knowledge of Merger Partner, as of the date hereof no current or former employee of Merger Partner or any Subsidiary has made, during the last twelve (12) months, an oral or, during the last three (3) years, a written complaint of discrimination, retaliation or other similar wrongdoing. True, correct and complete information regarding any closed charges or complaints filed since December 31, 2020 through the date of this Agreement with the Equal Employment Opportunity Commission, the OFCCP or similar Governmental Entity (or, with respect to discrimination, retaliation, or similar wrongdoing, pursuant to internal complaint procedures) has been made available to Public Company.
- (g) Since January 1, 2020, neither Merger Partner nor any Subsidiary has caused (i) a plant closing as defined in the Worker Adjustment and Retraining Notification Act (the "WARN Act") affecting any site of employment or one or more operating units within any site of employment of Merger Partner or any Subsidiary or (ii) a mass layoff as defined in the WARN Act, nor has Merger Partner or any Subsidiary been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar foreign, state or local Law.

3.15 Employee Benefit Plans.

- (a) Section 3.15(a) of Merger Partner Disclosure Schedule contains a complete and accurate list of all material Merger Partner Benefit Plans. Complete and accurate copies of (i) all Merger Partner Benefit Plans which have been reduced to writing, together with all amendments thereto, (ii) written summaries of all material unwritten Merger Partner Benefit Plans, (iii) all related trust agreements, insurance contracts, and most recent summary plan description together with the summary or summaries of material modifications thereto, (iv) the most recently filed annual report on IRS Form 5500 and all schedules thereto and (for all funded plans) all plan financial statements for the last plan year for each Merger Partner Benefit Plan, (v) all reports regarding the satisfaction of the nondiscrimination requirements of Sections 410(b), 401(k), and 401(m) of the Code for the past year, (vi) the most recent determination letter or opinion letter issued by the Internal Revenue Service with respect to any Merger Partner Benefit Plan intended to be qualified under Section 401(a) of the Code, and (vii) any material written or electronic communications from or to the Internal Revenue Service, the Department of Labor ("DOL") or any other Governmental Entity received in the three (3) year period preceding the date of this Agreement with respect to a Merger Partner Benefit Plan (including any voluntary correction submissions), have been delivered or made available to Public Company. No Merger Partner Benefit Plan is or has been subject to non-U.S. Law.
- (b) Neither Merger Partner, any of its Subsidiaries, nor any of their respective ERISA Affiliates sponsors, maintains or contributes to or in the last six years has sponsored, maintained, contributed to or been required to contribute to, or has (or is reasonably expected to have) any material current or contingent liability with respect to any (i) "employee pension benefit plan," as defined in Section 3(2) of ERISA, that is subject to Title IV or Section 302 of ERISA or Section 412 of the Code, (ii) a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA, or (iii) a "multiple employer plan" subject to Sections 4063 or 4064 of ERISA or Section 413(c) of the Code. No liability under Section 302 or Title IV of ERISA or Section 412 of the Code has been incurred by Merger Partner, any of its Subsidiaries or any of their respective ERISA Affiliates that has not been satisfied in full.
- (c) Each Merger Partner Benefit Plan has been administered in all material respects in accordance with its terms and each of Merger Partner, its Subsidiaries and its ERISA Affiliates has met its obligations in all material respects with respect to each Merger Partner Benefit Plan and has timely made all required contributions, premiums and other payments thereto, in each case in all material respects. Merger Partner, its Subsidiaries, each of its ERISA Affiliates and each Merger Partner Benefit Plan are in compliance in all material respects with the currently applicable provisions of ERISA and the Code (including Section 409A of the Code) and the regulations thereunder. Each Merger Partner Benefit Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the IRS to be qualified under Section 401(a) of the Code and, no fact or event has occurred that would adversely affect the qualification or tax exemption of any such Merger Partner Benefit Plan. There is no legally binding commitment to create any additional Merger Partner Benefit Plans or to modify any existing Merger Partner Benefit Plans.

- (d) There are no Legal Proceedings (except claims for benefits payable in the normal operation of Merger Partner Plans and proceedings with respect to qualified domestic relations orders) against or involving any Merger Partner Benefit Plan or asserting any rights or claims to benefits under any Merger Partner Benefit Plan that could give rise to any material liability. No Merger Partner Benefit Plan is or within the last three (3) calendar years has been the subject of, or has received or provided notice that it is the subject of, examination by a Governmental Entity or a participant in a government sponsored amnesty, voluntary compliance or similar program.
- (e) The execution and delivery of this Agreement and the consummation of the Transactions will not, either alone or in combination with another event, (A) entitle any current or former employee, director, officer or independent contractor of Merger Partner or any of its Subsidiaries to severance pay, retention bonuses, non-competition payments or any other material compensatory payment, (B) accelerate the time of payment or vesting, or increase the amount of compensation due to any such employee, director, officer or independent contractor, or (C) directly or indirectly require Merger Partner to transfer or set aside any assets to fund any material benefits under any Merger Partner Benefit Plan. No director, officer, employee or other service provider of Merger Partner or any of its Subsidiaries is entitled to a gross up, make whole or other similar payment as a result of the imposition of Taxes under Section 4999 or Section 409A of the Code pursuant to any agreement or arrangement with the Merger Partner or any of its Subsidiaries.
- 3.16 Compliance with Laws. Each of Merger Partner, its Subsidiaries has since January 1, 2020 conducted, and is conducting, its business and operations in compliance in all material respects with all applicable Laws. Since January 1, 2020, neither Merger Partner nor any Subsidiary has received any notice or other communication from any Governmental Entity or other Person alleging any material noncompliance with any applicable Law. To the Knowledge of Merger Partner, neither Merger Partner nor any Subsidiary has any material liability for failure to comply with any Law and, to the Knowledge of Merger Partner, there is no act, omission, event or circumstance that would reasonably be expected to give rise to any such liability. Neither Merger Partner nor any Subsidiary has conducted any internal investigation with respect to any actual, potential or alleged violation of any Law by any manager, member or other equity holder, officer or Merger Partner Employee or concerning any actual or alleged Fraud.
- 3.17 <u>Unlawful Payments</u>. Since January 1, 2020, Merger Partner and its Subsidiaries are and have been in compliance in all material respects with the Foreign Corrupt Practices Act, 15 U.S.C. §§ 78dd-1, et seq., the Organization for Economic Cooperation and Development Convention Against Bribery of Foreign Public Officials in International Business Transactions and legislation implementing such convention, all other international anti-bribery conventions and all applicable anti-corruption or bribery Laws in any jurisdiction in which Merger Partner or any Subsidiary has conducted its business (collectively, "<u>Anti-Bribery Laws</u>"). Since January 1, 2020, neither Merger Partner nor any Subsidiary has received any written communication from any Governmental Entity that alleges that Merger Partner or any Subsidiary, or any current or former Representatives thereof, is or may be in violation of, or has, or may have, any liability under, any Anti-Bribery Laws, and no such actual or potential violation of Anti-Bribery Laws has been discovered by or brought to the attention of Merger Partner or any Subsidiary since January 1, 2020, neither Merger Partner nor any Subsidiary has made or anticipates making any disclosures to any Governmental Entity for potential violations of Anti-

Bribery Laws. To the Knowledge of Merger Partner, none of Merger Partner and its Subsidiaries' current or former Representatives is currently an officer, agent or employee of a Governmental Entity. Neither Merger Partner nor any Subsidiary nor any of their respective current or former Representatives has directly or indirectly offered, given, reimbursed, paid or promised to pay, or authorized the payment of, any money or other thing of value (including any fee, gift, sample, travel expense or entertainment) or any commission payment payable to (a) any Person who is an official, officer, agent, employee or representative of any Governmental Entity or of any existing or prospective customer (whether or not owned by a Governmental Entity), (b) any political party or official thereof or (c) any candidate for political or political party office, in each case while knowing or having reason to believe that all or any portion of such money or thing of value would be offered, given, reimbursed, paid or promised, directly or indirectly, for purposes not allowable under the Anti-Bribery Laws, to any such official, officer, agent, employee, representative, political party, political party official, candidate, individual affiliated with any such customer, political party or official or political office.

3.18 Permits. Each of Merger Partner and its Subsidiaries owns or holds all material Permits that are required for Merger Partner and its Subsidiaries, respectively, to conduct their business as presently conducted or as proposed to be conducted. Each such Permit is in full force and effect. Merger Partner or the applicable Subsidiary, as the case may be, is in compliance in all material respects with the terms of each such Permit; and, as of the date of this Agreement, to the Knowledge of Merger Partner, no suspension or cancellation of such Permit is threatened and, to the Knowledge of Merger Partner, there is no basis for believing that such Permit will not be renewable upon expiration, other than in the Ordinary Course of Business. Each such Permit will continue in full force and effect immediately following the Closing. Merger Partner has made available to Public Company all such Permits as of the date of this Agreement.

3.19 Insurance.

(a) Merger Partner has made available to Public Company a true and complete list, as of the date of this Agreement, all material insurance policies (including fire, theft, casualty, comprehensive general liability, workers compensation, business interruption, environmental, director and officer liability, product liability and automobile insurance policies and bond and surety arrangements) to which Merger Partner or any Subsidiary is a party, a named insured or otherwise the beneficiary of coverage (the "Merger Partner Insurance Policies"), all of which are in full force and effect. The Merger Partner Insurance Policies are of the type and in the amounts customarily carried by organizations conducting businesses or owning assets similar to those of Merger Partner. There is no material claim pending under the Merger Partner Insurance Policies as to which coverage has been questioned, denied or disputed by the underwriter of such policy. All premiums due and payable under all of the Merger Partner Insurance Policies have been paid, and, to the Knowledge of Merger Partner, neither Merger Partner nor any Subsidiary may be liable for retroactive premiums or similar payments, and Merger Partner and its Subsidiaries are otherwise in compliance with the terms of the Merger Partner Insurance Policies in all material respects. Merger Partner has no Knowledge of any threatened termination of, or premium increase with respect to, any of the Merger Partner Insurances Policies. Each Merger Partner Insurance Policy will continue to be enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing. There is no material claim by Merger Partner or any of its Subsidiaries pending under any Merger Partner Insurance Policy in favor of Merger Partner and its Subsidiaries that has been denied or disputed by the insurer other than denials and disputes in the Ordinary Course of Business.

(b) Section 3.19(b) of Merger Partner Disclosure Schedule identifies all claims asserted by Merger Partner pursuant to any Merger Partner Insurance Policy since December 31, 2019 and describes the nature and status of each such claim.

3.20 Cybersecurity and Privacy.

- (a) Each of Merger Partner and its Subsidiaries comply, and have complied, in all respects with all: (i) applicable Privacy Laws; (ii) policies, statements, and contractual obligations relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security, disposal, destruction, disclosure, or transfer of Personal Information of the Merger Partner and its Subsidiaries; and (iii) all applicable industry standards including, without limitation, PCI DSS, and all other applicable requirements of the payment card brands (clauses (i)-(iii) of this Section 3.20(a)) collectively, the "Merger Partner Privacy Requirements"). The consummation of any of the Transactions will not violate any applicable Merger Partner Privacy Requirements.
- (b) Merger Partner and its Subsidiaries have taken all organizational, physical, administrative, and technical measures required by the Merger Partner Privacy Requirements and consistent with standards prudent in the industry in which Merger Partner and its Subsidiaries operate to protect: (i) the integrity, security, and operations of all Merger Partner IT Systems; and (ii) all Personal Information and all other data owned, controlled, or stored by Merger Partner and its Subsidiaries from and against data security incidents or other misuse. In connection with each third party servicing, outsourcing, hosting or otherwise processing Personal Information collected, held, or processed by or on behalf of Merger Partner and its Subsidiaries, Merger Partner and its Subsidiaries has in accordance with the Merger Partner Privacy Requirements entered into valid, binding and enforceable written data processing agreements with any such third party that ensures appropriate protection for Personal Information. None of Merger Partner and its Subsidiaries sells, rents or otherwise makes available, or permits any third party to, sell, rent or otherwise make available, to any Person any Personal Information, except as stated in the applicable privacy policies and in compliance with Privacy Laws.
- (c) There have been no actual or suspected data security incidents, data breaches, ransomware incidents, or other adverse events or incidents related to any Merger Partner IT Systems, Personal Information, or data in the custody or control of Merger Partner or its Subsidiaries or any third party acting on behalf of Merger Partner or its Subsidiaries. Merger Partner and its Subsidiaries have not been subject to any claims or proceedings related to any data security incidents, ransomware incidents, or any violations of any Merger Partner Privacy Requirements, and there are no facts or circumstances which could reasonably serve as the basis for any such allegations or claims. Merger Partner and its Subsidiaries have taken commercially reasonable steps to provide for archival, back-up, recovery and restoration of its material business data.

- (d) Merger Partner and its Subsidiaries own or have a valid right to access and use all Merger Partner IT Systems. To the Knowledge of Merger Partner, the Merger Partner IT Systems do not contain any viruses, worms, Trojan horses, bugs, faults or other devices, errors, contaminants or effects that are designed to (A) disrupt or adversely affect the functionality of any Merger Partner IT Systems or (B) enable or assist any Person to access without authorization any Merger Partner IT Systems. The Merger Partner IT Systems have not materially malfunctioned or failed within the past two years. Merger Partner and its Subsidiaries have taken reasonable, and to ensure the continued, uninterrupted, and error-free operation of the Merger Partner IT Systems. No Person has gained unauthorized access to any Merger Partner IT Systems.
- 3.21 <u>Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission based upon arrangements made by or on behalf of Merger Partner nor any Subsidiary in respect of any of the Transactions.
- 3.22 <u>Blocker Organization, Standing and Power.</u> Such Blocker is either a corporation or a limited liability company, as applicable, duly organized, validly existing and in good standing under the Laws of the State of Delaware. Except as would not, individually or in the aggregate, reasonably be expected to have the effect of preventing, delaying beyond the Outside Date, or making illegal, the Transactions (including the Mergers), such Blocker is duly qualified to conduct business and is in good standing under the Laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification. Such Blocker has all requisite power and authority to (i) carry on the businesses in which it is engaged, (ii) own and use the properties owned and used by it and (iii) perform its obligations under all Contracts to which it is a party. Merger Partner has made available to Public Company complete and accurate copies of the Organizational Documents of such Blocker as in effect on the date of this Agreement. Such Blocker is not in material default under or in material violation of any provision of its Organizational Documents.

3.23 Blocker Capitalization.

- (a) Section 3.23(a) of the Merger Partner Disclosure Schedule sets forth the authorized capital of such Blocker and, as of the date of this Agreement, the issued and outstanding equity interests of such Blocker.
- (b) Section 3.23(b) of Merger Partner Disclosure Schedule sets forth a complete and accurate list, as of the date of the Agreement, of the members and holders of limited liability company interests, or the stockholders, as applicable, of such Blocker, showing the number and class(es) held by each holder. Section 3.23(b) of Merger Partner Disclosure Schedule also indicates all equity securities that are subject to a threshold value, vesting or otherwise subject to a repurchase or redemption right, indicating the name of the applicable member or interest holder, the threshold value, vesting schedule (including any acceleration provisions with respect thereto), and the repurchase price payable by applicable Blocker. All of the equity securities of such Blocker have been duly authorized, validly issued, and free of all preemptive rights. To the actual knowledge of such Blocker, all of the equity securities of such Blocker have been offered, issued and sold by such Blocker in compliance in all material respects with all applicable federal and state securities Laws.

- (c) Except as set forth in Section 3.23(b), (A) there are no equity securities of any class of such Blocker, or any security exchangeable into or exercisable for such equity securities, issued, reserved for issuance or outstanding, (B) there are no options, warrants, securities, calls, rights, commitments or agreements to which such Blocker is a party or by which such Blocker is bound obligating such Blocker to issue, exchange, transfer, deliver or sell, or cause to be issued, exchanged, transferred, delivered or sold, additional equity securities of such Blocker or any security or rights convertible into or exchangeable or exercisable for any such equity securities, or obligating such Blocker to grant, extend, accelerate the vesting of, otherwise modify or amend or enter into any such option, warrant, equity security, call, right, commitment or agreement, and (C) such Blocker has no obligation (contingent or otherwise) to purchase, redeem or otherwise acquire any of its equity securities or to make any distribution in respect thereof.
- (d) Except for the Organizational Documents of such Blocker or as set forth on Section 3.23(d) of the Merger Partner Disclosure Schedule, there is no agreement, written or, to the actual knowledge of such Blocker, oral, between such Blocker and any holder of its equity securities or, to the actual knowledge of such Blocker, among any holders of the equity securities of such Blocker, relating to the sale or transfer (including agreements relating to rights of first refusal, co-sale rights or "drag along" rights), registration under the Securities Act or the securities Laws of any other jurisdiction, or voting, of the equity securities of such Blocker.
- 3.24 No Business. Except for the ownership of limited liability company interests in Merger Partner, such Blocker (i) has not conducted any business and (ii) does not have and has not had any employees, assets, or liabilities (contingent or otherwise) other than reasonable reserves of cash or other liquid amounts as may be necessary or reasonably appropriate in order to provide for ongoing corporate maintenance of such Blocker.

3.25 Blocker Authority; No Conflict; Required Filings and Consents.

(a) Such Blocker has all requisite power and authority to execute and deliver this Agreement and the other agreements contemplated hereby and to perform its respective obligations hereunder and thereunder. The execution and delivery by such Blocker of this Agreement and the other agreements contemplated hereby and, subject to obtaining the Blocker Approvals, which are the only approval required from the securityholders of such Blocker, the performance by such Blocker of this Agreement and the consummation by such Blocker of the Transactions have been duly and validly authorized by all necessary company or other corporate action on the part of such Blocker. This Agreement and all other agreements contemplated hereby have been duly and validly executed and delivered by such Blocker party thereto and, assuming the due execution and delivery hereof and thereof by the other parties hereto and thereto, constitutes or will constitute a valid and binding obligation of such Blocker, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, and other similar Laws affecting creditors' rights generally and by general principles of equity. The manager or managing member, as applicable, of each Blocker has (i) determined that the Transactions are fair to, advisable, and in the best interests of such Blocker and its respective members and (ii) adopted this Agreement and approved the Transactions, including the applicable Blocker Merger, in its capacity as a manager or managing member, as applicable, of such Blocker.

- (b) Subject to the filing of the Blocker Certificate of Merger for such Blocker Merger as required by Delaware Law, neither the execution and delivery by such Blocker of this Agreement or any other agreement contemplated hereby, nor the performance by such Blocker of its obligations hereunder or thereunder, nor the consummation by such Blocker of the Transactions, will (i) conflict with or violate any provision of the Organizational Documents of such Blocker, each as amended or restated to date, (ii) require on the part of such Blocker any notice to or filing with, or any permit, authorization, consent or approval of, any Governmental Entity, (iii) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, any contract, lease, sublease, license, sublicense, franchise, permit, indenture, agreement or mortgage for borrowed money, instrument of Indebtedness, Lien (other than Permitted Liens) or other arrangement to which such Blocker is a party or by which such Blocker is bound or to which any of the assets of such Blocker are subject, (iv) result in the imposition of any Lien other than Permitted Liens upon any assets of such Blocker or (v) violate any order, writ, injunction, decree, statute, rule or regulation applicable to such Blocker or any of its properties or assets, except in the case of clauses (ii) through (v) as would not, individually or in the aggregate, reasonably be expected to have the effect of preventing, delaying beyond the Outside Date, or making illegal, the Transactions (including the Mergers).
- 3.26 <u>Blocker Compliance with Laws</u>. To the actual knowledge of such Blocker, such Blocker, has since January 1, 2020 conducted, and is conducting, its business and operations in compliance in all material respects with all applicable Laws. To the actual knowledge of such Blocker, since January 1, 2020, such Blocker has not received any notice or other communication from any Governmental Entity or other Person alleging any material noncompliance with any applicable Law. To the actual knowledge of such Blocker, such Blocker does not have any material liability for failure to comply with any Law and, to the actual knowledge of such Blocker, there is no act, omission, event or circumstance that would reasonably be expected to give rise to any such liability.
- 3.27 <u>Blocker Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission based upon arrangements made by or on behalf of such Blocker with respect to any of the Transactions.

3.28 Blocker Tax Matters.

- (a) Each Blocker has properly filed on a timely basis all income and other material Tax Returns that it was required to file, and all such Tax Returns were true, correct and complete in all material respects. Each Blocker has paid on a timely basis all income or other material Taxes, whether or not shown on any Tax Return, that were due and payable.
- (b) All material Taxes that each Blocker was required by Law to withhold or collect have been duly withheld or collected and, to the extent required, have been properly paid to the appropriate Governmental Entity, in each case in compliance in all material respects with applicable Law.

- (c) No examination or audit of any Tax Return of a Blocker by any Governmental Entity is currently in progress or, to the Knowledge of such Blocker, has been threatened by any Governmental Entity. No deficiencies for Taxes of any Blocker have been claimed, proposed or assessed by any Governmental Entity in writing. No Blocker has been informed in writing by any jurisdiction in which such Blocker does not file a Tax Return that the jurisdiction believes that such Blocker was required to file any Tax Return that was not filed or is subject to Tax in such jurisdiction. No Blocker has (i) waived any statute of limitations with respect to Taxes or agreed to extend the period for assessment or collection of any Taxes, which waiver or extension is still in effect, (ii) requested any extension of time within which to file any Tax Return (other than any automatic extension granted in the ordinary course of business and consistent with past custom and practice of such Blocker), or (iii) executed or filed any power of attorney with any taxing authority, which is still in effect.
- (d) To the actual knowledge of such Blocker, such Blocker has not participated in any "reportable transaction" as defined in Treasury Regulations Section 1.6011-4(b) or a "listed transaction" within the meaning of Treasury Regulations Section 1.6011-4(b)(2) or any analogous provision of state or local Law.
 - (e) There are no Liens with respect to Taxes on any of the assets of any Blocker, other than Permitted Liens.
- (f) With the exception of customary commercial leases or contracts, if any, that are not primarily related to Taxes entered into in the Ordinary Course of Business and liabilities thereunder, no Blocker (i) has any liability under Treasury Regulations Section 1.1502-6 (or any comparable or similar provision of state, local or non-U.S. law), as a transferee or successor, or pursuant to any contractual obligation for any Taxes of any Person, or (ii) is a party to or bound by any Tax indemnity, Tax sharing, Tax allocation or similar agreement.
- (g) During the two-year period ending on the date hereof, no Blocker (including any predecessor thereof) has constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).
 - (h) Each Blocker is, and has been at all times since its inception, classified for U.S. federal and state income Tax purposes as a corporation.
- 3.29 No Other Representations and Warranties. Except as previously set forth in this Article III or in any certificate delivered by Merger Partner to Public Company and/or Merger Subsidiary pursuant to this Agreement or otherwise in the case of Fraud, Merger Partner and Blockers make no representation or warranty, express or implied, at law or in equity, with respect to it or any of its assets, liabilities or operations, and any such other representations or warranties are hereby expressly disclaimed.
- 3.30 <u>Reliance</u>. Each of Merger Partner and each Blocker acknowledges that, except for the representations and warranties contained in Article IV, neither Public Company nor any other Person on behalf of Public Company has made, and each of Merger Partner and each Blocker (as to itself) has not relied on, any other express or implied representation or warranty by or on behalf of Public Company or such Person.

ARTICLE IV REPRESENTATIONS AND WARRANTIES OF PUBLIC COMPANY AND MERGER SUBSIDIARY

Each of Public Company and Merger Subsidiary represents and warrants to Merger Partner and the Blockers, as of the date of this Agreement and as of the Closing Date, that, except (a) as set forth in Public Company Disclosure Schedule or (b) as disclosed in Public Company SEC Reports filed since January 1, 2020 that were publicly available on EDGAR at least one (1) Business Day prior to the date of this Agreement (but (i) without giving effect to any amendment thereof filed with, or furnished to the SEC on or after the Business Day prior to the date of this Agreement and (ii) excluding any risk factor or similar disclosure under the headings "Risk Factors", "Forward-Looking Statements" or other similar cautionary, predictive or forward-looking disclosures contained therein), it being understood and agreed that any matter disclosed in such Public Company SEC Reports shall be deemed to be disclosed in a section of the Public Company Disclosure Schedule only to the extent to which its relevance is reasonably apparent on the face of the disclosure in such Public Company SEC Reports the statements contained in this Article IV are true and correct as of the date of this Agreement, except to the extent such representations and warranties are specifically made as of a particular date (in which case such representations and warranties will be true and correct as of such date). The Public Company Disclosure Schedule shall be arranged in sections and paragraphs corresponding to the numbered and lettered sections and paragraphs contained in this Article IV; provided, however, that the disclosures in any section or paragraph in this Article IV and (b) such other sections or paragraphs in this Article IV (whether or not there is a specific cross reference) to the extent that it is reasonably apparent on the face of the disclosure that such disclosure also qualifies or applies to such other section or paragraph.

4.1 <u>Organization, Standing and Power</u>. Each of Public Company and Merger Subsidiary is a corporation or limited liability company and duly organized, validly existing and in good standing under the Laws of the State of Delaware. Except as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect, each of Public Company and Merger Subsidiary is duly qualified to conduct business and is in good standing under the Laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification. Each of Public Company and Merger Subsidiary has all requisite power and authority to (i) carry on the businesses in which it is engaged, (ii) own and use the properties owned and used by it and (iii) perform its obligations under all Public Company Contracts. Public Company has made available to Merger Partner complete and accurate copies of the Organizational Documents of Public Company and Merger Subsidiary as in effect on the date of this Agreement. Each of Public Company and Merger Subsidiary is not in material default under or in material violation of any provision of its Organizational Documents.

4.2 Capitalization.

(a) The authorized capital stock of Public Company consists of 275,000,000 shares of Public Company Common Stock and 50,000,000 shares of Public Company Preferred Stock, \$0.0001 par value per share ("Public Company Preferred Stock"). As of December 30, 2022 (the "Measurement Date"), there were (a) 48,844,619 shares of Public Company Common Stock outstanding, (b) no shares of Public Company Preferred Stock outstanding, (c) 100,152 shares of Public Company Common Stock held in treasury and (d) no shares of Public Company Common Stock are owned by any Subsidiary of Public Company. Since the Measurement Date through the date of this Agreement, Public Company has not issued any shares of Public Company Common Stock or Public Company Preferred Stock except for issuances of shares of Public Company Common Stock upon the exercise of Public Company Stock Options, Public Company RSUs, Public Company PSUs, or Public Company Warrants that were outstanding at such time.

(b) As of December 30, 2022, there were outstanding options to purchase 5.845,733 shares of Public Company Common Stock (each, a "Public Company Stock Option" and collectively, the "Public Company Stock Options"); 1,258,447 restricted stock units with respect to 1,258,447 shares of Public Company Stock (each, a "Public Company RSU" and collectively, the "Public Company RSUs"); and 1,130,000 performance stock units with respect to 1,130,000 shares of Public Company Stock, assuming achievement of the applicable performance measures at the maximum level, (each, a "Public Company PSU" and collectively, the "Public Company PSUs"). Public Company has made available to Merger Partner complete and accurate copies of all stock or equity related plans, agreements, or arrangements of Public Company (collectively, the "Public Company Stock Plans") and the forms of all award agreements evidencing awards under the Public Company Stock Plans. Section 4.2(b) of the Public Company Disclosure Schedule contains a correct and complete list, as of December 30, 2022, of each award issued under the Public Company Stock Plans, including the date of grant, number of Public Company Common Shares, and, where applicable, outstanding dividend equivalent rights, exercise price, whether such award is intended to be an incentive stock option, and vesting schedule. As of December 30, 2022, Public Company had reserved 11,313,227 shares of Public Company Common Stock for issuance to employees pursuant to the Public Company Stock Plans, of which 2,718,149 shares remain available for issuance thereunder as of the date thereof. Public Company has not granted, issued or authorized the grant or issuance of any Public Company Stock Options on the Business Day prior to the date of this Agreement or on the date of this Agreement. With respect to each Public Company Stock Option, Public Company RSU, Public Company PSU, and any other award granted with respect to Public Company Common Stock (whether outstanding or previously exercised or settled, as applicable) (i) each such award or grant was duly authorized no later than the date on which the grant or award was by its terms to be effective by all necessary company or other corporate action, including, as applicable, approval by the Public Company Board (or a duly constituted and authorized committee thereof), or a duly authorized delegate thereof, and any required stockholder approval by the necessary number of votes or written consents, (ii) each such grant or award was made in accordance with the terms of the applicable Public Company Stock Plan, the Securities Act, the Exchange Act, to the extent applicable, and all other applicable Laws and are not and have not been the subject of any internal investigation, review or inquiry, and each Public Company Stock Option has an exercise price per Public Company Common Share equal to or greater than the fair market value of a Public Company Common Share on the date of such grant.

- (c) Section 4.2(c) of the Public Company Disclosure Schedule lists the number of shares of Public Company Common Stock reserved for future issuance pursuant to warrants or other outstanding rights (other than Public Company Stock Options, Public Company RSUs, and Public Company PSUs) to purchase shares of Public Company Common Stock outstanding as of the close of business on the Business Day prior to the date of this Agreement (such outstanding warrants or other rights, the "Public Company Warrants") and the agreement or other document under which such Public Company Warrants were granted, and the exercise price, the date of grant and the expiration date thereof.
- (d) Except as set forth in this Section 4.2 or in ARTICLE II, (A) there are no equity securities of any class of Public Company, or any security exchangeable into or exercisable for such equity securities, issued, reserved for issuance or outstanding and (B) there are no options, warrants, equity securities, calls, rights, commitments or agreements of any character to which Public Company or any of its Subsidiaries is a party or by which Public Company or any of its Subsidiaries is bound obligating Public Company or any of its Subsidiaries to issue, exchange, transfer, deliver or sell, or cause to be issued, exchanged, transferred, delivered or sold, additional shares of capital stock or other equity interests of Public Company or any security or rights convertible into or exchangeable or exercisable for any such shares or other equity interests, or obligating Public Company or any of its Subsidiaries to grant, extend, accelerate the vesting of, otherwise modify or amend or enter into any such option, warrant, equity security, call, right, commitment or agreement. Except as set forth in the Support Agreements or pursuant to any Public Company Stock Plan, Public Company is not a party to or is bound by any, and to the Knowledge of Public Company, there are no, agreements or understandings with respect to the voting (including voting trusts and proxies) or sale or transfer (including agreements imposing transfer restrictions) of any shares of capital stock or other equity interests of Public Company. Except as contemplated by this Agreement or described in this Section 4.2(d), there are no registration rights to which Public Company or any of its Subsidiaries is a party or by which it or they are bound with respect to any equity security of any class of Public Company. Stockholders of Public Company are not entitled to dissenters' or appraisal rights under applicable Delaware Law in connection with the Mergers.
- (e) All outstanding shares of Public Company Common Stock are, and all shares of Public Company Common Stock subject to issuance as specified in Sections 4.2(b) and 4.2(c) or pursuant to ARTICLE II, upon issuance on the terms and conditions specified in the instruments pursuant to which they are issuable, will be, duly authorized, validly issued, fully paid and nonassessable and not subject to or issued in violation of any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of the DGCL, Public Company's certificate of incorporation or bylaws or any agreement to which Public Company is a party or is otherwise bound.

4.3 Subsidiaries.

(a) Section 4.3 of Public Company Disclosure Schedule sets forth each Subsidiary of Public Company, all of which are wholly-owned by Public Company. Except as listed on Section 4.3 of the Public Company Disclosure Schedule, Public Company does not own or control directly or indirectly or have any direct or indirect equity participation or similar interest in, or any obligation to provide funding to, any corporation, partnership, limited liability company, joint venture, trust or other business association or entity.

- (b) Each Subsidiary of Public Company (i) is duly organized, validly existing and in standing under the Laws of the jurisdiction of its formation, (ii) is duly qualified to conduct business and is in good standing under the Laws of each jurisdiction in which the nature of its businesses or the ownership or leasing of its properties requires such qualification except as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect, and (iii) has all requisite power and authority to carry on the businesses in which it is engaged and to own and use the properties owned and used by it. Public Company has delivered to Merger Partner complete and accurate copies of the Organizational Documents of each Subsidiary of Public Company. There are no outstanding or authorized options, warrants, rights, agreements or commitments to which Public Company or any Subsidiary of Public Company is a party or which are binding on any of them providing for the issuance, disposition or acquisition of any equity interests of any such Subsidiary other than to Public Company or another Subsidiary.
- (c) All of the outstanding capital stock of or other voting securities of or ownership interests in, each Subsidiary of Public Company, is owned by Public Company, directly or indirectly, free and clear of any Lien and free of any transfer restriction (other than transfer restrictions of general applicability as may be provided under the Securities Act or other applicable securities Laws), including any restriction on the right to vote, sell or otherwise dispose of such capital stock or other voting securities or ownership interests.
- (d) Except for its interests in each Subsidiary of Public Company, Public Company does not own, directly or indirectly, any capital stock, membership interest, partnership interest, joint venture interest or other equity interest in any other Person. There are no outstanding contractual obligations of Public Company or any of its Subsidiaries in any Person.

4.4 Authority; No Conflict; Required Filings and Consents.

(a) Each of Public Company and Merger Subsidiary has all requisite power and authority to execute and deliver this Agreement and the other agreements contemplated hereby and to perform its respective obligations hereunder and thereunder. The execution and delivery by Public Company and Merger Subsidiary of this Agreement and the other agreements contemplated hereby and, subject to obtaining the Public Company Stockholder Approval, which is the only approval required from the shareholders of Public Company and its Subsidiaries, the performance by Public Company and Merger Subsidiary of this Agreement and the consummation by Public Company and Merger Subsidiary of the Transactions have been duly and validly authorized by all necessary corporate action on the part of Public Company and Merger Subsidiary, respectively. This Agreement and all other agreements contemplated hereby have been duly and validly executed and delivered by Public Company and Merger Subsidiary and, assuming the due execution and delivery hereof and thereof by the other parties hereto and thereto, constitutes or will constitute a valid and binding obligation of each of Public Company and Merger Subsidiary, as applicable, enforceable against it in accordance with its terms, except as such enforceability may be limited by bankruptcy, insolvency, moratorium, and other similar Laws affecting creditors' rights generally and by general principles of equity.

- (b) The Public Company Board (at a meeting or meetings duly called and held and at which all members were present) has unanimously: (i) determined that the Transactions are fair to, advisable and in the best interests of Public Company and its stockholders; (ii) authorized, approved and declared advisable this Agreement and the Transactions, including the issuance of shares of Public Company Common Stock to the stockholders of Blockers and Merger Partner pursuant to the terms of this Agreement; and (iii) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the stockholders of Public Company vote to approve the Public Company Stockholder Matters and the Other Public Company Stockholder Matters. The board of directors of Merger Subsidiary (by unanimous written consent) has: (A) determined that the Transactions are fair to, advisable, and in the best interests of Merger Subsidiary and its sole stockholder; (B) authorized, approved and declared advisable this Agreement and the Transactions, including the Merger Partner Merger; and (C) determined to recommend, upon the terms and subject to the conditions set forth in this Agreement, that the sole stockholder of Merger Subsidiary vote to adopt this Agreement and thereby approve the Transactions.
- (c) Subject to the filing of the each Blocker Certificate of Merger and the Merger Partner Certificate of Merger as required by Delaware Law, neither the execution and delivery by Public Company or Merger Subsidiary of this Agreement or any other agreement contemplated hereby, nor the performance by Public Company or Merger Subsidiary of their respective obligations hereunder or thereunder, nor the consummation by Public Company or Merger Subsidiary of the Transactions, will (i) conflict with or violate any provision of the Organizational Documents of Public Company or Merger Subsidiary, each as amended or restated to date, (ii) require on the part of Public Company or Merger Subsidiary or any of their respective Subsidiaries any notice to or filing with, or any permit, authorization, consent or approval of, any Governmental Entity, (iii) conflict with, result in a breach of, constitute (with or without due notice or lapse of time or both) a default under, result in the acceleration of obligations under, create in any party the right to accelerate, terminate, modify or cancel, or require any notice, consent or waiver under, any contract, lease, sublease, license, sublicense, franchise, permit, indenture, agreement or mortgage for borrowed money, instrument of Indebtedness, Lien (other than Permitted Liens) or other arrangement to which Public Company or any of its Subsidiaries is a party or by which any of them are bound or to which any of their assets are subject, (iv) result in the imposition of any Lien other than Permitted Liens upon any assets of Public Company or any of its Subsidiaries or (v) violate any order, writ, injunction, decree, statute, rule or regulation applicable to Public Company or any of its Subsidiaries or any of their respective properties or assets of any of the foregoing, except in the case of clauses (ii) through (v) as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect.
- 4.5 <u>Operations of Merger Subsidiary</u>. Merger Subsidiary a wholly owned subsidiary of Public Company and has not engaged in any business activities or conducted any operations of any kind, entered into any agreement or arrangement with any Person, or incurred, directly or indirectly, any liabilities, in each case except in connection with its incorporation and the negotiation of this Agreement.

4.6 SEC Filings; Financial Statements.

- (a) Public Company has timely filed or furnished (as applicable) all forms, reports, certifications and other documents required to be filed by Public Company with the SEC since January 1, 2020. All such registration statements, forms, reports and other documents filed by Public Company with the SEC prior to the date hereof are referred to herein as the "Public Company SEC Reports." None of the Subsidiaries of Public Company are required to make any filings with the SEC. All of Public Company SEC Reports (a) were filed on a timely basis, and (b) at the time filed, complied as to form in all material respects with the requirements of the Securities Act and the Exchange Act applicable to such Public Company SEC Reports and (c) do not, as of the date hereof, contain a material misstatement, untrue fact or omission except as set forth in disclosures of the Public Company SEC Reports. The certifications and statements required by (i) Rule 13a-14 or Rule 15d-14(a) under the Exchange Act and (ii) 18 U.S.C. §1350 (Section 906 of the Sarbanes-Oxley Act) relating to the Public Company SEC Reports (collectively, the "Certifications") are accurate and complete and comply as to form and content with all applicable Laws. The Public Company SEC Reports at the time filed (a) complied as to form in all material respects with applicable accounting requirements and the published rules and regulations of the SEC with respect thereto. (b) were prepared in accordance with GAAP applied on a consistent basis throughout the periods involved and at the dates involved (except as may be indicated in the notes to such financial statements or, in the case of unaudited interim financial statements, as permitted by the SEC on Form 10-O under the Exchange Act) and (c) fairly presented in accordance with GAAP the consolidated financial position of Public Company and its Subsidiaries as of the dates indicated and the consolidated results of its operations, cash flows and stockholders' equity for the periods indicated, except that the unaudited interim financial statements were or are subject to normal and recurring year-end adjustments. Other than as expressly disclosed in the Public Company SEC Reports filed prior to the date hereof or have been permitted or required by any Governmental Authority, there has been no material change in Public Company's accounting methods or principles that would be required to be disclosed in Public Company's financial statements in accordance with GAAP. The interactive data in eXtensible Business Reporting Language (XBRL) included or incorporated by reference in the Public Company SEC Reports fairly presents the information called for in all material respects and has been prepared in accordance with the SEC's rules and guidelines applicable thereto.
- (b) As of the date of this Agreement, there are no outstanding or unresolved comments received from the SEC with respect to any Public Company SEC Report and, to the Knowledge of Public Company, none of the Public Company SEC Reports is the subject of ongoing SEC review. As of the date of this Agreement, there has been no material correspondence between the SEC and Public Company since January 1, 2020 that is not set forth in the Public Company SEC Reports or that has not otherwise been disclosed to Merger Partner prior to the date of this Agreement.
- (c) Public Company maintains accurate books and records reflecting its assets and liabilities and maintains proper and adequate system of internal control over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) that is sufficient to provide assurance that (i) transactions are executed with management's authorization, (ii) transactions are recorded as necessary to permit preparation of the financial statements of Public Company and to maintain accountability for Public Company's assets, (iii) access to assets of

Public Company is permitted only in accordance with management's authorization, (iv) the reporting of assets of Public Company is compared with existing assets at regular intervals, and (v) accounts, notes and other receivables and inventory were recorded accurately, and proper and adequate procedures are implemented to effect the collection thereof on a current and timely basis. Public Company has disclosed to Public Company's auditors and audit committee (and made available to Merger Partner a summary of the significant aspects of such disclosure) (A) all significant deficiencies and material weaknesses, if any, in the design or operation of internal control over financial reporting that are reasonably likely to adversely affect Public Company's ability to record, process, summarize and report financial information and (B) any Fraud, whether or not material, that involves management or other employees who have a significant role in Public Company's or its Subsidiaries' internal control over financial reporting. Public Company has not identified any material weaknesses in the design or operation of Public Company's internal control over financial reporting.

- (d) Neither Public Company nor any of its Subsidiaries has any Liability, except for: (i) Liabilities disclosed, reflected or reserved against in Public Company's most recent interim consolidated unaudited balance sheet as of September 30, 2022; (ii) Liabilities that have been incurred by Public Company or its Subsidiaries since September 30, 2022 in the Ordinary Course of Business; (iii) Liabilities for performance of obligations of Public Company or any of its Subsidiaries under Public Company Contracts (other than those arising as a result of a breach or default thereunder or as a result of failure to comply with applicable Law); (iv) Liabilities for payment of fees and expenses incurred in connection with the Transactions; (v) Liabilities which would not, individually or in the aggregate, reasonably be expected to be material to Public Company and its Subsidiaries, taken as a whole; and (vi) Liabilities described in Section 4.6(d) of the Public Company Disclosure Schedule.
- (e) The information in the Registration Statement to be supplied by or on behalf of Public Company for inclusion or incorporation by reference in the Registration Statement or supplied by or on behalf of Public Company for inclusion in any Regulation M-A Filing, shall not at the time the Registration Statement or any such Regulation M-A Filing is filed with the SEC, at any time it is amended or supplemented or at the time the Registration Statement is declared effective by the SEC, as applicable, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein not misleading, in light of the circumstances under which they were made. The information to be supplied by or on behalf of Public Company for inclusion or incorporation by reference in the Proxy Statement/Prospectus to be sent to the stockholders of Public Company in connection with the Public Company Stockholder Meeting, which information shall be deemed to include all information about or relating to Public Company, the Public Company Stockholder Matters, the Other Public Company Stockholder Matters (as applicable) or the Public Company Stockholder Meeting, shall not, on the date the Proxy Statement/Prospectus is first mailed to stockholders of Public Company, or at the time of the Public Company Stockholder Meeting or as of the Effective Time, contain any statement that, at such time and in light of the circumstances under which it shall be made, is false or misleading with respect to any material fact, or omit to state any material fact necessary to correct any statement in any earlier communication with respect to the solicitation of proxies for the Public Company Meeting that has become, in light of the circumstances in which they were made, false or misleading.

- (f) Public Company's independent registered accounting firm has at all times since its first date of service to Public Company has been: (i) a registered public accounting firm (as defined in Section 2(a)(12) of the Sarbanes-Oxley Act); (ii) to the Knowledge of Public Company, "independent" with respect to Public Company within the meaning of Regulation S-X under the Exchange Act; and (iii) to the Knowledge of Public Company, in compliance with subsections (g) through (l) of Section 10A of the Exchange Act and the rules and regulations promulgated by the SEC and Public Company Board.
- (g) Between January 1, 2020 and the date of this Agreement, there have been no formal investigations regarding financial reporting or accounting policies and practices discussed with, reviewed by or initiated at the direction of the chief executive officer, chief financial officer, principal accounting officer or general counsel of Public Company, the Public Company Board or any committee thereof, other than ordinary course audits or reviews of accounting policies and practices or internal controls required by the Sarbanes-Oxley Act.
- 4.7 <u>Absence of Certain Changes</u>. Since December 31, 2021, except for the execution and performance of this Agreement and the discussions, negotiations and transactions related thereto, (a) there has occurred no event or development which, individually or in the aggregate, has had, or could reasonably be expected to have in the future, a Public Company Material Adverse Effect, (b) Public Company and the Subsidiaries have conducted their businesses in the Ordinary Course of Business and (c) neither Public Company nor any Subsidiary has taken any of the actions set forth in clauses (a) through (s) of Section 5.2.

4.8 Tax Matters.

- (a) Each of Public Company and its Subsidiaries has properly filed on a timely basis all income and other material Tax Returns that it was required to file, and all such Tax Returns were true, correct and complete in all material respects. Each of Public Company and its Subsidiaries has paid on a timely basis all income or other material Taxes, whether or not shown on any Tax Return, that were due and payable.
- (b) Neither Public Company nor any of its Subsidiaries is or has ever been a member of an affiliated group with which it has filed (or been required to file) consolidated, combined, unitary or similar U.S. federal Tax Returns, other than a group of which the common parent is Public Company. With the exception of customary commercial leases or contracts that are not primarily related to Taxes entered into in the Ordinary Course of Business and liabilities thereunder, neither Public Company nor any of its Subsidiaries (i) has any liability under Treasury Regulations Section 1.1502-6 (or any comparable or similar provision of state, local or non-U.S. law), as a transferee or successor, or pursuant to any contractual obligation for any Taxes of any Person other than Public Company or any of its Subsidiaries, or (ii) is a party to or bound by any Tax indemnity, Tax sharing, Tax allocation or similar agreement.

- (c) All material Taxes that Public Company or any of its Subsidiaries was required by Law to withhold or collect have been duly withheld or collected and, to the extent required, have been properly paid to the appropriate Governmental Entity, in each case in compliance in all material respects with applicable law.
- (d) Public Company has delivered or made available to Merger Partner (i) complete and correct copies of all income and other material Tax Returns of Public Company and any of its Subsidiaries relating to Taxes for all taxable periods for which the applicable statute of limitations has not yet expired, (ii) complete and correct copies of all private letter rulings, revenue agent reports, information document requests, notices of proposed deficiencies, deficiency notices, protests, petitions, closing agreements, settlement agreements, pending ruling requests and any similar documents submitted by, received by, or agreed to by or on behalf of Public Company or any of its Subsidiaries relating to Taxes for all taxable periods for which the statute of limitations has not yet expired, and (iii) complete and correct copies of all material agreements, rulings, settlements or other Tax documents with or from any Governmental Entity relating to Tax incentives of Public Company or any of its Subsidiaries.
- (e) No examination or audit of any Tax Return of Public Company or any of its Subsidiaries by any Governmental Entity is currently in progress or, to the Knowledge of Public Company, has been threatened by any Governmental Entity. No deficiencies for Taxes of Public Company or any of its Subsidiaries have been claimed, proposed or assessed by any Governmental Entity in writing. Neither Public Company nor any of its Subsidiaries has been informed in writing by any jurisdiction in which Public Company or any of its Subsidiaries does not file a Tax Return that the jurisdiction believes that Public Company or any of its Subsidiaries was required to file any Tax Return that was not filed or is subject to Tax in such jurisdiction. Neither Public Company nor any of its Subsidiaries has (i) waived any statute of limitations with respect to Taxes or agreed to extend the period for assessment or collection of any Taxes, which waiver or extension is still in effect, (ii) requested any extension of time within which to file any Tax Return (other than any automatic extension granted in the ordinary course of business and consistent with past custom and practice of Public Company), or (iii) executed or filed any power of attorney with any taxing authority, which is still in effect.
- (f) Neither Public Company nor any of its Subsidiaries has made any payment or provided any benefit, is obligated to make any payment or provide any benefit, or is a party to any plan, program, policy, agreement or arrangement that could obligate it to make any payment or provide any benefit that may be treated as an "excess parachute payment" under Section 280G of the Code (without regard to Sections 280G(b)(4) and 280G(b)(5) of the Code).
- (g) Neither Public Company nor any of its Subsidiaries has been a United States real property holding corporation within the meaning of Section 897(c)(2) of the Code during the applicable period specified in Section 897(c)(1)(A)(ii) of the Code.
- (h) There are no Liens with respect to Taxes upon any of the assets or properties of Public Company or any of its Subsidiaries, other than Permitted Liens.
- (i) Neither Public Company nor any of its Subsidiaries has participated in any "reportable transaction" as defined in Treasury Regulations Section 1.6011-4(b) or a "listed transaction" as set forth in Treasury Regulations Section 301.6111-2(b)(2) or any analogous provision of state or local law.

(j) During the two-year period ending on the date hereof, Public Company nor any predecessor thereof has constituted either a "distributing corporation" or a "controlled corporation" in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code (or so much of Section 356 of the Code as relates to Section 355 of the Code).

Other than Section 4.16, this Section 4.8 provides the sole and exclusive representations and warranties of Public Company and its Subsidiaries in respect of Tax matters

4.9 <u>Assets</u>. Except as would not be, individually or in the aggregate, material to Public Company and its Subsidiaries, taken as a whole, Public Company or its applicable Subsidiary is the true and lawful owner of, and has good title to, all of the assets (tangible or intangible) purported to be owned by Public Company or such Subsidiary, free and clear of all Liens other than Permitted Liens. Public Company and each Subsidiary owns or leases all material tangible assets sufficient for the conduct of its businesses as presently conducted, which tangible assets are reflected in the Public Company Financial Statements (other than to the extent disposed of in the Ordinary Course of Business). Except as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect, each such tangible asset is free from defects, has been maintained in accordance with normal industry practice, is in good operating condition and repair (subject to normal wear and tear) and is suitable for the purposes for which it presently is used.

4.10 Owned and Leased Real Property.

- (a) Neither Public Company nor any of its Subsidiaries owns any real property.
- (b) Public Company has delivered to Merger Partner complete and accurate copies of all material Leases of Public Company and its Subsidiaries. Neither Public Company nor any Subsidiary occupies any space other than pursuant to a Lease. With respect to each such Lease, except as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect:
- (i) such Lease is legal, valid, binding, enforceable and in full force and effect against Public Company or its Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Public Company, against each other party thereto;
- (ii) such Lease will continue to be legal, valid, binding, enforceable and in full force and effect against Public Company or the Subsidiary that is the party thereto, as applicable, and, to the Knowledge of Public Company, against each other party thereto immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing;
- (iii) none of Public Company, any Subsidiary of Public Company or, to the Knowledge of Public Company, any other party, is in breach or violation of, or default under, any such Lease, and no event has occurred, is pending or, to the Knowledge of Public Company, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute any such breach or default by Public Company or any Subsidiary of Public Company or, to the Knowledge of Public Company, any other party under such Lease; and no event has occurred that would give rise to a termination right under such Lease;

- (iv) neither Public Company nor any Subsidiary of Public Company has assigned, transferred, conveyed, mortgaged, subleased, licensed, deeded in trust or encumbered any interest in the leasehold or subleasehold; and
- (v) to the Knowledge of Public Company, there are no Liens (other than Permitted Liens) applicable to the real property subject to such Lease which would reasonably be expected to impair the current uses or the occupancy by Public Company or any Subsidiary of Public Company of the property subject thereto.
- (c) Each of Public Company and its Subsidiaries owns, and has good and valid title to, or, in the case of leased properties and assets, valid leasehold interests in, all tangible properties or tangible assets and equipment used or held for use in its business or operations or purported to be owned by it, in each case, that are material to Public Company and its Subsidiaries, taken as a whole, including: (A) all such tangible assets reflected on the Public Company Financial Statements and (B) all other such tangible assets reflected in the books and records of Public Company or any of its Subsidiaries as being owned by Public Company or such Subsidiary. All of such assets are owned or, in the case of leased assets, leased by Public Company or any of its Subsidiaries free of any Liens other than Permitted Liens.

4.11 Intellectual Property.

- (a) Section 4.11(a) of the Public Company Disclosure Schedule sets forth a true, correct and complete list of all Patents, Copyright registrations, Trademark registrations, applications for Trademark registration, material unregistered Trademarks, Design registrations and applications for Design registration, and Internet domain names owned (in whole or in part) by or exclusively licensed to Public Company or any of its Subsidiaries.
- (b) (i) Public Company and its Subsidiaries (A) solely own all Intellectual Property owned or purported to be owned by Public Company or any of its Subsidiaries and (B) possess, hold or have obtained valid and enforceable licenses or other rights under all other Intellectual Property necessary to operate the business of Public Company in the manner now operated and as currently proposed to be operated by it as disclosed in the Registration Statement and Prospectus, except, in each case of (A) and (B), as such failure to own or obtain such licenses would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect; (ii) none of the Intellectual Property owned by, exclusively licensed to, or to the Knowledge of Public Company, otherwise licensed to or used by, Public Company and its Subsidiaries (collectively, the "Public Company Intellectual Property") has been adjudged invalid or unenforceable, in whole or in part; (iii) to the Knowledge of Public Company, no third party is infringing, misappropriating, or otherwise violating any Public Company Intellectual Property, except, where such infringement, misappropriation or other violation would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect; (iv) since January 1, 2020 to the date of this Agreement, to the Knowledge of Public Company, there has been no actual or threatened action, suit, proceeding or claim by others challenging the validity, enforceability or scope of, or challenging

Public Company's ownership of or rights in or to, any Public Company Intellectual Property, and Public Company is not aware of any facts or circumstances that would render any Public Company Intellectual Property invalid or unenforceable by Public Company or one of its Subsidiaries, or of inadequate scope to protect the interests of Public Company in conducting its business, except in each case, as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect; (v) since January 1, 2020 to the date of this Agreement, to the Knowledge of Public Company, there has been no actual, or threatened action, suit, proceeding or claim by a third party alleging that Public Company or any of its Subsidiaries infringes, misappropriates, or otherwise violates, any Intellectual Property of third parties, and Public Company has not received any notice alleging, or is otherwise aware of, any facts or circumstances that would give rise to such an action, proceeding or claim, except, in each case, where such infringement, misappropriation or other violation would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect; (vi) Public Company has taken all reasonable steps to protect, maintain and safeguard the Public Company Intellectual Property, including the execution of appropriate nondisclosure, confidentiality agreements and invention assignment agreements and invention assignments with their directors, officers, employees, contractors and other agents and since January 1, 2020 to the date of this Agreement, to the Knowledge of Public Company, no such individual is in or has been in violation of any term of such nondisclosure, confidentiality agreements and invention assignment agreements; and (vii) since January 1, 2020 to the date of this Agreement, to the Knowledge of Public Company, no material technology employed by Public Company or its Subsidiaries has been obtained or is being used by Public Company or its Subsidiaries in violation of any contractual or legal obligation binding on Public Company or its Subsidiaries or any of its officers, directors or employees, which violation relates to the breach of a confidentiality obligation, obligation to assign Intellectual Property to a previous employer or obligation otherwise not to use the Intellectual Property of a third party.

(c) To the Knowledge of Public Company, the material information technology systems of Public Company and its Subsidiaries are adequate for their respective businesses as currently conducted in all material respects.

4.12 Contracts.

(a) Section 4.12(a) of Public Company Disclosure Schedule lists the following agreements in effect as of the date of this Agreement (each a "<u>Public Company Contract</u>") to which Public Company or any of its Subsidiaries is a party as of the date of this Agreement and which contains any remaining material rights or obligations:

(i) any agreement (or group of related agreements) (A) which calls for performance over a period of more than one year, (B) which involves more than the sum of \$500,000, (C) in which Public Company or any of its Subsidiaries has granted "most favored nation" pricing provisions or exclusive marketing or distribution rights or has agreed to purchase goods or services exclusively from a certain party, (D) which limits the freedom of Public Company or its Subsidiaries to engage in any line of business or compete with any Person or (E) which contains any non-solicitation provision, in each case, except for restrictions that would not materially affect the ability of Public Company and its Subsidiaries to conduct its business, in each case, that is not terminable by Public Company or its Subsidiaries, as applicable, on 90 days' notice or less;

- (ii) any agreement relating to the disposition or acquisition of material assets or any ownership interests in any Person, in each case, involving payments in excess of \$500,000, other than agreements in which the applicable acquisition or disposition has been consummated and there are no material ongoing liabilities or obligations;
 - (iii) any agreement providing for any material royalty, milestone or similar payments by Public Company or its Subsidiaries;
 - (iv) any agreement concerning the establishment or operation of a partnership, joint venture or limited liability company;
- (v) any agreement (or group of related agreements) under which Public Company or any of its Subsidiaries has created, incurred, assumed or guaranteed (or may create, incur, assume or guarantee) material Indebtedness (including capitalized lease obligations) or under which it has imposed (or may impose) a Lien (other than Permitted Liens) on any of its assets, tangible or intangible;
- (vi) any agreement for the disposition of any material assets or business of Public Company or any of its Subsidiaries or any agreement for the acquisition of material assets or business of any other Person (other than purchases of inventory or components in the Ordinary Course of Business);
- (vii) any noncompetition agreement restricting Public Company or any of its Subsidiaries from competing in any line of business or jurisdiction;
- (viii) any written employment agreement or consulting agreement that is not on Public Company's standard form of employment agreement or consulting agreement, each of which has been previously made available to Merger Partner (excluding employment agreements that are terminable "at will" without the payment of severance or other amounts upon termination, and consulting agreements which are terminable on 30 days or less notice without the payment of additional consideration);
 - (ix) any agreement providing for severance, retention, change in control payments, or transaction-based bonuses or incentives;
- (x) any material settlement agreement or agreement that includes the admission of wrongdoing by Public Company or any of its Subsidiaries or any of their respective officers or directors;
- (xi) any agreement with respect to an interest rate, currency or other swap or derivative transaction (other than those solely between Public Company or any of its Subsidiaries);

(xii) any agreement between Public Company or any of its Subsidiaries, on the one hand, and any director, officer or employee of Public Company or any of its Subsidiaries or any Person beneficially owning five percent or more of the outstanding Public Company Common Stock or any of their respective Affiliates, on the other hand;

(xiii) except as set forth in the Organizational Documents of Public Company or in Public Company's standard form of Public Company RSU grant agreement, any agreement that contains a put, call or similar right pursuant to which Public Company or any of its Subsidiaries or after the Merger Partner Merger Effective Time, Merger Partner or any of its Subsidiaries or Affiliates, could be required to purchase or sell, as applicable, in a single transaction or series of transactions, (A) any equity interests of any Person or (B) any asset or group of related assets that have a fair market value or purchase price of more than \$500,000 individually or in the aggregate;

(xiv) any agreement relating to royalty, dividend or similar arrangement to be paid, or received, by Public Company or any of its Subsidiaries that is based on the revenue or profits of Public Company or any of its Subsidiaries or any material agreement involving fixed price or fixed volume arrangements;

(xv) any agreement under which the consequences of a default or termination would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect;

(xvi) any agreement under which Public Company or any of its Subsidiaries has (A) granted a license, sublicense, covenant not to sue or assert, immunity from suit, right to coexist or other right under any Intellectual Property to any other Person; (B) received from any other Person a grant of a license, sublicense, covenant not to sue or assert, immunity from suit, right to coexist or other right under any Intellectual Property (other than licenses to any shrink wrap, click wrap or other software that is generally commercially available and not customized costing less than \$50,000 per year or as a one-time fee); (C) assigned any rights under any Intellectual Property to any other Person; or (D) received an assignment of any rights under Intellectual Property from any Person (other than assignments from employees on the form assignment agreement used by Public Company with such employees); and

(xvii) any other agreement (or group of related agreements) that is material to Public Company and its Subsidiaries, taken as a whole, including any material contacts (as defined in Item 601(b)(10) of Regulation S-K as promulgated under the Securities Act).

(b) Public Company has made available to Merger Partner a complete and accurate copy of each Public Company Contract (as amended to date) as of the date of this Agreement. With respect to each Public Company Contract: (i) Public Company Contract is legal, valid, binding and enforceable and in full force and effect against Public Company or the Subsidiary that is the party thereto, as applicable, and, to Public Company's Knowledge, against each other party thereto; (ii) Public Company Contract will continue to be legal, valid, binding and enforceable and in full force and effect against Public Company or the Subsidiary that is the party thereto, as applicable, and, to Public Company's Knowledge, against each other party thereto immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing; and (iii) neither Public Company or Subsidiary nor, to the

Knowledge of Public Company, any other party, is in breach or violation of, or default under, any such Public Company Contract, and no event has occurred, is pending or, to the Knowledge of Public Company, is threatened, which, after the giving of notice, with lapse of time, or otherwise, would constitute any such breach or default by Public Company, any Subsidiary or, to the Knowledge of Public Company, any other party under such Public Company Contract, in each case, where any such breach or violation would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect.

4.13 Litigation. As of the date hereof, there is no Legal Proceeding pending or, to the Knowledge of Public Company, threatened with respect to, (A) against or affecting (i) Public Company or any Subsidiary of Public Company, (ii) or any current or former officer, director, employee, consultant, agent or member of any of the foregoing in its, his or her capacity as such, or (iii) any of the material assets owned or used by Public Company and its Subsidiaries and (B) no notice of any Legal Proceeding involving or relating to Public Company or any of its Subsidiaries, whether pending or threatened, has been received by Public Company or any of its Subsidiaries, in each case as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect or that would have the effect of preventing, delaying beyond the Outside Date, or making illegal, the Transactions (including the Mergers). As of the date of this Agreement, there are no material judgments, orders, injunctions, decrees, stipulations or awards (whether rendered by a court, administrative agency or other Governmental Entity, by arbitration or otherwise) against or involving Public Company or any Subsidiary of Public Company. As of the date of this Agreement, there is no material Legal Proceeding by Public Company or any Subsidiary of Public Company, or which Public Company or any Subsidiary of Public Company has commenced preparations to initiate, against any other Person.

4.14 Environmental Matters.

- (a) Since January 1, 2020, Public Company and its Subsidiaries have complied with all applicable Environmental Laws except as would not, individually or in the aggregate, reasonably be expected to result in a Public Company Material Adverse Effect. There is no pending or, to the Knowledge of Public Company, threatened material Legal Proceeding relating to any Environmental Law involving Public Company or any Subsidiary of Public Company.
- (b) To the Knowledge of Public Company, neither Public Company nor any Subsidiary of Public Company has any material liabilities or material obligations arising from the release or threatened release of any Materials of Environmental Concern into the environment.
- (c) Neither Public Company nor any Subsidiary of Public Company is a party to or bound by any court order, administrative order, consent order or other agreement between Public Company or any Subsidiary of Public Company and any Governmental Entity entered into in connection with any material legal obligation or material liability arising under any Environmental Law.

(d) Public Company has no Knowledge of any material environmental liability relating to any solid or hazardous waste transporter or treatment, storage or disposal facility that has been used by Public Company or any Subsidiary of Public Company.

4.15 <u>Labor and Employment</u>.

- (a) Public Company has made available to Merger Partner a true and correct list of all current Public Company Employees as of the date of this Agreement. Each such current Public Company Employee is retained at-will and none of such current Public Company Employees is a party to a written employment agreement or contract with Public Company or any of its Subsidiaries contradicting their at-will employment. Each current Public Company Employee has entered into Public Company's or such Subsidiary's standard form of confidentiality, non-competition and assignment of inventions agreement, a copy of which has previously been delivered to Merger Partner. All of the agreements referenced in the preceding sentence will continue to be in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing.
- (b) Since January 1, 2020, neither Public Company nor any Subsidiary of Public Company has breached or violated any (i) applicable Law respecting employment and employment practices, terms and conditions of employment and wages and hours, including any such Law respecting employment discrimination, employee classification (for overtime purposes or as employee versus independent contractor), workers' compensation, family and medical leave, the Immigration Reform and Control Act and occupational safety and health requirements, or (ii) employment or other individual service provider agreement, in each case except as would not, individually or in the aggregate, reasonably be expected to be material to Public Company and its Subsidiaries, taken as a whole. As of the date hereof, no material claims, controversies, investigations, audits or other Legal Proceedings are pending or, to the Knowledge of Public Company, threatened, with respect to such Laws or agreements, either by private Persons or by Governmental Entities.
- (c) Neither Public Company nor any Subsidiary of Public Company is a party to or bound by any collective bargaining agreement, nor has either of them experienced any actual or, to the Knowledge of Public Company, threatened strikes, grievances, claims of unfair labor practices or other collective bargaining disputes. Public Company has no Knowledge of any material organizational effort made or threatened (including the filing of a petition for certification) either currently or within the past two (2) years, by or on behalf of any labor union with respect to Public Company Employees.
- (d) Public Company has made available to Merger Partner a true and correct list of all consultants and independent contractors as of the date of this Agreement engaged by or for the benefit of Public Company or its Subsidiaries. Each such consultant or independent contractor is or was a party to a written agreement or contract with Public Company or its Subsidiaries. Public Company and its Subsidiaries have not incurred, and, to the Knowledge of Public Company, no circumstances exist under which Public Company and its Subsidiaries could reasonably be expected to incur, any material liability arising from the misclassification of employees as consultants or independent contractors, or from the misclassification of consultants or independent contractors as employees. Each such consultant and independent contractor has entered into Public Company's standard form of confidentiality and assignment of inventions agreement with Public Company, a copy of which has previously been made available to Merger Partner.

- (e) As of the date hereof, no charges or complaints are open and pending against Public Company or any of its Subsidiaries with the Equal Employment Opportunity Commission, the OFCCP, or similar Governmental Entity or pursuant to internal complaint procedures, and, to the Knowledge of Public Company, as of the date hereof no current or former employee of Public Company or any of its Subsidiaries has made, during the last twelve (12) months, an oral or, during the last three years, a written complaint of discrimination, retaliation or other similar wrongdoing. True, correct and complete information regarding any closed charges or complaints filed since December 31, 2020 through the date of this Agreement with the Equal Employment Opportunity Commission, the OFCCP or similar Governmental Entity (or, with respect to discrimination, retaliation, or similar wrongdoing, pursuant to internal complaint procedures) has been made available to Merger Partner.
- (f) Since January 1, 2020, neither Public Company nor any of its Subsidiaries has caused (i) a plant closing as defined in the WARN Act affecting any site of employment or one or more operating units within any site of employment of Public Company or any of its Subsidiaries or (ii) a mass layoff as defined in the WARN Act, nor has Public Company or any of its Subsidiaries been affected by any transaction or engaged in layoffs or employment terminations sufficient in number to trigger application of any similar foreign, state or local Law.

4.16 Employee Benefit Plans.

- (a) Section 4.16(a) of Public Company Disclosure Schedule contains a complete and accurate list of all material Public Company Benefit Plans. Complete and accurate copies of (i) all Public Company Benefit Plans which have been reduced to writing, together with all amendments thereto, (ii) written summaries of all material unwritten Public Company Benefit Plans, (iii) all related trust agreements, insurance contracts, and most recent summary plan description together with the summary or summaries of material modifications thereto, (iv) the most recently filed annual report on IRS Form 5500 and all schedules thereto and (for all funded plans) all plan financial statements for the last plan year for each Public Company Benefit Plan, (v) all reports regarding the satisfaction of the nondiscrimination requirements of Sections 410(b), 401(k), and 401(m) of the Code for the past year, (vi) the most recent determination letter or opinion letter issued by the Internal Revenue Service with respect to any Public Company Benefit Plan intended to be qualified under Section 401(a) of the Code, and (vii) any material written or electronic communications from or to the Internal Revenue Service, the DOL or any other Governmental Entity received in the three (3) year period preceding the date of this Agreement with respect to a Public Company Benefit Plan (including any voluntary correction submissions), have been delivered or made available to Public Company. No Public Company Benefit Plan is or has been subject to non-U.S. Law.
- (b) Neither Public Company, any of its Subsidiaries, nor any of their respective ERISA Affiliates sponsors, maintains or contributes to or in the last six years has sponsored, maintained, contributed to or been required to contribute to, or has (or is reasonably expected to have) any material current or contingent liability with respect to any (i) "employee

pension benefit plan," as defined in Section 3(2) of ERISA, that is subject to Title IV or Section 302 of ERISA or Section 412 of the Code, (ii) a "multiemployer plan" within the meaning of Section 4001(a)(3) of ERISA, or (iii) a "multiple employer plan" subject to Sections 4063 or 4064 of ERISA or Section 413(c) of the Code. No liability under Section 302 or Title IV of ERISA or Section 412 of the Code has been incurred by Public Company, any of its Subsidiaries or any of their respective ERISA Affiliates that has not been satisfied in full.

- (c) Each Public Company Benefit Plan has been administered in all material respects in accordance with its terms and each of Public Company, its Subsidiaries and its ERISA Affiliates has met its obligations in all material respects with respect to each Public Company Benefit Plan and has timely made all required contributions, premiums and other payments thereto, in each case in all material respects. Public Company, its Subsidiaries, each of its ERISA Affiliates and each Public Company Benefit Plan are in compliance in all material respects with the currently applicable provisions of ERISA and the Code (including Section 409A of the Code) and the regulations thereunder. Each Public Company Benefit Plan that is intended to be qualified under Section 401(a) of the Code has been determined by the IRS to be qualified under Section 401(a) of the Code and, no fact or event has occurred that would adversely affect the qualification or tax exemption of any such Public Company Benefit Plan. There is no legally binding commitment to create any additional Public Company Benefit Plans or to modify any existing Public Company Benefit Plans.
- (d) There are no Legal Proceedings (except claims for benefits payable in the normal operation of Public Company Plans and proceedings with respect to qualified domestic relations orders) against or involving any Public Company Benefit Plan or asserting any rights or claims to benefits under any Public Company Benefit Plan that could give rise to any material liability. No Public Company Benefit Plan is or within the last three (3) calendar years has been the subject of, or has received or provided notice that it is the subject of, examination by a Governmental Entity or a participant in a government sponsored amnesty, voluntary compliance or similar program.
- (e) The execution and delivery of this Agreement and the consummation of the Transactions will not, either alone or in combination with another event, (A) entitle any current or former employee, director, officer or independent contractor of Public Company or any of its Subsidiaries to severance pay, retention bonuses, non-competition payments or any other material compensatory payment, (B) accelerate the time of payment or vesting, or increase the amount of compensation due to any such employee, director, officer or independent contractor, or (C) directly or indirectly require Public Company to transfer or set aside any assets to fund any material benefits under any Public Company Benefit Plan. No director, officer, employee or other service provider of Public Company or any of its Subsidiaries is entitled to a gross up, make whole or other similar payment as a result of the imposition of Taxes under Section 4999 or Section 409A of the Code pursuant to any agreement or arrangement with Public Company or any of its Subsidiaries.

- 4.17 <u>Compliance with Laws</u>. Each of Public Company and its Subsidiaries, to the Knowledge of Public Company, has since January 1, 2020 conducted, and is conducting, its business and operations in compliance in all material respects with all applicable Laws. Since January 1, 2020, neither Public Company nor any Subsidiary, to the Knowledge of Public Company, has received any notice or other communication from any Governmental Entity or other Person alleging any material noncompliance with any applicable Law. To the Knowledge of Public Company, neither Public Company nor any Subsidiary has any material liability for failure to comply with any Law and, to the Knowledge of Public Company, there is no act, omission, event or circumstance that would reasonably be expected to give rise to any such liability. Neither Public Company nor any Subsidiary has conducted any internal investigation with respect to any actual, potential or alleged violation of any Law by any manager, member or other equity holder, officer or Public Company Employee or concerning any actual or alleged Fraud.
- 4.18 <u>Unlawful Payments</u>. Since January 1, 2020, Public Company and its Subsidiaries are and have been in compliance in all material respects with all Anti-Bribery Laws. Since January 1, 2020, neither Public Company nor any Subsidiary has received any written communication from any Governmental Entity that alleges that Public Company or any of its Subsidiaries, or any current or former Representatives thereof, is or may be in violation of, or has, or may have, any liability under, any Anti-Bribery Laws, and no such actual or potential violation of Anti-Bribery Laws has been discovered by or brought to the attention of Public Company or any of its Subsidiaries since January 1, 2020. Since January 1, 2020, neither Public Company nor any of its Subsidiaries has made or anticipates making any disclosures to any Governmental Entity for potential violations of Anti-Bribery Laws. To Public Company's Knowledge, none of Public Company and its Subsidiaries' current or former Representatives is currently an officer, agent or employee of a Governmental Entity. Neither Public Company nor any Subsidiary nor any of their respective current or former Representatives has directly or indirectly offered, given, reimbursed, paid or promised to pay, or authorized the payment of, any money or other thing of value (including any fee, gift, sample, travel expense or entertainment) or any commission payment payable to (a) any Person who is an official, officer, agent, employee or representative of any Governmental Entity or of any existing or prospective customer (whether or not owned by a Governmental Entity), (b) any political party or official thereof or (c) any candidate for political or political party office, in each case while knowing or having reason to believe that all or any portion of such money or thing of value would be offered, given, reimbursed, paid or promised, directly or indirectly, for purposes not allowable under the Anti-Bribery Laws, to any such official, officer, agent, employee, representative, political party,
- 4.19 Permits. Each of Public Company and its Subsidiaries owns or holds all material Permits that are required for Public Company and its Subsidiaries, respectively, to conduct their business as presently conducted or as proposed to be conducted. Each such Permit is in full force and effect. Public Company or the applicable Subsidiary, as the case may be, is in compliance in all material respects with the terms of each such Permit; and, as of the date of this Agreement, to the Knowledge of Public Company, no suspension or cancellation of such Permit is threatened and, to the Knowledge of Public Company, there is no basis for believing that such Permit will not be renewable upon expiration, other than in the Ordinary Course of Business. Each such Permit will continue in full force and effect immediately following the Closing. Public Company has made available to Merger Partner all such Permits as of the date of this Agreement.

4.20 Insurance.

- (a) Public Company has made available to Merger Partner a true and complete list, as of the date of this Agreement, all material insurance policies (including fire, theft, casualty, comprehensive general liability, workers compensation, business interruption, environmental, director and officer liability, product liability and automobile insurance policies and bond and surety arrangements) to which Public Company or any of its Subsidiaries is a party, a named insured or otherwise the beneficiary of coverage (the "Public Company Insurance Policies"), all of which are in full force and effect. The Public Company Insurance Policies are of the type and in the amounts customarily carried by organizations conducting businesses or owning assets similar to those of Public Company. There is no material claim pending under the Public Company Insurance Policies as to which coverage has been questioned, denied or disputed by the underwriter of such policy. All premiums due and payable under all of the Public Company Insurance Policies have been paid, and, to the Knowledge of Public Company, neither Public Company nor any Subsidiary may be liable for retroactive premiums or similar payments, and Public Company and its Subsidiaries are otherwise in compliance with the terms of the Public Company Insurance Policies in all material respects. Public Company has no Knowledge of any threatened termination of, or premium increase with respect to, any of the Public Company Insurances Policies. Each Public Company Insurance Policy will continue to be enforceable and in full force and effect immediately following the Closing in accordance with the terms thereof as in effect immediately prior to the Closing. There is no material claim by Public Company or any of its Subsidiaries pending under any Public Company Insurance Policy in favor of Public Company and its Subsidiaries that has been denied or disputed by the insurer other than denials and disputes in the Ordinary Course of Business.
- (b) Section 4.20(b) of Public Company Disclosure Schedule identifies all claims asserted by Public Company pursuant to any Public Company Insurance Policy since December 31, 2019 and describes the nature and status of each such claim.

4.21 Cybersecurity and Privacy.

- (a) Each of Public Company and its Subsidiaries comply, and have complied, in all material respects with all: (i) applicable Privacy Laws; (ii) policies, statements, and contractual obligations relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security, disposal, destruction, disclosure, or transfer of Personal Information of Public Company and its Subsidiaries; and (iii) all applicable industry standards including, without limitation, PCI DSS, and all other applicable requirements of the payment card brands (clauses (i)-(iii) of this Section 4.21(a)) collectively, the "Public Company Privacy Requirements"). The consummation of any of the Transactions will not violate any applicable Public Company Privacy Requirements.
- (b) Public Company and its Subsidiaries have taken all organizational, physical, administrative, and technical measures required by the Public Company Privacy Requirements and consistent with standards prudent in the industry in which Public Company and its Subsidiaries operate to protect: (i) the integrity, security, and operations of all Public Company IT Systems; and (ii) all Personal Information and all other data owned, controlled, or stored by Public Company and its Subsidiaries from and against data security incidents or other

misuse. In connection with each third party servicing, outsourcing, hosting or otherwise processing Personal Information collected, held, or processed by or on behalf of Public Company and its Subsidiaries, Public Company and its Subsidiaries has in accordance with Public Company Privacy Requirements entered into valid, binding and enforceable written data processing agreements with any such third party that ensures appropriate protection for Personal Information. None of Public Company and its Subsidiaries sells, rents or otherwise makes available, or permits any third party to, sell, rent or otherwise make available, to any Persona any Personal Information, except as stated in the applicable privacy policies and in compliance with Privacy Laws.

- (c) There have been no actual or suspected data security incidents, data breaches, ransomware incidents, or other adverse events or incidents related to any Public Company IT Systems, Personal Information, or data in the custody or control of Public Company or its Subsidiaries or any third party acting on behalf of Public Company or its Subsidiaries. Public Company and its Subsidiaries have not been subject to any claims or proceedings related to any data security incidents, ransomware incidents, or any violations of any Public Company Privacy Requirements, and there are no facts or circumstances which could reasonably serve as the basis for any such allegations or claims. Public Company and its Subsidiaries have taken commercially reasonable steps to provide for archival, back-up, recovery and restoration of its material business data.
- (d) Public Company and its Subsidiaries own or have a valid right to access and use all Public Company IT Systems. To Public Company's Knowledge, the Public Company IT Systems do not contain any viruses, worms, Trojan horses, bugs, faults or other devices, errors, contaminants or effects that are designed to (A) disrupt or adversely affect the functionality of any Public Company IT Systems or (B) enable or assist any Person to access without authorization any Public Company IT Systems. The Public Company IT Systems have not materially malfunctioned or failed within the past two years. Public Company and its Subsidiaries have taken reasonable, and to ensure the continued, uninterrupted, and error-free operation of the Public Company IT Systems. No Person has gained unauthorized access to any Public Company IT Systems.
- 4.22 <u>Brokers</u>. No broker, finder or investment banker is entitled to any brokerage, finder's or other fee or commission based upon arrangements made by or on behalf of Public Company or Merger Subsidiary respect of any of the Transactions.
- 4.23 <u>Transactions with Affiliates</u>. Except as set forth in the Public Company SEC Reports filed prior to the date of this Agreement, since December 31, 2021, no event has occurred that would be required to be reported by Public Company pursuant to Item 404 of Regulation S-K.
 - 4.24 [Reserved].
- 4.25 <u>Nasdaq Market Quotation</u>. As of the date of this Agreement, the issued and outstanding shares of Public Company Common Stock are registered pursuant to Section 12(b) of the Exchange Act and are listed for trading on the Nasdaq under the symbol "CLXT." As of the date of this Agreement, Public Company is in compliance in all material respects with the rules of the Nasdaq and there is no action or proceeding pending or, to the Knowledge of Public Company threatened, against Public Company by the Nasdaq, the Financial Industry Regulatory Authority or the SEC with respect to any intention by such entity to deregister the Public Company Common Stock or terminate the listing of Public Company Common Stock on the Nasdaq.

- 4.26 No Shareholder Rights Plan; No Antitakeover Law. As of the date of this Agreement, no "fair price," "moratorium," "control share acquisition" or other anti-takeover statute or similar domestic or foreign Law applies with respect to Public Company or any of its Subsidiaries in connection with this Agreement, the Mergers, the issuance of the Merger Consideration or any of the other Transactions. There is no stockholder rights plan, "poison pill" or similar anti-takeover agreement or plan in effect to which Public Company or any of its Subsidiaries is subject, party or otherwise bound which will be triggered by the execution and delivery of this Agreement or the Transactions.
- 4.27 <u>Valid Issuance</u>. The Public Company Class A Common Stock and Public Company Class B Common Stock to be issued in the Mergers will, when issued in accordance with the provisions of this Agreement, be validly issued, fully paid and nonassessable.
- 4.28 No Other Representations and Warranties. Except as previously set forth in this Article IV or in any certificate delivered by Public Company or Merger Subsidiary to Merger Partner pursuant to this Agreement or otherwise in the case of Fraud, neither Public Company nor Merger Subsidiary makes any representation or warranty, express or implied, at law or in equity, with respect to it or any of its assets, liabilities or operations, and any such other representations or warranties are hereby expressly disclaimed.
- 4.29 <u>Reliance</u>. Public Company acknowledges that, except for the representations and warranties contained in Article III, none of Merger Partner, any Blocker or any other Person has made, and Public Company has not relied on, any other express or implied representation or warranty by or on behalf of Merger Partner, any Blocker or any other Person.

ARTICLE V PRE-CLOSING COVENANTS

5.1 Operation of Merger Partner's Business. Except as contemplated by this Agreement, in connection with implementing the Transactions in a manner consistent with this Agreement, as set forth on Section 5.1 of Merger Partner Disclosure Schedule or as required by applicable Law (including any COVID-19 Measures), during the Pre-Closing Period, without the written consent of Public Company (which consent shall not be unreasonably withheld, conditioned or delayed), Merger Partner shall use commercially reasonable efforts to, and shall cause each Subsidiary to use commercially reasonable efforts to, conduct its operations only in the Ordinary Course of Business and in compliance with all applicable Laws in all material respects. Without limiting the generality of the foregoing, during the Pre-Closing Period and except as set forth on Section 5.1 of Merger Partner Disclosure Schedule or in connection with implementing the Transactions in a manner consistent with this Agreement, Merger Partner shall not, and shall cause each of its Subsidiaries not to, without the written consent of Public Company (such consent shall not be unreasonably withheld, conditioned or delayed):

- (a) issue or sell any equity interests in Merger Partner or any options, warrants or rights to acquire any such equity interests, or declare, accrue, set aside or pay any dividend or make any other distribution in respect of its equity interests, or repurchase or redeem any equity interests in Merger Partner or any Subsidiary; provided that Merger Partner (i) may issue additional equity interests to Merger Partner Employees, to the extent contemplated in the Allocation Schedule, and (ii) may issue additional Series F Preferred Units in Merger Partner to other Persons for a per unit cash purchase price of \$2.00 and otherwise on terms consistent with the terms and conditions for issuances of Series F Preferred Units in Merger Partner prior to the date of this Agreement (which threshold and minimum subscription price is subject to adjustment adjusted, as applicable and appropriate, to reflect fully the effect of any reclassification, split, reverse split, distribution (including any distribution of securities convertible into membership units of Merger Partner), reorganization, recapitalization or other like change with respect to the equity securities of Merger Partner or any securities convertible or exchangeable into or exercisable equity securities of Merger Partner); provided further that any interests issued to Merger Partner Support Holders to pursuant to clause (i) or (ii) shall be "Subject Units" under the Merger Partner Support Agreement pursuant to documentation reasonably satisfactory to Public Company;
- (b) split, combine or reclassify any equity interests other than the conversion of preferred units of Merger Partner to voting common units of Merger Partner as contemplated by the Merger Partner Member Approval and in a manner consistent with the Allocation Schedule;
- (c) create, incur or assume any Indebtedness (other than (x) interest accrued with respect to Indebtedness outstanding as of the date hereof in accordance with its terms, (y) new Indebtedness for borrowed money not exceeding \$5,000,000 in the aggregate, or (z) Indebtedness payable in connection with leases for equipment); (ii) assume, guarantee, endorse or otherwise agree to be liable (whether directly, contingently or otherwise) for the obligations of any other Person, other than for Subsidiaries of Merger Partner; or (iii) make any loans, advances or capital contributions to, or investments in, any other Person (other than investments of cash in cash equivalents in the Ordinary Course of Business);
- (d) except as required to comply with applicable Law or pursuant to agreements, plans or arrangements existing on the date hereof, (i) adopt, enter into, terminate or amend any employment or severance plan, agreement or arrangement, any Merger Partner Plan or any collective bargaining agreement, that in each case, would result in materially increased costs to Merger Partner or Public Company after the Closing, (ii) increase the compensation or fringe benefits of, or pay any bonus to, any director, officer, employee or consultant, except for increases in compensation consistent with the Ordinary Course of Business and in no event more than 20 percent or that would require disclosure in the Registration Statement after it is filed with the SEC, (iii) amend or accelerate the payment, right to payment or vesting of any compensation or benefits, including any outstanding equity awards, that in each case, would result in materially

increased costs to Merger Partner or Public Company after the Closing, (iv) except as contemplated in any other provision of this Section 5.1(d), pay any material benefit not provided for as of the date of this Agreement under any Merger Partner Plan, or the removal of existing restrictions in any benefit plans or agreements or awards made thereunder, or (v) take any action to fund any Merger Partner Plan other than the payment of premiums due or contributions owed in the Ordinary Course of Business;

- (e) acquire, sell, lease, in-license, out-license, abandon, allow to lapse, agree to coexist with respect to or otherwise dispose of any material assets or property (including any Intellectual Property or any shares or other equity interests in or securities of any Subsidiary or any other Person), in each case, other than in the Ordinary Course of Business;
- (f) mortgage or pledge any of its material property or material assets or enter into an agreement that subjects any such material property or material assets to any Lien other than Permitted Liens;
- (g) form any Subsidiary or acquire any equity interest or other interest in any other Person (other than investments of cash and cash equivalents in the Ordinary Course of Business) or enter into a joint venture with any other Person;
- (h) forgive any loans to any Person, including its employees, officers, directors or Affiliates, other than the settlement of accounts receivable in the Ordinary Course of Business;
- (i) change the nature or scope of its business being carried on as of the date of this Agreement in any material respect or commence any new business not being ancillary or incidental to such business;
- (j) change its accounting methods, principles or practices in any material respect, except insofar as may be required by a generally applicable change in GAAP or applicable Law;
- (k) except as required by applicable Law, make or change any material Tax election, amend any material Tax Return, change an annual accounting period, enter into any closing agreement, waive or extend any statute of limitations with respect to Taxes, settle or compromise any Tax liability, claim or assessment, or surrender any right to claim a material refund of Taxes;
 - (1) settle any Legal Proceedings involving cash payments by Merger Partner in excess of \$1,000,000 in the aggregate;
- (m) amend its Organizational Documents (other than, for the avoidance of doubt, as contemplated by Section 5.16 or otherwise in connection with implementing the Transactions in a manner consistent with this Agreement);
 - (n) make or commit to make capital expenditures in excess of \$4,000,000 in the aggregate; or

(o) agree in writing or otherwise to take any of the foregoing actions.

Nothing contained in this Agreement shall give Public Company, directly or indirectly, the right to control or direct the operations of Merger Partner and its Subsidiaries prior to the Merger Partner Merger Effective Time. Prior to the Merger Partner Merger Effective Time, the Merger Partner and its Subsidiaries shall exercise, consistent with the terms and conditions of this Agreement, complete unilateral control and supervision over its business operations

- 5.2 Operation of Public Company's Business. Except as contemplated by this Agreement, in connection with implementing the Transactions in a manner consistent with this Agreement, as set forth on Section 5.2 of Public Company Disclosure Schedule or as required by applicable Law (including any COVID-19 Measures), during the Pre-Closing Period, without the written consent of Merger Partner (which consent shall not be unreasonably withheld, conditioned or delayed), Public Company shall use commercially reasonable efforts to, and shall cause each Subsidiary to use commercially reasonable efforts to, conduct its operations only in the Ordinary Course of Business and in compliance with all applicable Laws in all material respects and take all necessary steps (including timely making all filings required by the SEC) to prepare for the filing of the annual meeting proxy statement (including all required Form 10-K, Item III information). Without limiting the generality of the foregoing, during the Pre-Closing Period and except as set forth on Section 5.2 of Public Company Disclosure Schedule, or in connection with implementing the Transactions in a manner consistent with this Agreement, Public Company shall not, and shall cause each Subsidiary not to, without the written consent of Merger Partner (such consent shall not be unreasonably withheld, conditioned or delayed):
- (a) Issue, grant, pledge or sell, or authorize the issuance of, any equity interests in Public Company or any of its Subsidiaries or any options, warrants or rights to acquire any such equity interests, or other securities or any other instrument convertible into or exchangeable for any equity interests of Public Company or its Subsidiaries, or declare, accrue, set aside or pay any dividend or make any other distribution in respect of its equity interests, or repurchase or redeem any equity interests in Public Company or any of its Subsidiaries;
- (b) split, combine or reclassify any equity interests, effect any merger, consolidation, share exchange, business combination, recapitalization or similar transaction other than the Mergers;
- (c) (i) create, incur or assume any Indebtedness (other than interest accrued with respect to Indebtedness outstanding as of the date hereof in accordance with its terms); (ii) assume, guarantee, endorse or otherwise agree to be liable (whether directly, contingently or otherwise) for the obligations of any other Person; or (iii) make any loans, advances or capital contributions to, or investments in, any other Person (other than investments of cash in cash equivalents in the Ordinary Course of Business);
 - (d) hire, engage or offer to hire any new officers, directors, employees or consultants;

- (e) except as required to comply with applicable Law or pursuant to agreements, plans or arrangements existing on the date hereof and disclosed in Section 5.2 of Public Company Disclosure Schedule, (i) adopt, enter into, terminate or amend any employment or severance plan, agreement or arrangement, any Public Company Benefit Plan or any collective bargaining agreement, (ii) increase the compensation or fringe benefits of, or pay any bonus to, any director, officer, employee or consultant, (iii) amend or accelerate the payment, right to payment or vesting of any compensation or benefits, including any outstanding Public Company Stock Options, Public Company RSUs, or Public Company PSUs, (iv) except as contemplated in any other provision of this Section 5.2, pay any material benefit not provided for as of the date of this Agreement under any Public Company Benefit Plan, or the removal of existing restrictions in any benefit plans or agreements or awards made thereunder, or (v) take any action to fund any Public Company Benefit Plan other than the payment of premiums due or contributions owed in the Ordinary Course of Business;
- (f) acquire, sell, lease, in-license, out-license or dispose of any material assets or material property (including any Intellectual Property or any shares or other equity interests in or securities of any Subsidiary or any other Person), other than in the Ordinary Course of Business;
- (g) mortgage or pledge any of its material property or material assets or enter into an agreement that subjects any such material property or material assets to any Lien other than Permitted Liens;
- (h) discharge or satisfy any Lien other than in the Ordinary Course of Business or as required by Law or a contract existing on the date of this Agreement or entered into in compliance with this Agreement;
- (i) form any Subsidiary or acquire any equity interest or other interest in any other Person (other than investments of cash and cash equivalents in the Ordinary Course of Business) or enter into a joint venture with any other Person;
 - (j) other than as contemplated by the transactions set forth in this Agreement, amend its Organizational Documents;
- (k) (i) materially change pricing or royalties or other payments set or charged by Public Company or any of its Subsidiaries to its customers or licensees or (ii) agree to materially change pricing or royalties or other payments set or charged by Persons who have licensed rights to Intellectual Property to Public Company or any of its Subsidiaries;
 - (l) accelerate collection of accounts receivable or delay payment of accounts payable, other than in accordance with GAAP;
- (m) forgive any loans to any Person, including its employees, officers, directors or Affiliates, other than the settlement of accounts receivable in the Ordinary Course of Business;
- (n) change the nature or scope of its business being carried on as of the date of this Agreement in any material respect or commence any new business not being ancillary or incidental to such business;

- (o) change its accounting methods, principles or practices in any material respect, except insofar as may be required by a generally applicable change in GAAP or applicable Law;
- (p) except as required by applicable Law, make or change any material Tax election, amend any material Tax Return, change an annual accounting period, enter into any closing agreement, waive or extend any statute of limitations with respect to Taxes, settle or compromise any Tax liability, claim or assessment, or surrender any right to claim a material refund of Taxes;
 - (q) make or commit to make any capital expenditure in excess of \$25,000 per item or \$50,000 in the aggregate;
- (r) institute or settle any Legal Proceeding (other than to enforce the terms of this Agreement and the other agreements relating to the transaction or otherwise in accordance with Section 7.4); or
 - (s) agree in writing or otherwise to take any of the foregoing actions.
- 5.3 Operation of Blockers. During the Pre-Closing Period and except (a) as set forth on Section 5.3 of Merger Partner Disclosure Schedule, (b) in connection with implementing the Transactions in a manner consistent with this Agreement, (c) the distribution of cash or other liquid amounts held by such Blocker as of the date of this Agreement or received by such Blocker in a Merger Partner distribution permitted under Section 5.1, (d) in the Ordinary Course of Business, or (e) as required by the terms of this Agreement, each Blocker shall not, without the written consent of Public Company (such consent shall not be unreasonably withheld, conditioned or delayed) conduct any business, take any action or enter into, amend or terminate any agreement or commitment. Further, during the Pre-Closing Period, each Blocker, to the extent it has not already done so, shall file all Tax Returns due on or prior to the Closing Date and pay all Taxes reasonably owing in connection with such Tax Returns.

5.4 Public Company No Solicitation.

(a) Public Company agrees that, during the Pre-Closing Period, it shall not, and shall not authorize any of its Representatives to (and it shall use its reasonable best effort to cause its Representatives not to), directly or indirectly: (i) solicit, initiate or knowingly encourage, induce or facilitate the communication, making, submission or announcement of any Acquisition Proposal or Acquisition Inquiry or take any action that could reasonably be expected to lead to an Acquisition Proposal or Acquisition Inquiry; (ii) furnish any information regarding Public Company to any Person for the purpose of encouraging, or in response to, an Acquisition Proposal or Acquisition Inquiry, provided that the foregoing clause (ii) shall not prohibit filings required by applicable securities Laws, including but not limited to Section 15(c) or 15(d) of the Exchange Act, or stock exchange rule, or otherwise directing such Person to Public Company's SEC filings; (iii) engage in discussions (other than to inform any Person of the existence of the provisions in this Section 5.4) or negotiations with any Person with respect to any Acquisition Proposal or Acquisition Inquiry; (iv) approve, endorse or recommend any Acquisition Proposal (subject to Section 5.9); (v) execute or enter into any letter of intent or any Contract

contemplating or otherwise relating to any Acquisition Transaction (other than a confidentiality agreement permitted under this Section 5.4(a)); (vi) publicly propose to do any of the foregoing; or (vii) agree, resolve or commit (or, for the avoidance of doubt, Public Company Board or any committee thereof to resolve, agree or commit) to do any of the foregoing; provided, however, that, notwithstanding anything contained in this Section 5.4 and subject to compliance with this Section 5.4, prior to obtaining the Public Company Stockholder Approval, Public Company may furnish non-public information regarding Public Company to, and enter into discussions or negotiations with, any Person in response to an unsolicited bona fide written Acquisition Proposal by such Person, which Public Company Board determines in good faith, after consultation with Public Company's outside financial advisors and outside legal counsel, constitutes, or would be reasonably likely to result in, a Superior Offer (and is not withdrawn) if: (A) neither Public Company nor any of its Representatives shall have materially breached this Section 5.4, (B) Public Company Board concludes in good faith based on the advice of outside legal counsel, that the failure to take such action would be reasonably likely to be inconsistent with the fiduciary duties of Public Company Board under applicable Law; (C) at least two (2) Business Days prior to initially furnishing any such nonpublic information to, or entering into discussions with, such Person, Public Company gives Merger Partner written notice of the identity of such Person and of Public Company's intention to furnish nonpublic information to, or enter into discussions with, such Person; (D) Public Company receives from such Person an executed confidentiality agreement containing provisions (including non-disclosure, use restrictions, non-solicitation and no-hire provisions), in the aggregate, at least as favorable to Public Company as those contained in the Confidentiality Agreement; and (E) at least two (2) Business Days prior to furnishing any such non-public information to such Person, Public Company furnishes such non-public information to Merger Partner (to the extent such information has not been previously furnished by Public Company to Merger Partner). Without limiting the generality of the foregoing, Public Company acknowledges and agrees that, in the event any Representative of Public Company (whether or not such Representative is purporting to act on behalf of Public Company) takes any action that, if taken by Public Company, would constitute a breach of this Section 5.4, the taking of such action by such Representative shall be deemed to constitute a breach of this Section 5.4 by Public Company for purposes of this Agreement.

(b) If Public Company or any Representative of Public Company receives an Acquisition Proposal or Acquisition Inquiry at any time during the Pre-Closing Period, then Public Company shall promptly (and in any event within 24 hours) advise Merger Partner orally and in writing of (i) the receipt of such Acquisition Proposal or Acquisition Inquiry, (ii) any non-public information provided to a Person who has made an Acquisition Proposal or Acquisition Inquiry in response to a request from such Person, (iii) the identity of the Person making or submitting such Acquisition Proposal or Acquisition Inquiry, and (iv) the material terms thereof as well as complete copies of any written Acquisition Proposals, Acquisition Inquiries or any other written communications from such Person or its Representatives, including any proposed agreements, and Public Company thereafter shall keep Merger Partner reasonably informed, on a reasonably current basis, with respect to the status and material terms of any such Acquisition Proposal or Acquisition Inquiry, including informing Merger Partner on a reasonably current basis (and, in any event, within 24 hours) of any material amendment or modification or proposed material amendment or modification Proposal or Acquisition Inquiry.

(c) Public Company shall immediately cease and cause to be terminated any existing discussions, negotiations and communications with any Person that relate to any Acquisition Proposal or Acquisition Inquiry that has not already been terminated as of the date of this Agreement and request the destruction or return of any non-public information of Public Company provided to such Person as soon as practicable after the date of this Agreement.

5.5 Merger Partner Non-Solicitation.

(a) Merger Partner agrees that, during the Pre-Closing Period, neither it nor any of its Subsidiaries shall, nor shall it or any of its Subsidiaries authorize any of their respective Representatives to (and it and each of its Subsidiaries shall use its reasonable best efforts to cause its Representatives not to), directly or indirectly: (i) solicit, initiate or knowingly encourage, induce or facilitate the communication, making, submission or announcement of any Acquisition Proposal or Acquisition Inquiry or take any action that could reasonably be expected to lead to an Acquisition Proposal or Acquisition Inquiry; (ii) furnish any information regarding Merger Partner or any of its Subsidiaries to any Person for the purpose of encouraging, or in response to, an Acquisition Proposal or Acquisition Inquiry, provided that the foregoing shall not prohibit filings required by Law, including but not limited to Section 15(c) or 15(d) of the Exchange Act, or stock exchange rule; (iii) engage in discussions (other than to inform any Person of the existence of the provisions in this Section 5.5) or negotiations with any Person with respect to any Acquisition Proposal or Acquisition Inquiry; (iv) approve, endorse or recommend any Acquisition Proposal; (v) execute or enter into any letter of intent or any Contract contemplating or otherwise relating to any Acquisition Transaction; (vi) publicly propose to do any of the foregoing; or (vii) agree, resolve or commit to do any of the foregoing; provided, however, that notwithstanding anything contained in this Section 5.5 and subject to compliance with this Section 5.5, prior to obtaining the Merger Partner Member Approval applicable to it, Merger Partner may furnish non-public information regarding it to, and enter into discussions or negotiations with, any Person in response to an unsolicited bona fide written Acquisition Proposal by such Person, which its board of managers determines in good faith, after consultation with its outside financial advisors and outside legal counsel, constitutes, or would be reasonably likely to result in, a Superior Offer (and is not withdrawn) if: (A) none of Merger Partner nor any of its Representatives shall have materially breached this Section 5.5; (B) the board of managers of Merger Partner concludes in good faith based on the advice of outside legal counsel, that the failure to take such action would be reasonably likely to be inconsistent with the fiduciary duties of such board of managers under applicable Law; (C) at least two (2) Business Days prior to initially furnishing any such nonpublic information to, or entering into discussions with, such Person, Merger Partner gives Public Company written notice of the identity of such Person and of Merger Partner's intention to furnish nonpublic information to, or enter into discussions with, such Person; (D) it receives from such Person an executed confidentiality agreement containing provisions (including non-disclosure, use restrictions, non-solicitation and no-hire provisions), in the aggregate, at least as favorable to it as those contained in the Confidentiality Agreement; and at least two (2) Business Days prior to furnishing any such non-public information to such Person, Merger Partner furnishes such non-public information to Public Company (to the extent such information has not been previously furnished by Merger

Partner to Public Company). Without limiting the generality of the foregoing, Merger Partner acknowledges and agrees that, in the event any Representative of it or any of its Subsidiaries (whether or not such Representative is purporting to act on behalf of it or any of its Subsidiaries) takes any action that, if taken by it, would constitute a breach of this Section 5.5, the taking of such action by such Representative shall be deemed to constitute a breach of this Section 5.5 by Merger Partner for purposes of this Agreement.

- (b) If Merger Partner or its Subsidiaries or Representatives receives an Acquisition Proposal or Acquisition Inquiry at any time during the Pre-Closing Period, then it shall promptly (and in any event within 24 hours) advise Public Company orally and in writing of (i) the receipt of such Acquisition Proposal or Acquisition Inquiry, (ii) any non-public information provided to a Person who has made an Acquisition Proposal or Acquisition Inquiry in response to a request from such Person, (iii) the identity of the Person making or submitting such Acquisition Proposal or Acquisition Inquiry, and (iv) the material terms thereof as well as complete copies of any written Acquisition Proposals, Acquisition Inquiries or any other written communications from such Person or its Representatives, including any proposed agreements, and Merger Partner thereafter shall keep Public Company reasonably informed, on a reasonably current basis, with respect to the status and material terms of any such Acquisition Proposal or Acquisition Inquiry, including informing Public Company on a reasonably current basis (and, in any event, within 24 hours) of any material amendment or modification or proposed material amendment or modification Proposal or Acquisition Inquiry.
- (c) Merger Partner shall immediately cease and cause to be terminated any existing discussions, negotiations and communications with any Person that relate to any Acquisition Proposal or Acquisition Inquiry that has not already been terminated as of the date of this Agreement and request the destruction or return of any non-public information of Merger Partner or its Subsidiaries provided to such Person as soon as practicable after the date of this Agreement.

5.6 Notification of Certain Matters.

(a) During the Pre-Closing Period, Merger Partner shall promptly (and in no event later than one (1) Business Day after Merger Partner becomes aware of same) notify Public Company (and, if in writing, furnish copies of any relevant documents) if any of the following occurs: (i) any notice or other communication is received from any Person alleging that the consent of such Person is or may be required in connection with any of the Transactions; (ii) any Legal Proceeding against or involving or otherwise affecting Merger Partner, to the Knowledge of Merger Partner, a Blocker or any of their respective Subsidiaries is commenced, or, to the Knowledge of Merger Partner, threatened against Merger Partner, a Blocker or any of their respective Subsidiaries or, to the Knowledge of Merger Partner, any director or officer of Merger Partner, a Blocker or any of their respective Subsidiaries, (iii) any inaccuracy in any representation or warranty made by it in this Agreement; or (iv) the failure of the Merger Partner or, to the Knowledge of Merger Partner, any Blocker to comply with any of their respective covenants or obligations in this Agreement; in the case of (iii) and (iv) that will or could reasonably be expected to result in the failure by Merger Partner or any Blocker to satisfy any of the conditions set forth in Section 6.1. Any failure to timely provide a notification under clauses (iii) and (iv) above shall be disregarded for purposes of determining compliance with Section

6.1(c) and for all purposes under Section 8.1. No notification given to Public Company pursuant to this Section 5.6(a) shall change, limit or otherwise affect any of the representations, warranties, covenants or obligations of Merger Partner, the Blockers or any of their respective Subsidiaries contained in this Agreement or Merger Partner Disclosure Schedule for purposes of Article VI, as applicable.

(b) During the Pre-Closing Period, Public Company shall promptly (and in no event later than one (1) Business Day after Public Company becomes aware of same) notify Merger Partner (and, if in writing, furnish copies of any relevant documents) if any of the following occurs: (i) any notice or other communication is received from any Person alleging that the consent of such Person is or may be required in connection with any of the Transactions; (ii) any Legal Proceeding against or involving or otherwise affecting Public Company or its Subsidiaries is commenced, or, to the Knowledge of Public Company, threatened against Public Company or, to the Knowledge of Public Company, any director or officer of Public Company, (iii) any inaccuracy in any representation or warranty made by it in this Agreement; or (iv) the failure of Public Company or Merger Subsidiary to comply with any of their respective covenants or obligations in this Agreement; in the case of (iii) and (iv) that will or could reasonably be expected to result in the failure by Public Company or Merger Subsidiary to satisfy any of the conditions set forth in Section 6.2. Any failure to timely provide a notification under clauses (iii) and (iv) above shall be disregarded for purposes of determining compliance with Section 6.2(c) and for all purposes under Section 8.1. No notification given to Merger Partner pursuant to this Section 5.6(b) shall change, limit or otherwise affect any of the representations, warranties, covenants or obligations of Public Company contained in this Agreement or Public Company Disclosure Schedule for purposes of Article VI, as applicable.

5.7 Proxy Statement/Prospectus; Registration Statement.

(a) As promptly as practical after the execution of this Agreement and contingent upon receipt from Merger Partner of the information required by the following sentence, Public Company, with the cooperation of Merger Partner, shall prepare and file with the SEC the Registration Statement, in which the Proxy Statement/Prospectus will be included. Merger Partner shall (i) provide to Public Company as promptly as practical (and in any event by January 22, 2023) all information, including financial statements of Merger Partner and its Subsidiaries for the fiscal years 2021 and 2020 and for the interim period ended September 30, 2022 and 2021 and descriptions of its business and financial condition, as Public Company may reasonably request for preparation of the Registration Statement and the Proxy Statement/Prospectus and (ii) cause the timely cooperation of its independent public accountants in connection with the preparation and filing of the Registration Statement and the Proxy Statement/Prospectus, including by causing such accountants to provide a consent to the inclusion of such accountant's reports in respect of the audited financial statements of Merger Partner in the Registration Statement and/or in the Proxy Statement/Prospectus (as applicable) and to the reference to such accountant firm as an "expert" therein. Each Blocker shall provide to Public Company as promptly as practical (and in any event by January 22, 2023) all information about itself as Public Company may reasonably request and which is required to be included in the Registration Statement and the Proxy Statement/Prospectus. Public Company shall respond to any comments of the SEC and shall use reasonable best efforts to have the Registration Statement declared effective under the Securities Act as promptly as practicable

after such filing, and Public Company shall cause the Proxy Statement/Prospectus to be mailed to its stockholders at the earliest practicable time after the Registration Statement is declared effective under the Securities Act. Public Company shall notify Merger Partner promptly upon the receipt of any comments from the SEC or its staff and of any request by the SEC or its staff for amendments or supplements to the Registration Statement, the Proxy Statement/Prospectus or any filing pursuant to Section 6.2(b) or for additional information and shall promptly supply Merger Partner with copies of all correspondence between Public Company or any of its representatives, on the one hand, and the SEC, or its staff, on the other hand, with respect to the Registration Statement, the Proxy Statement/Prospectus, the Merger or any filing pursuant to Section 5.7(b). Public Company shall use commercially reasonable efforts to cause all documents that it is responsible for filing with the SEC under this Section 5.7 to comply in all material respects with all applicable requirements of Law and the rules and regulations promulgated thereunder. Whenever either Public Company or Merger Partner shall become aware of the occurrence of any event which is required to be set forth in an amendment or supplement to the Proxy Statement/Prospectus, the Registration Statement or any filing pursuant to Section 6.2(b), Public Company or Merger Partner, as the case may be, shall promptly inform the others of such occurrence and cooperate in filing with the SEC or its staff, and/or mailing to stockholders of Public Company, such amendment or supplement.

- (b) Notwithstanding anything to the contrary stated above, prior to filing and mailing, as applicable, the Registration Statement or Proxy Statement/Prospectus (or any amendment or supplement thereto) or responding to any comments of the SEC with respect thereto, Public Company shall provide Merger Partner and each Blocker a reasonable opportunity to review and comment on such document or response and shall consider in good faith any such comments proposed by Merger Partner or each Blocker (but only with respect to matters pertaining to such Blocker). Public Company will advise Merger Partner and each Blocker, promptly after Public Company receives notice thereof, of the time when the Registration Statement has become effective or any supplement or amendment has been filed, of the issuance of any stop order or the suspension of the qualification of Public Company Common Stock for offering or sale in any jurisdiction, of the initiation or threat of any proceeding for any such purpose, or of any request by the SEC for the amendment or supplement of the Registration Statement or for additional information.
- (c) Public Company and Merger Partner and, to the extent applicable, the Blockers shall promptly make all necessary filings of such party with respect to the Mergers and the issuance of the Public Company Class A Common Stock and Public Company Class B Common Stock constituting Merger Consideration under the Securities Act, the Exchange Act, applicable state blue sky laws and the rules and regulations thereunder.
- 5.8 <u>Nasdaq Listing</u>. Public Company agrees to use its reasonable best efforts: (a) to maintain its existing listing on Nasdaq and to obtain approval of the listing of the combined corporation on Nasdaq; (b) to the extent required by the rules and regulations of Nasdaq, to prepare and submit to Nasdaq a notification form for the listing of the shares of Public Company Class A Common Stock to be issued in connection with the Mergers; (c) if not previously implemented, to prepare and timely submit to Nasdaq a notification form of any reverse stock split effected with the Merger Party's consent and, to the extent required to effect the foregoing, to submit a copy of the amendment to Public Company's certificate of incorporation to effect

such reverse stock split certified by the Secretary of State of the State of Delaware to Nasdaq; (d) to the extent required by Nasdaq Listing Rule 5110, to file an initial listing application for the Public Company Class A Common Stock on Nasdaq and to cause such Nasdaq initial listing application ("Nasdaq Listing Application") to be approved; and (e) in the event of receipt of a Nasdaq delisting determination, Public Company shall request a hearing to appeal the delisting determination and shall pay the appropriate fee to Nasdaq to appeal the delisting determination. Public Company shall promptly inform Merger Partner of all material verbal or written communications between Nasdaq and Public Company or its representatives. Public Company and Merger Partner will use commercially reasonable efforts to coordinate with respect to compliance with Nasdaq rules and regulations. Merger Partner will cooperate with Public Company to cause the Nasdaq Listing Application to be approved and shall promptly furnish to Public Company all information concerning Merger Partner, the Blockers and their respective equity holders that may be required or reasonably requested in connection with any action contemplated by this Section 5.8.

5.9 Public Company Stockholders' Meeting.

- (a) As promptly as reasonably practicable after the Registration Statement is declared effective by the SEC, Public Company shall take all action reasonably necessary under applicable Law to call, give notice of and hold a meeting of the holders of Public Company Common Stock for the purpose of seeking approval of (i) the Blocker Mergers, (ii) the issuance of Public Company Class A Common Stock and Public Company Class B Common Stock under this Agreement and in accordance with the Nasdaq Listing Rules; (iii) the amendment of Public Company's Organizational Documents, in the form of the Amended Certificate and the Amended By-Laws, to, among other things, authorize the shares of Public Company Class A Common Stock and Public Company Class B Common Stock to be issued pursuant to this Agreement in connection with the Mergers and (iv) any other proposals the Parties reasonably deem necessary or desirable to consummate the Transactions (the matters contemplated by the foregoing Sections 5.9(a) (i), (ii) and (iii) are collectively referred to as the "Public Company Stockholder Matters," and the matters contemplated by Section 5.9(a) (iv) are referred to herein as, the "Other Public Company Stockholder Matters," and such meeting, the "Public Company Stockholders' Meeting").
- (b) Public Company will take all necessary actions to duly call, give notice of, convene and hold the Public Company Stockholders' Meeting as promptly as practicable (but in no event later than sixty (60) days) after the Registration Statement is declared effective by the SEC. Absent a Public Company Board Adverse Recommendation Change pursuant to Section 5.9(c), Public Company shall use its reasonable best efforts to solicit the holders of Public Company Common Stock to obtain the Public Company Stockholder Approval. Public Company shall take reasonable measures to ensure that all proxies solicited in connection with Public Company Stockholders' Meeting are solicited in compliance with all applicable Laws. Notwithstanding anything to the contrary contained herein, if on the date of Public Company Stockholders' Meeting, or a date preceding the date on which Public Company Stockholders' Meeting is scheduled, Public Company reasonably believes that (i) it will not receive proxies sufficient to obtain approval of the Public Company Stockholder Matters, whether or not a quorum would be present, or (ii) it will not have sufficient shares of Public Company Common Stock represented (whether in Person or by proxy) to constitute a quorum necessary to conduct

the business of Public Company Stockholders' Meeting, Public Company may make one or more successive postponements or adjournments of Public Company Stockholders' Meeting as long as the date of Public Company Stockholders' Meeting is not postponed or adjourned more than an aggregate of 60 days in connection with any postponements or adjournments. Public Company and Merger Partner shall coordinate and reasonably cooperate in connection with soliciting and obtaining the Public Company Stockholder Approval, including the preparation and delivery of any information relating to the parties hereto, the Mergers and the Transactions as may be reasonably requested by each of the Public Company and Merger Partner.

- (c) Public Company agrees that, subject to Section 5.9(d): (i) Public Company Board shall recommend that the holders of Public Company Common Stock vote to approve Public Company Stockholder Matters and the Other Public Company Stockholder Matters and (ii) the Proxy Statement/Prospectus shall include a statement to the effect that Public Company Board recommends that Public Company's stockholders vote to approve Public Company Stockholder Matters and the Other Public Company Stockholder Matters (the recommendation of Public Company Board with respect to Public Company Stockholder Matters and the Other Public Company Stockholder Matters being referred to as the "Public Company Board Recommendation"). Neither Public Company Board nor any committee thereof shall (x) withhold, amend, qualify, withdraw or modify (and Public Company Board or any committee thereof shall not resolve to or publicly propose to withhold, amend, qualify, withdraw or modify) the Public Company Board Recommendation in a manner adverse to Merger Partner; or (y) within ten (10) Business Days' of Merger Partner's written request to do so, fail to recommend after the commencement of an Acquisition Proposal through a tender or exchange offer pursuant to Rule 14d-2 under the Exchange Act for outstanding shares of Public Company Common Stock, against acceptance of such tender offer or exchange offer by its stockholders (which request may only be made once with respect to any such Acquisition Proposal and each material modification thereto) (the actions set forth in the foregoing clauses collectively, a "Public Company Board Adverse Recommendation Change").
- (d) Notwithstanding anything to the contrary contained in Section 5.4(a) of this Agreement, if at any time prior to the approval of Public Company Stockholder Matters at Public Company Stockholders' Meeting by the Required Public Company Stockholder Vote:
- (i) if Public Company has received a written Acquisition Proposal (which Acquisition Proposal did not arise in connection with a breach of Section 5.3) from any Person that has not been withdrawn and after consultation with outside legal counsel, Public Company Board shall have determined, in good faith, that such Acquisition Proposal is a Superior Offer, Public Company Board may make a Public Company Board Adverse Recommendation Change if and only if all of the following apply: (A) Public Company Board determines in good faith, after consultation with Public Company's outside legal counsel, that the failure to do so would be reasonably likely to be inconsistent with the fiduciary duties of Public Company Board to Public Company's stockholders under applicable Law; (B) Public Company shall have given Merger Partner prior written notice of its intention to consider making a Public Company Board Adverse Recommendation Change at least three (3) Business Days prior to making any such Public Company Board Adverse Recommendation Change (a "Determination Notice") (which notice shall not constitute a Public Company Board Adverse

Recommendation Change); and (C) (1) Public Company shall have provided to Merger Partner a copy of such Acquisition Proposal in accordance with Section 5.4(b), (2) Public Company shall have given Merger Partner the three (3) Business Days after the Determination Notice to propose revisions to the terms of this Agreement or make another proposal and shall have made its Representatives reasonably available to negotiate in good faith with Merger Partner (to the extent Merger Partner desires to negotiate) with respect to such proposed revisions or other proposal, if any, and (3) after considering the results of any such negotiations and giving effect to the proposals made by Merger Partner, if any, after consultation with outside legal counsel, Public Company Board shall have determined, in good faith, that such Acquisition Proposal is a Superior Offer and that the failure to make Public Company Board Adverse Recommendation Change would be reasonably likely to be inconsistent with the fiduciary duties of Public Company Board to Public Company's stockholders under applicable Law. For the avoidance of doubt, the provisions of this Section 5.9(d)(i) shall also apply to any material change to the facts and circumstances relating to such Acquisition Proposal and require a new Determination Notice, except that the references to three (3) Business Days shall be deemed to be two (2) Business Days.

(ii) Other than in connection with an Acquisition Proposal, Public Company Board may make a Public Company Board Adverse Recommendation Change in response to a Public Company Change in Circumstance, if and only if: (A) Public Company Board determines in good faith, after consultation with Public Company's outside legal counsel, that the failure to do so would be reasonably likely to be inconsistent with the fiduciary duties of Public Company Board to Public Company's stockholders under applicable Law; (B) Public Company shall have given Merger Partner a Determination Notice at least three (3) Business Days prior to making any such Public Company Board Adverse Recommendation Change; and (C) (1) Public Company shall have specified Public Company Change in Circumstance in reasonable detail, (2) Public Company shall have given Merger Partner the three (3) Business Days after the Determination Notice to propose revisions to the terms of this Agreement or make another proposal, and shall have made its Representatives reasonably available to negotiate in good faith with Merger Partner (to the extent Merger Partner desires to do so) with respect to such proposed revisions or other proposal, if any, and (3) after considering the results of any such negotiations and giving effect to the proposals made by Merger Partner, if any, after consultation with outside legal counsel, Public Company Board shall have determined, in good faith, that the failure to make Public Company Board Adverse Recommendation Change in response to such Public Company's stockholders under applicable Law. For the avoidance of doubt, the provisions of this Section 5.9(d)(ii) shall also apply to any material change to the facts and circumstances relating to such Public Company Change in Circumstance and require a new Determination Notice, except that the references to three (3) Business Days shall be deemed to be two (2) Business Days.

(e) Nothing contained in Section 5.3(a) shall prohibit Public Company or Public Company Board from (i) complying with Rules 14d-9 and 14e-2(a) promulgated under the Exchange Act, (ii) issuing a "stop, look and listen" communication or similar communication of the type contemplated by Section 14d-9(f) under the Exchange Act or (iii) otherwise making any disclosure to Public Company's stockholders; that Public Company Board determines in good faith, after consultation with its outside legal counsel, is required under applicable securities Laws; <u>provided</u> that in each of cases (i), (ii) and (iii) in no event shall Public Company make a Public Company Board Adverse Recommendation Change.

5.10 Merger Partner and Blocker Member Approval.

- (a) Merger Partner shall use reasonable best efforts to secure the Written Consent necessary to secure the Merger Partner Member Approval as promptly as practical following the date that the SEC declares the Registration Statement effective and each Blocker shall secure the Written Consents necessary to secure the applicable Blocker Approvals as promptly as practical following the date that the SEC declares the Registration Statement effective. As promptly and practical following the receipt of Merger Partner Member Approval and Blocker Approvals, Merger Partner shall deliver to Public Company copy of the Written Consents evidencing the Merger Partner Member Approval and Blocker Approvals.
- (b) As promptly as practical following the date that the SEC declares the Registration Statement effective, Merger Partner shall mail or otherwise distribute the Disclosure Statement, in a form reasonably acceptable to Public Company, to the equity holders of Merger Partner and, to the extent required by Law, to the equity holders of any Blockers, and shall promptly inform Public Company of the date on which such Disclosure Statement (including the notices contained therein) was sent to such equity holders. The Disclosure Statement shall include a written notice to all members of Merger Partner and the Blockers that did not execute the Written Consents informing them that this Agreement and the Mergers were adopted and approved by the equityholders of Merger Partner and the Blockers. Public Company and its counsel shall be given an adequate opportunity to review and comment on the Disclosure Statement, and Merger Partner shall reflect all reasonable comments of Public Company or its counsel thereon. Public Company and Merger Partner shall coordinate and reasonably cooperate in connection with soliciting and obtaining the Merger Partner Member Approval, including the preparation and delivery of any information relating to the parties hereto, the Mergers and the Transactions as may be reasonably requested by each of the Public Company and Merger Partner.
- (c) Merger Partner shall ensure that the Disclosure Statement does not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading (provided that Merger Partner shall not be responsible for the accuracy or completeness of any information concerning Public Company or Merger Subsidiary furnished by Public Company in writing for inclusion in the Disclosure Statement).
- (d) Public Company shall ensure that any information furnished by Public Company to Merger Partner in writing for inclusion in the Disclosure Statement does not contain any untrue statement of a material fact or omit to state a material fact necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading.

5.11 Access to Information. During the Pre-Closing Period and subject to the terms of the Confidentiality Agreement, Public Company and Merger Partner shall each (and shall cause each of their respective Subsidiaries to) use commercially reasonable efforts to afford the officers, attorneys, accountants, tax advisors, lenders and other authorized representatives of the other party reasonable access upon reasonable notice and during normal business hours and without unreasonable interference with the operation of the business of such other party to all Personnel, offices, properties, books and records of such party and its Subsidiaries, so that the other party may have full opportunity to make such investigation as it shall desire to make of the management, business, properties and affairs of said party and the Subsidiaries. Public Company and Merger Partner shall (and shall cause each Subsidiary to) furnish to the other parties such financial and operating data and other information as to the business of said party and the Subsidiaries as the other parties shall reasonably request. Each Blocker shall furnish to Public Company and Merger Partner such financial and operating data and other information as to the business of such Blocker as the other parties shall reasonably request in connection with this Agreement and the transactions contemplated hereby but only to the extent that such information is reasonably available. Notwithstanding the foregoing, nothing herein will require a party or its Subsidiaries to (i) provide the other party with access or information that said party is expressly prohibited by applicable Law from granting or disclosing or that would breach such party's confidentiality obligations to a third party in effect as of the date of this Agreement, or (ii) take any action that would, in the advice of counsel, constitute a waiver of the attorney-client privilege or the attorney work product privilege in the event of a legal proceeding with the other party; provided, that in the event that a party or any Subsidiary relies on this sentence to withhold access or disclosure, said party shall, to the extent permitted by Law, applicable confidentiality obligations and the protection of such attorney-client privilege, promptly notify the other party of the nature of the withheld information and provide the other party of a reasonable opportunity to seek an appropriate remedy or waive compliance with the terms of this Agreement.

5.12 Closing Efforts.

- (a) Upon the terms and subject to the conditions of this Agreement, each of Public Company and Merger Partner shall each use its reasonable best efforts to take all actions and to do all things necessary, proper or advisable to consummate the Transactions to be completed at Closing as promptly as practicable, including using its reasonable best efforts to ensure that the conditions to the obligations of the other parties to consummate the Merger are satisfied.
- (b) Public Company and Merger Partner shall each use its reasonable best efforts to obtain, at its expense, all waivers, permits, consents, approvals or other authorizations from Governmental Entities or from third parties pursuant to the Public Company Contracts or Merger Partner Contracts (as applicable), and to effect all registrations, filings and notices with or to Governmental Entities, in each case, as may be required for such party to consummate the Transactions and to otherwise comply with all applicable Laws in connection with the consummation of the Transactions.
- 5.13 <u>Public Disclosure</u>. No party shall issue any press release or public announcement relating to the subject matter of this Agreement without the prior written approval of the other parties; <u>provided</u>, <u>however</u>, that (a) Merger Partner and Blockers each acknowledge and agree that Public Company (i) may issue, without the approval of any other party, an initial joint press release with respect to this Agreement and the matters contemplated hereby, in form and substance mutually agreed by Public Company, Merger Partner and each Blocker (provided that

such press release shall not name any Blocker or its Affiliate or investment adviser without the prior consent of such Blocker), (ii) intends to publicly file this Agreement with the SEC, and (iii) may seek confidential treatment under applicable SEC rules with respect to certain matters and terms contained in this Agreement; (b) Public Company or Merger Partner may make any public disclosure it believes in good faith is required by applicable Law or stock market rule (in which case the disclosing party shall use reasonable best efforts to advise the other parties and provide them with a copy of the proposed disclosure prior to making the disclosure); and (c) Public Company and its Affiliates shall not be bound by the provisions of this Section 5.13 following the Closing Date.

- 5.14 Antitrust. If at any time after the date hereof but prior to the Closing, Merger Partner determines, in its reasonable discretion after consultation with Public Company and its legal advisors, that the Transactions require filings under the HSR Act to be submitted with the applicable Governmental Entities, then as promptly as reasonably practicable following such determination, Public Company and Merger Partner shall cooperate in the preparation of such filings and, upon completion thereof (in form and substance mutually agreeable to Merger Partner and Public Company), each of Merger Partner and Public Company shall make or cause to be made any such required filings under the HSR Act.
- 5.15 <u>Warrant Restructuring</u>. From and after the date hereof until the date that is five (5) Business Days prior to the Closing Date or the earlier termination of this Agreement pursuant to Section 8.1, Merger Partner shall use commercially reasonable efforts to obtain any and all consents from each holder of Merger Partner Warrants necessary to effectuate the treatment of the Merger Partner Warrants set forth in Section 2.1.
- 5.16 <u>Transaction Documentation</u>. From the date hereof until the time when the Registration Statement becomes effective or the earlier termination of this Agreement pursuant to Section 8.1, Public Company and Merger Partner shall use reasonable best efforts to negotiate and settle on mutually agreeable final forms of the Tax Receivable Agreement, the Exchange Agreement, the Amended Certificate, the Amended By-Laws, the Amended Operating Agreement, the Registration Rights Agreement and the Asset Transfer Agreements as promptly as practicable following the date hereof.
- 5.17 Merger Partner Funding. No later than two (2) Business Days following Merger Partner's receipt of a written request from Public Company, Merger Partner shall make available to Public Company and its Subsidiaries, funds up to a maximum aggregate amount of \$3,000,000 in cash (the "Interim Funding"); provided, however, that in the event Merger Partner elects to extend the Outside Date pursuant to Section 8.1(b), the maximum aggregate amount of the Interim Funding shall automatically increase to \$4,000,000. The Interim Funding shall be made available on the terms and conditions consistent with those set forth on Exhibit E and on such other terms and conditions as are mutually agreed by Merger Partner and Public Company.
- 5.18 Section 16 Matters. Prior to the Closing, each of Public Company and Merger Partner, as applicable, shall use all reasonable efforts to approve in advance in accordance with the applicable requirements of Rule 16b-3 promulgated under the Exchange Act, the acquisition of Up-C Units, the exchange of Up-C Units for shares of Public Company Class A Common Stock (and any dispositions of Merger Partner Common Units or shares of Public Company Class B Common Stock thereby) resulting from the transactions contemplated by this Agreement by each officer or director (including by deputization) of the Public Company or the Merger Partner who is subject to Section 16 of the Exchange Act (or who will become subject to Section 16 of the Exchange Act) as a result of the transactions contemplated hereby.

ARTICLE VI CONDITIONS TO CONSUMMATION OF THE MERGERS

- 6.1 <u>Conditions to Obligations of Public Company and Merger Subsidiary.</u> The obligation of each of Public Company and Merger Subsidiary to consummate the Mergers is subject to the satisfaction of the following conditions precedent, each of which may be waived in writing in the sole discretion of Public Company:
- (a) the Fundamental Representations contained in Article III shall have been true and correct in all material respects as of the date of this Agreement and shall be true and correct in all material respects on and as of the Closing Date with the same force and effect as if made on and as of such date (except to the extent such representations and warranties are specifically made as of a particular date, in which case such representations and warranties shall be true and correct in all material respects as of such particular date); the representations and warranties contained in Article III (other than the Fundamental Representations contained in Article III) shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date except (i) in each case, or in the aggregate, where the failure to be true and correct would not reasonably be expected to result in a Merger Partner Material Adverse Effect (without giving effect to any references therein to any Merger Partner Material Adverse Effect or other materiality qualifications), or (ii) for those representations and warranties which address matters only as of a particular date (which representations shall have been true and correct, subject to the qualifications as set forth in the preceding clause (i), as of such particular date);
- (b) Each of Merger Partner and the Blockers shall have performed or complied with, in all material respects, its agreements and covenants required to be performed or complied with by it under this Agreement as of or prior to the Closing;
- (c) Public Company shall have received a certificate of an officer of Merger Partner certifying to the satisfaction of the conditions set forth in Sections 6.1(a) and (b);
- (d) no judgment, order, decree, stipulation or injunction shall be in effect that would reasonably be expected to (i) temporarily or permanently prevent or enjoin consummation of the Transactions, or (ii) cause the Transactions to be rescinded following consummation of such transaction;
- (e) there shall have occurred no Effect since the date of this Agreement that, individually or taken together with all other Effects, has had, or would reasonably be expected to result in, a Merger Partner Material Adverse Effect;
- (f) Public Company shall have received copies of the Written Consents evidencing the Merger Partner Member Approval and the Blocker Approvals;

- (g) Public Company shall have obtained Stockholder Approval;
- (h) The Registration Statement shall have become effective under the Securities Act and no stop order suspending the effectiveness of the Registration Statement shall have been issued and no Legal Proceeding for that purpose, and no similar Legal Proceeding with respect to the Proxy Statement/Prospectus, shall have been initiated or threatened in writing by the SEC or its staff that has not been withdrawn;
- (i) If filings under the HSR Act are required, as determined pursuant to Section 5.14, any applicable waiting period shall have expired or been terminated:
- (j) The Nasdaq Listing Application shall have been approved, and the shares of the Public Company Class A Common Stock to be issued as Merger Consideration shall have been approved for listing (subject to official notice of issuance) on Nasdaq;
- (k) Merger Partner shall have delivered to Public Company a copy of (i) the Tax Receivable Agreement, and the Exchange Agreement, each duly executed by Merger Partner and each Electing Member and (ii) the Registration Rights Agreement, duly executed by each Electing Member; and
- (l) Merger Partner shall have delivered to Public Company a copy of the Amended Operating Agreement, duly executed by Merger Partner and each Electing Member.
- 6.2 <u>Conditions to Obligations of Merger Partner and the Blockers</u>. The obligations of Merger Partner and the Blockers to consummate the Mergers is subject to the satisfaction of the following conditions precedent, each of which may be waived in writing in the sole discretion of Merger Partner:
- (a) the Fundamental Representations contained in Article IV shall have been true and correct in all material respects as of the date of this Agreement and shall be true and correct in all material respects on and as of the Closing Date with the same force and effect as if made on and as of such date (except to the extent such representations and warranties are specifically made as of a particular date, in which case such representations and warranties shall be true and correct in all material respects as of such particular date). The representations and warranties contained in Article IV (other than the Fundamental Representations contained in Article IV) shall have been true and correct as of the date of this Agreement and shall be true and correct on and as of the Closing Date with the same force and effect as if made on the Closing Date except (i) in each case, or in the aggregate, where the failure to be true and correct would not reasonably be expected to result in a Public Company Material Adverse Effect (without giving effect to any references therein to any Public Company Material Adverse Effect or other materiality qualifications), or (ii) for those representations and warranties which address matters only as of a particular date (which representations shall have been true and correct, subject to the qualifications as set forth in the preceding clause (i), as of such particular date);
- (b) each of Public Company and Merger Subsidiary shall have performed or complied, in all material respects, with its agreements and covenants required to be performed or complied with under this Agreement as of or prior to the Closing;

- (c) Public Company shall have received a certificate of an officer of Merger Partner certifying to the satisfaction of the conditions set forth in Sections 6.2(b) and (c);
- (d) no judgment, order, decree, stipulation or injunction shall be in effect, that would reasonably be expected to (i) temporarily or permanently prevent or enjoin prevent consummation of the Transactions, or (ii) cause the Transactions to be rescinded following consummation of such transaction;
- (e) there shall have occurred no Effect since the date of this Agreement that, individually or taken together with all other Effects, has had, or would reasonably be expected to result in, a Public Company Material Adverse Effect;
 - (f) Public Company shall have obtained the Public Company Stockholder Approval;
 - (g) Merger Partner shall have obtained the Written Consents evidencing the Merger Partner Member Approval and the Blocker Approvals;
- (h) Public Company shall have delivered to Merger Partner an amendment to the license agreement between Public Company and Cellectis S.A. in substantially the form attached hereto as Exhibit F;
- (i) The Registration Statement shall have become effective under the Securities Act and no stop order suspending the effectiveness of the Registration Statement shall have been issued and no Legal Proceeding for that purpose, and no similar Legal Proceeding with respect to the Proxy Statement/Prospectus, shall have been initiated or threatened in writing by the SEC or its staff that has not been withdrawn;
- (j) (i) the existing shares of Public Company Common Stock shall be listed on Nasdaq and not subject to any pending de-listing notice (except as set forth in the Public Company Disclosure Schedule) or de-listing appeal process that would not be remedied by the consummation of the Transactions, (ii) the Nasdaq Listing Application shall have been approved, and (iii) the shares of the Public Company Class A Common Stock to be issued as Merger Consideration shall have been approved for listing (subject to official notice of issuance) on Nasdaq;
- (k) The Support Agreements will continue to be in full force and effect as of immediately following the Merger Partner Merger Effective Time;
- (l) If filings under the HSR Act are required, as determined pursuant to Section 5.14, any applicable waiting period shall have expired or been terminated;
- (m) Merger Partner shall have received a written resignation, in a form reasonably satisfactory to Merger Partner, dated as of the Closing Date and effective as of the Merger Partner Merger Effective Time, executed by each of the officers and directors of Public Company that are not included in the Exhibit D-1 or Exhibit D-2, from their positions as such, of Public Company, including any characterization of the resigning officers cessation from employment;

- (n) Public Company shall have delivered to Merger Partner a copy of the Tax Receivable Agreement, the Exchange Agreement and the Registration Rights Agreement each duly executed by Public Company;
- (o) Public Company shall have delivered to Merger Partner a copy of the Amended Operating Agreement, duly executed by Public Company;
- (p) Public Company shall have delivered to Merger Partner an agreement evidencing termination of Public Company as a party to that certain Management Services Agreement by and among Cellectis S.A., Cellectis, Inc. and Public Company, in form and substance reasonably acceptable to Merger Partner, effective as of or prior to the Closing; and
- (q) Public Company shall have taken all steps necessary to effectuate Section 2.1(b)(i)(E), including amending the Public Company Stock Plans as needed to issue the awards contemplated by Section 2.1(b)(i)(E) and filing applicable registration statements with respect thereto.

ARTICLE VII ADDITIONAL AGREEMENTS

7.1 Employee Matters.

- (a) Solely to the extent any of the eligible participants in the Public Company 2021 Executive Severance Plan (the "Executive Severance Plan") set forth on Schedule 7.1 are employed by Public Company immediately prior to the Closing and who (a) resign employment with Public Company, Merger Partner, and their affiliates on the Closing Date for any reason or (b) are terminated by Public Company or Merger Partner other than for Cause, death, or Disability (as such terms are defined in the Executive Severance Plan) on the Closing Date, shall be deemed to undergo a Qualifying Termination during a Change-In-Control Period and shall, effective as of the Closing, be entitled to receive the corresponding Severance Benefits as set forth in the Executive Severance Plan and applicable Participation Agreement. Any capitalized terms used in this Section 7.1 and not otherwise defined in this Agreement shall have the meanings ascribed thereto in the Executive Severance Plan.
- (b) Nothing in this Agreement shall confer upon any employee, officer, director or consultant of Public Company, Merger Partner, Blockers or any of their respective Subsidiaries any right to continue in the employ or service of Public Company, Merger Partner or any Subsidiary thereof, or shall interfere with or restrict in any way the rights of Public Company, Merger Partner or any Subsidiary thereof to discharge or terminate the services of any employee, officer, director or consultant at any time for any reason whatsoever, with or without cause. Nothing in this Agreement shall be deemed to (i) establish, amend, or modify any Public Company Benefit Plan or Merger Partner Benefit Plan or any other benefit or employment plan, program, agreement or arrangement, or (ii) alter or limit the ability of Public Company, Merger Partner or any Subsidiary to amend, modify or terminate any particular Public Company Benefit Plan, Merger Partner Benefit Plan or any other benefit or employment plan, program, agreement or arrangement after the Closing. Without limiting the generality of Section 10.3 and subject to 7.2, nothing in this Agreement, express or implied, is intended to or shall confer upon any Person, including any current or former employee, officer, director or consultant of Public Company, Merger Partner, any Blocker or any of their respective Subsidiaries, any third party beneficiary rights or any other right, benefit or remedy of any nature whatsoever under or by reason of this Agreement.

7.2 D&O Indemnification.

(a) From the Closing through the sixth (6th) anniversary of the Closing Date occurs, Public Company shall indemnify and hold harmless each Person who is now, or has been at any time prior to the date hereof, or who becomes prior to the Closing, a director, manager or officer of Merger Partner, Public Company or any of their respective Subsidiaries (the "Indemnified Persons"), against all claims, losses, liabilities, damages, judgments, fines and reasonable fees, costs and expenses, including attorneys' fees and disbursements, incurred in connection with any claim, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to the fact that the Indemnified Person is or was an officer, director, manager, employee, trustee, fiduciary or agent of Merger Partner, Public Company or any of their respective Subsidiaries, or, while a director, manager or officer of Merger Partner, Public Company or any of their respective Subsidiaries, is or was serving at the request of Merger Partner, Public Company or any of their respective Subsidiaries as a director, manager, officer, employee or agent of another Person, whether asserted or claimed prior to, at or after the Closing, to the extent permitted under the applicable certificate of incorporation and bylaws or other organizational document. Each Indemnified Person will be entitled to advancement of expenses (including attorneys' fees) incurred in the defense of any such claim, action, suit, proceeding or investigation from Public Company following receipt by Public Company from the Indemnified Party of a request therefor; provided that any Person to whom expenses are advanced provides an undertaking, to the extent then required by Delaware Law, to repay such advances if it is ultimately determined that such Person is not entitled to indemnification. From the Closing through the sixth anniversary of the Closing Date, the certificate of incorporation and bylaws of Public Company will contain provisions at least as favorable as the provisions relating to the indemnification, advance of expenses and elimination of liability for monetary damages set forth in the certificate of incorporation and bylaws of Public Company immediately prior to the Closing.

(b) Prior to the Closing, Public Company shall purchase a six-year prepaid "D&O tail policy" (the "Public Company Tail Policy") for the non-cancellable extension of the directors' and officers' liability coverage of Public Company's existing directors' and officers' insurance policies for a claims reporting or discovery period of at least six years from and after the Closing with respect to any claim related to any period of time at or prior to the Merger Partner Merger Effective Time with terms, conditions, retentions and limits of liability to be mutually agreed by Public Company and Merger Partner prior to the Closing (which approval will not be unreasonably withheld, conditioned or delayed), but that are no more favorable than the coverage provided under Public Company's existing policies as of the date of this Agreement with respect to coverage of any actual or alleged error, misstatement, misleading statement, act, omission, neglect, breach of duty or any matter claimed against a director or officer of Public Company by reason of him or her serving in such capacity that existed or occurred at or prior to the Closing (including in connection with this Agreement or the Mergers).

- (c) Prior to the Closing, Merger Partner shall purchase a six-year prepaid "D&O tail policy" (the "Merger Partner Tail Policy") for the non-cancellable extension of the directors' and officers' liability coverage of Merger Partner's existing directors' and officers' insurance policies for a claims reporting or discovery period of at least six years from and after the Effective Time with respect to any claim related to any period of time at or prior to the Effective Time with terms, conditions, retentions and limits of liability that are no less favorable than the coverage provided under Merger Partner's existing policies as of the date of this Agreement with respect to any actual or alleged error, misstatement, misleading statement, act, omission, neglect, breach of duty or any matter claimed against a manager, director or officer of Merger Partner by reason of him or her serving in such capacity that existed or occurred at or prior to the Closing (including in connection with this Agreement or the Merger.
- (d) Public Company shall pay all expenses, including reasonable attorneys' fees, that may be incurred by a Person in successfully enforcing such Person's rights provided in this Section 7.2.
- (e) Public Company and Merger Partner agree that all rights to exculpation, indemnification and advancement of expenses for acts or omissions occurring at or prior to the Closing, whether asserted or claimed prior to, at or after the Closing, now existing in favor of the current or former directors, managers, officers or employees, as the case may be, of Public Company, Merger Partner or any of their respective Subsidiaries as provided in their respective certificates of incorporation or bylaws or other organization documents or in any agreement in existence immediately prior to the Closing shall survive the Mergers and shall continue in full force and effect. The provisions of this Section 7.2 are intended to be in addition to the rights otherwise available to the current officers, managers and directors of Public Company, Merger Partner or any of their respective Subsidiaries by Law, charter, statute, bylaw or agreement, and shall operate for the benefit of, and shall be enforceable by, each of the Indemnified Persons, their heirs and their representatives. The obligations set forth in this Section 7.2 shall not be terminated, amended or otherwise modified in any manner that adversely affects any Indemnified Person, or any Person who is a beneficiary under the policies referred to in this Section 7.2 and their heirs and representatives, without the prior written consent of such affected Indemnified Person or other Person.
- (f) If Public Company or any of its successors or assigns shall (i) consolidate with or merge into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger, or (ii) transfer all or substantially all of its properties and assets to any Person, then, and in each such case, proper provisions shall be made so that the successors and assigns of such Person shall assume all of the obligations of such Person set forth in this Section 7.2.
- (g) Nothing in this Agreement is intended to, shall be construed to or shall release, waive or impair any rights to directors' and officers' insurance claims under any policy that is or has been in existence with respect to Merger Partner, Public Company or any of their respective Subsidiaries for any of their respective directors, managers, officers or other employees, it being understood and agreed that the indemnification provided for in this Section 7.2 is not prior to or in substitution for any such claims under such policies.

7.3 Tax Matters.

- (a) Public Company shall pay all transfer, documentary, sales, use, stamp, registration, value added or other similar Taxes incurred in connection with the Transactions (collectively, the "<u>Transfer Taxes</u>") and file all necessary Tax Returns with respect to all Transfer Taxes, and, if required by applicable Law, Public Company and Merger Partner shall, and shall cause their respective Affiliates to, join in the execution of any such Tax Returns and other document. Notwithstanding any other provision of this Agreement, Public Company and Merger Partner shall (and shall cause their respective Affiliates to) cooperate in good faith to minimize, to the extent permissible under applicable Law, the amount of any such Transfer Taxes.
- (b) For U.S. federal income tax purposes (and for purposes of any applicable state or local income tax that follows U.S. federal income tax treatment), each of the Parties intends that (the "<u>Intended Tax Treatment</u>") (i) the Public Company's acquisition of interests in Merger Partner by operation of the Merger Partner Merger, and Public Company's issuance of Public Company Class A Common Stock and Public Company Class B Common Stock to Participating Members, Electing Members, holders of Merger Partner Warrants, and holders of Merger Partner PIUs at the Merger Partner Merger Effective Time, shall be treated as a transaction governed by Section 351 of the Code, (ii) the Blocker Mergers will each qualify as a "reorganization" within the meaning of Section 368(a) of the Code, and (iii) and the other changes to the capital structure of Merger Partner at Closing contemplated by Sections 2.1(b)(i)(B) and 2.1(b)(ii) shall be treated as a non-taxable amendment to the partnership agreement under Section 761(c) of the Code, or as otherwise nontaxable under Revenue Ruling 84-52, 1984-1 C.B. 157 (or otherwise under Section 721 of the Code). This Agreement is hereby adopted as and shall constitute a "plan of reorganization" within the meaning of Treasury Regulations Section 1.368-2(g) with respect to each of the Blocker Mergers.
- (c) Each of the Parties intends for the Transactions to qualify for the Intended Tax Treatment and will not, without the Unitholder Representative's or Public Company's consent (not to be unreasonably withheld, conditioned, or delayed), take any inconsistent position on any Tax Return or during the course of any audit, litigation or other proceeding with respect to Taxes, except as otherwise required by a determination within the meaning of Section 1313(a) of the Code. Each of the Parties agrees to promptly notify all other Parties of any challenge to the Intended Tax Treatment by any Governmental Authority.
- (d) No Party shall knowingly take or cause to be taken any action, or fail to take or cause to be taken any action, which action or failure to act would reasonably be expected to prevent the Transactions from so qualifying for the Intended Tax Treatment.
- (e) Public Company and Merger Partner shall use reasonable best efforts to execute and deliver customary representations reasonably requested by counsel to Public Company or counsel to Merger Partner, as applicable, for purposes of rendering opinions regarding the Intended Tax Treatment and other tax matters in connection with the Transactions, at such time or times as may be reasonably requested by such counsel, including in connection with the Closing and any filing of any Form S-4. In the event that the SEC (or its staff) requires a tax opinion regarding the Intended Tax Treatment, Merger Partner shall use its reasonable best efforts to cause Akin Gump Strauss Hauer & Feld LLP to deliver such opinion (subject to customary assumptions and limitations) to Merger Partner. For the avoidance of doubt, the delivery of tax opinions by counsel to Public Company or counsel to Merger Partner shall not be a condition to Closing under this Agreement.

- (f) The Unitholder Representative shall prepare, or cause to be prepared, at the expense of Merger Partner, all income Tax Returns with respect to Pass-Through Income Taxes of Merger Partner (and its Subsidiaries) for any Pre-Closing Tax Period (other than the portion of any Straddle Period ending on the Closing Date) that are due after the Closing Date (taking into account applicable extensions) (each, a "Pass-Through Return"). Each Pass-Through Return, as prepared by the Unitholder Representative, shall be submitted to the Public Company for its review and approval no later than 45 days prior to its due date (taking into account applicable extensions). Public Company shall propose any comments on such Pass-Through Returns no later than 15 days prior to their due date (taking into account applicable extensions), and the Unitholder Representative and the Unitholder Representative shall consider in good faith any such comments. Public Company shall cause all Pass-Through Returns as prepared by the Unitholder Representative, to be timely filed and will promptly provide a copy of all filed Pass-Through Returns to the Unitholder Representative.
- (g) In the event of any proposed audit, adjustment, assessment, examination, claim or other controversy or proceeding relating to Pass-Through Income Taxes or any Pass-Through Return (a "<u>Tax Contest</u>") with respect to a Straddle Period, Public Company will, within 15 days of it (or any of its Subsidiaries) being notified of such Tax Contest, notify the Unitholder Representative of such Tax Contest in writing. All Tax Contests for any taxable periods initiated or continuing after the Closing Date shall be controlled in accordance with the terms of the Merger Partner Operating Agreement.
- (h) After the Closing, Public Company (and its Subsidiaries) will not, without the consent of the Unitholder Representative (which consent will not be unreasonably withheld, conditioned or delayed) or as required by applicable Law, (i) amend or otherwise modify any Pass-Through Return, (ii) extend or waive, or cause to be extended or waived, any statute of limitations or other period for the assessment of any Pass-Through Income Taxes for Pre-Closing Tax Periods (other than the portion of any Straddle Period ending on the Closing Date) or audit of any Pass-Through Return, (iii) initiate with a taxing authority discussions or examinations or voluntary disclosures with respect to Pass-Through Taxes for Pre-Closing Tax Period or (iv) make or change any income election or accounting method or practice with respect to Pass-Through Income Taxes for Pre-Closing Tax Periods (other than the portion of any Straddle Period ending on the Closing Date) or Pass-Through Returns.
- (i) Public Company, Merger Partner and the Unitholder Representative shall reasonably cooperate (and cause its Affiliates to reasonably cooperate), as and to the extent reasonably requested by each other Party and the Unitholder Representative, in connection with the preparation and filing of Tax Returns or any Tax Contest. Such cooperation shall include the provision of available records and information that are reasonably relevant to any such matter and making employees available on a mutually convenient basis to provide additional information and explanation with respect thereto.

- (j) Public Company and the sole member of each Blocker shall reasonably cooperate (and cause their respective affiliates to reasonably cooperate) with respect to the administration of tax matters relating to such Blocker (including the preparation of Tax Returns and the conduct of any tax audits, adjustments, assessments, examinations, claims or other controversies or proceedings).
- (k) Merger Partner has in effect, or shall make effective starting no later than the taxable year of the Closing, a valid election under Section 754 of the Code.
- 7.4 <u>Transaction Litigation</u>. In the event that any litigation related to this Agreement or the Transactions is brought, or, to the Knowledge of Public Company, threatened, against Public Company, any members of Public Company's Board or the Public Company Support Stockholders under the Public Company Support Agreements (such litigation, "<u>Public Company Transaction Litigation</u>"), Public Company shall promptly notify Merger Partner of such Public Company Transaction Litigation and shall keep Merger Partner reasonably informed with respect to the status thereof. Public Company shall give Merger Partner a reasonable opportunity to participate in the defense or settlement, at Merger Partner's sole expense and subject to a customary joint defense agreement, of any Public Company Transaction Litigation and shall consider in good faith Merger Partner's advice with respect to such Public Company Transaction Litigation; <u>provided</u> that Public Company shall retain the right to control the defense of such Public Company Transaction Litigation in its sole discretion and the disclosure of information to Merger Partner in connection therewith shall be subject to the provisions of Section 5.11 (without giving effect to the words "upon reasonable notice and during normal business hours"); <u>provided</u>, that Public Company shall not settle or moot, or agree to settle or moot, any Public Company Transaction Litigation without the prior written consent of Merger Partner.
- 7.5 <u>Asset Transfers</u>. Immediately prior to the consummation of the Mergers, Public Company shall consummate the Asset Transfers in accordance with the terms of one or more Transfer Agreements in the form(s) to be agreed between Public Company and Merger Partner as promptly following the date hereof as practicable (the "<u>Asset Transfer Agreements</u>"). In exchange for the Asset Transfers, Merger Partner will agree that the outstanding Interim Funding subject to repayment thereafter shall be reduced to zero (\$0) and Merger Partner shall issue to Public Company the number of Merger Partner Common Units set forth in the Allocation Schedule.

ARTICLE VIII TERMINATION AND AMENDMENT

- 8.1 <u>Termination</u>. This Agreement may be terminated at any time prior to the Closing, whether before or, subject to the terms hereof, after receipt of the approval of the stockholders of Public Company or the members of Merger Partner and/or the Blockers:
 - (a) by mutual written consent of Public Company and Merger Partner; or

- (b) by either Public Company or Merger Partner if the Merger shall not have been consummated by the Outside Date; provided, that if on the Outside Date, each of the conditions to closing set forth in Sections 6.1 and 6.2 shall not have been satisfied or waived by the parties entitled to the benefit thereof (other than conditions that, by their nature, are to be satisfied or waived at the Closing), then at the written election of Merger Partner (which written election shall include acknowledgment and agreement by Merger Partner that the Interim Funding available to Public Company shall be increased pursuant to Section 5.17), the Outside Date may be extended to June 30, 2023 (and in the case of such extension, any reference to the Outside Date in any other provision of this Agreement shall be a reference to the Outside Date, as extended); provided, however, that the right to terminate this Agreement under this Section this Section 8.1(b) shall not be available to a party if the failure of the Mergers to have been consummated on or before the Outside Date was primarily due to such party's action or failure to act, where such action or failure to act constitutes a breach of this Agreement (with any action or failure to act by Merger Partner); or
- (c) by either Public Company or Merger Partner if a Governmental Entity of competent jurisdiction shall have issued a nonappealable final order, decree or ruling or taken any other nonappealable final action, in each case having the effect of permanently restraining, enjoining or otherwise prohibiting any of the Mergers; provided, however, that the right to terminate this Agreement under this Section 8.1(c) shall not be available to a party if the issuance of such order, decree, ruling or the taking of such action was primarily due to such party's action or failure to act, where such action or failure to act constitutes a breach of this Agreement (with any action or failure to act by a Blocker being considered an action or failure to act by Merger Partner); or
- (d) by Public Company, if there has been a breach of or failure to perform any representation, warranty, covenant or agreement on the part of Merger Partner or a Blocker set forth in this Agreement, which breach or failure to perform (i) would cause the conditions set forth in Section 6.1(a) or 6.1(b) not to be satisfied and (ii) shall not have been cured, if capable of being cured, or waived within 30 days following receipt by Merger Partner of written notice of such breach or failure to perform from Public Company; provided, however, that the right to terminate this Agreement under this Section 8.1(d) shall not be available to Public Company if Public Company or Merger Subsidiary then in material breach of any representation, warranty or covenant set forth in this Agreement; or
- (e) by Merger Partner, if there has been a breach of or failure to perform any representation, warranty, covenant or agreement on the part of Public Company or Merger Subsidiary set forth in this Agreement, which breach or failure to perform (i) would cause the conditions set forth in Section 6.2(a) or 6.2(b) not to be satisfied and (ii) shall not have been cured, if capable of being cured, or waived within 30 days following receipt by Public Company of written notice of such breach or failure to perform from Merger Partner; provided, however, that the right to terminate this Agreement under this Section 8.1(e) shall not be available to Merger Partner if Merger Partner is then in material breach of any representation, warranty or covenant set forth in this Agreement; or
- (f) by Merger Partner, at any time prior to receipt of the Public Company Stockholder Approval, if: (i) the Public Company Board shall have effected a Public Company Board Adverse Recommendation Change; or (ii) Public Company shall have committed a material breach of Public Company's obligations under Section 5.4 of this Agreement; or

- (g) by Public Company, at any time prior to receipt of the Public Company Stockholder Approval, if each of the following occur: (i) Public Company shall have received a Superior Offer; (ii) Public Company shall have complied with its obligations under Section 5.4 with respect to such Superior Offer, including with respect to making a Public Company Board Recommendation Change with respect to such Superior Offer; (iii) the Public Company Board approves such Superior Offer, and Public Company substantially concurrently with the termination of this Agreement enters into, a definitive agreement with respect to such Superior Offer; and (iv) prior to or substantially concurrently with such termination, Public Company pays to Merger Partner the amounts contemplated by Sections 8.3(b) and 8.3(d); or
- (h) by either Public Company or Merger Partner if (i) Public Company Stockholders' Meeting (including any adjournments and postponements thereof) shall have been held and completed and Public Company's stockholders shall have taken a final vote on Public Company Stockholder Matters and the Other Public Company Stockholder Matters and (ii) any of the Public Company Stockholder Matters or the Other Public Company Stockholder Matters shall not have been approved at the Public Company Stockholders' Meeting (or at any adjournment or postponement thereof) by the Required Public Company Stockholder Vote; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement pursuant to this Section 8.1(h) shall not be available to Public Company if Public Company's actions or failure to act has been a principal cause of the failure of Public Company to obtain the Required Public Company Stockholder Vote and such action or failure to act constitutes a breach by Public Company of this Agreement; and
- (i) by Public Company, if (x) the Merger Partner Member Approval and the Blocker Approvals shall not have been obtained prior to 5:00 p.m., New York time, on the date that is ten (10) Business Days following the date that the SEC declares the Registration Statement effective or (y) Merger Partner shall have committed a material breach of Merger Partner's obligations under Section 5.5.

The party desiring to terminate this Agreement pursuant to this Section 8.1 (other than pursuant to Section 8.1(a)) shall give notice of such termination to the other party specifying the provisions hereof pursuant to which such termination is made and the basis therefore described in reasonable detail.

8.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement as provided in Section 8.1, this Agreement shall be of no further force or effect; <u>provided, however</u>, that (a) this Section 8.2, Section 8.3, Article X and the definitions of the defined terms in such Sections (including the definitions of such defined terms in Article IX) shall survive the termination of this Agreement and shall remain in full force and effect, and (b) the termination of this Agreement and the provisions of Section 8.3 shall not relieve any party of any liability for Fraud committed prior to such termination or for any Willful Breach prior to such termination of any representation, warranty, covenant, obligation or other provision contained in this Agreement.

8.3 Fees and Expenses.

(a) Except as otherwise expressly provided herein (including in this Section 8.3), whether or not the Mergers are consummated, each party will pay all fees and expenses (including legal and accounting fees and expenses) incurred by it in connection with the Transactions.

(b) If

- (i) (A) this Agreement is terminated pursuant to Section 8.1(b), Section 8.1(e) or Section 8.1(h) and (B) an Acquisition Proposal with respect to Public Company shall have been publicly announced or disclosed to Public Company or Public Company Board after the date of this Agreement but prior to the termination of this Agreement (which shall not have been withdrawn), and (C) within twelve (12) months after the date of such termination, Public Company enters into a definitive agreement with respect to or consummates an Acquisition Proposal; or
- (ii) this Agreement is terminated by Merger Partner pursuant to Section 8.1(f) or by Public Company pursuant to Section 8.1(g); then Public Company shall pay to Merger Partner an amount equal to \$1,000,000 (a "Public Company Termination Fee") within three (3) Business Days of the termination of this Agreement or, in the cause of clause (i) above, the date of the applicable triggering event.
- (c) If this Agreement is terminated by Public Company pursuant to Section 8.1(i) or by Merger Partner pursuant to Section 8.1(e) at a time when Public Company had the right to terminate this Agreement pursuant to Section 8.1(i), then Merger Partner shall pay to Public Company an amount equal to \$1,000,000 (a "Merger Partner Termination Fee") and reimburse Public Company for Public Company's reasonable and documented out-of-pocket costs and expenses incurred in connection with this Agreement (such reimbursement not to exceed \$500,000) within three (3) Business Days of the termination of this Agreement.
- (d) Any amounts due under this Section 8.3 shall be paid by wire transfer of same day funds. If a party fails to pay when due any amount payable by it under this Section 8.3, then such party shall (i) reimburse the other party for reasonable costs and expenses (including reasonable fees and disbursements of counsel) incurred by it in connection with the collection of such overdue amount and the enforcement by such party of its rights under this Section 8.3 and (ii) pay to the other party interest on such overdue amount (for the period commencing as of the date such overdue amount was originally required to be paid and ending on the date such overdue amount is actually paid in full) at a rate per annum equal to the "prime rate" (as published in The Wall Street Journal or any successor thereto) in effect on the date such overdue amount was originally required to be paid.
- (e) The parties agree that, subject to Section 8.2, (i) payment of the Public Company Termination Fee set forth in this Section 8.3 shall, in the circumstances in which it is owed in accordance with the terms of this Agreement, constitute the sole and exclusive remedy of Merger Partner and Blockers following the termination of this Agreement, it being understood that in no event shall Public Company be required to pay a Public Company Termination Fee on

more than one occasion and (ii) following payment of the Public Company Termination Fee (x) Public Company shall have no further liability to Merger Partner or the Blockers in connection with or arising out of this Agreement or the termination thereof, any breach of this Agreement by Public Company giving rise to such termination, or the failure of the Transactions to be consummated, (y) neither Merger Partner or a Blocker nor any of their respective Affiliates shall be entitled to bring or maintain any other claim, action or proceeding against Public Company or Merger Subsidiary or seek to obtain any recovery, judgment or damages of any kind against such parties (or any partner, member, stockholder, director, officer, employee, Subsidiary, Affiliate, agent or other Representative of such parties) in connection with or arising out of this Agreement or the termination thereof, any breach by any such parties giving rise to such termination or the failure of the Transactions to be consummated and (z) Merger Partner, Blockers and their respective Affiliates shall be precluded from any other remedy against Public Company, Merger Subsidiary and their respective Affiliates, at law or in equity or otherwise, in connection with or arising out of this Agreement or the termination thereof, any breach by such party giving rise to such termination or the failure of the Transactions to be consummated.

- (f) Each of the parties acknowledges that (i) the agreements contained in this Section 8.3 are an integral part of the Transactions, (ii) without these agreements, the parties would not enter into this Agreement and (iii) any amount payable pursuant to this Section 8.3 is not a penalty, but rather is liquidated damages in a reasonable amount that will compensate the applicable party in the circumstances in which such amount is payable.
- 8.4 <u>Amendment</u>. Prior to the Closing, this Agreement may be amended by Public Company and Merger Partner at any time before or after receipt of the Public Company Stockholder Approval or the Merger Partner Member Approval, but, after receipt of the Public Company Stockholder Approval or the Merger Partner Member Approval, no amendment shall be made which by Law requires further approval by such stockholders or members, as applicable, without such further approval; <u>provided</u>, <u>however</u>, that any such amendment that adversely effects a Blocker shall also require the approval of such Blocker. This Agreement may not be amended except by an instrument in writing signed by Public Company, Merger Partner and the applicable Blockers (if any).
- 8.5 Extension; Waiver. (a) At any time prior to the Closing, Public Company and Merger Partner may, to the extent legally allowed, (i) extend the time for the performance of any of the obligations or other acts of the other parties hereto, (ii) waive any inaccuracies in the representations and warranties contained herein or in any document delivered pursuant hereto and (iii) waive compliance with any of the agreements or conditions contained herein; (b) any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in a written instrument signed on behalf of such party; (c) such extension or waiver shall not be deemed to apply to any time for performance, inaccuracy in any representation or warranty, or noncompliance with any agreement or condition, as the case may be, other than that which is specified in the extension or waiver; and (d) the failure of any party to this Agreement to assert any of its rights under this Agreement or otherwise shall not constitute a waiver of such rights.

ARTICLE IX DEFINITIONS

For purposes of this Agreement, each of the following terms has the meaning set forth below.

"Acquisition Inquiry" means, with respect to a Party, an inquiry, indication of interest or request for information (other than an inquiry, indication of interest or request for information made or submitted by Merger Partner, on the one hand, or Public Company, on the other hand, to the other Party) that could reasonably be expected to lead to an Acquisition Proposal.

"Acquisition Proposal" means, with respect to a Party, any offer or proposal, whether written or oral (other than an offer or proposal made or submitted by or on behalf of Merger Partner or any of its Subsidiaries, on the one hand, or by or on behalf of Public Company or any of its Subsidiaries, on the other hand, to the other Party) contemplating or otherwise relating to or that could reasonably be interpreted to lead to any Acquisition Transaction with such Party.

"Acquisition Transaction" means any transaction or series of related transactions involving:

- (a) any merger, consolidation, amalgamation, share exchange, business combination, issuance of securities, acquisition of securities, reorganization, recapitalization, tender offer, exchange offer or other similar transaction: (i) in which a Party is a constituent entity; (ii) in which a Person or "group" (as defined in the Exchange Act and the rules promulgated thereunder) of Persons directly or indirectly acquires beneficial or record ownership of securities representing (A) with respect to Public Company more than 20% of the outstanding securities of any class of voting securities of a Party or any of its Subsidiaries; or (B) with respect to Merger Partner more than 35% of the outstanding securities of any class of voting securities of a Party or any of its Subsidiaries; or (iii) in which (Y) Public Company or any of its Subsidiaries issues securities representing more than 20% of the outstanding securities of any class of voting securities of Public Company or any of its Subsidiaries or (Z) Merger Partner or any of its Subsidiaries; or securities representing more than 35% of the outstanding securities of voting securities of Merger Partner or any of its Subsidiaries; or
- (b) any sale, lease, exchange, transfer, license, acquisition or disposition of any business or businesses or assets that constitute or account for 20% or more of the consolidated book value or the fair market value of the assets of a Party and its Subsidiaries, taken as a whole.
 - "Additional Merger Partner Unit" has the meaning set forth in Section 2.3.
- "Affiliate" means, with respect to a Person, any other Person who is an "affiliate" of that Person within the meaning of Rule 405 promulgated under the Securities Act.

"Agreement" has the meaning set forth in the first paragraph of this Agreement.

"Allocation Schedule" means the schedule attached hereto as Exhibit G (as such schedule may be updated, corrected, amended or modified in accordance with Section 2.3 from time to time), setting forth for each holder of a Merger Partner Unit and of a Merger Partner Warrant and of an equity interest in each Blocker: (i) the name and address for such Person; (ii) the number and class of interests of Merger Partner and each Blocker, as applicable, held by such Person; and (iii) the portion of the Merger Consideration attributable to such Person.

"Amended By-Laws" has the meaning set forth in Section 1.3(a).

"Amended Certificate" has the meaning set forth in Section 1.3(a).

"Amended Operating Agreement" has the meaning set forth in Section 1.3(c).

"Anti-Bribery Laws" has the meaning set forth in Section 3.17.

"Asset Transfer Agreements" has the meaning set forth in Section 7.5.

"Asset Transfers" has the meaning set forth in the Recitals.

"Blocker 1" has the meaning set forth in the Recitals.

"Blocker 2" has the meaning set forth in the Recitals.

"Blocker 3" has the meaning set forth in the Recitals.

"Blocker 4" has the meaning set forth in the Recitals.

"Blocker 5" has the meaning set forth in the Recitals.

"Blocker 6" has the meaning set forth in the Recitals.

"Blocker 7" has the meaning set forth in the Recitals.

"Blocker 8" has the meaning set forth in the Recitals.

"Blocker 9" has the meaning set forth in the Recitals.

"Blocker 10" has the meaning set forth in the Recitals.

"Blocker Approvals" means a written consent of the sole member of each of Blocker 2 through Blocker 10 wherein the sole member of each such Blocker adopted this Agreement and approved the Transactions, including the applicable Blocker Merger, in each case, consistent with the applicable provisions of such Blocker's Organizational Documents and Delaware Law.

"Blocker Certificate of Merger" has the meaning set forth in Section 1.1(a)(i).

"Blocker Merger Effective Time" has the meaning set forth in Section 1.1(a)(iii).

"Business Day" means any day other than (a) a Saturday or Sunday or (b) a day on which banking institutions located in Wilmington, Delaware or New York, New York are permitted or required by Law, executive order or governmental decree to remain closed.

- "CERCLA" means the federal Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended.
- "Certifications" has the meaning set forth in Section 4.6(a).
- "Closing" means the closing of the Transactions.
- "Closing Date" means (a) the second (2nd) Business Day after the satisfaction or waiver of the conditions set forth in Article VI (other than the delivery of items to be delivered at the Closing and other than satisfaction of those conditions that by their nature are to be satisfied at the Closing, it being understood that the occurrence of the Closing shall remain subject to the delivery of such items and the satisfaction or waiver of such conditions at the Closing) or (b) such other date as may be mutually agreed to in writing by Merger Partner and Public Company.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Confidentiality Agreement" means the Confidentiality Agreement dated September 7, 2022 between Merger Partner and Public Company.
 - "Copyright" has the meaning set forth in the definition of "Intellectual Property" herein.
 - "COVID-19" means SARS-CoV-2 or COVID-19, and any evolutions thereof or related or associated epidemics, pandemics or disease outbreaks.
- "COVID-19 Measures" means any quarantine, "shelter in place," "stay at home," workforce reduction, social distancing, shutdown, closure, sequester or any other applicable Law, order or recommendations of a Governmental Entity, or any applicable directive or guidance from any applicable industry group widely followed by the party's industry peer companies in response to COVID-19.
 - "Delaware Law" means, as the context requires, the DGCL and/or the DLLCA.
 - "Design" has the meaning set forth in the definition of "Intellectual Property" herein.
 - "Determination Notice" has the meaning set forth in Section 5.9(d)(i).
 - "DGCL" means the Delaware General Corporation Law, as amended.
- "<u>Disclosure Statement</u>" means a written information statement, which shall include a copy of the Proxy Statement/Prospectus and shall contain the information prescribed by Section 5.10(b).
 - "DLLCA" means the Delaware Limited Liability Company Act, as amended.
 - "DOL" has the meaning set forth in Section 3.15(a).
 - "Effect" has the meaning set forth in the definition of "Merger Partner Material Adverse Effect" herein.

- "Eighth Blocker Merger" has the meaning set forth in the Recitals.
- "Electing Member" has the meaning set forth in Section 2.5(b).
- "Election Deadline" has the meaning set forth in Section 2.5(c).
- "Election Form" has the meaning set forth in Section 2.5(a).
- "Election Period" has the meaning set forth in Section 2.5(c).

"Employee Benefit Plans" means all (a) "employee benefit plans," as defined in Section 3(3) of ERISA, together with plans or arrangements that would be so defined if they were not (i) otherwise exempt from ERISA by Section 3(3) of ERISA or another Section of ERISA, (ii) maintained outside the United States or (iii) individually negotiated or applicable only to one individual and (b) any other written or oral benefit arrangement or obligation to provide benefits as compensation for services rendered, including employment or consulting agreements (except for agreements that provide termination at no cost to Merger Partner), severance agreements, arrangements, plans or pay policies, stay or retention bonuses or compensation, incentive (including equity or equity-linked) plans, programs or arrangements, patent award programs, relocation benefits, loan, sick leave, vacation pay, plant closing benefits, salary continuation or insurance for disability, individual consulting, or other compensation arrangements, retirement, supplemental retirement, pension, deferred compensation, profit sharing, bonus, stock option or purchase plans or programs, hospitalization, medical insurance, life insurance, dental insurance, tuition reimbursement or scholarship programs, any plans subject to Section 125 of the Code and any plans providing benefits or payments in the event of a change of control, change in ownership or effective control, or sale of a substantial portion (including all or substantially all) of the assets of any business or portion thereof.

"Employment Agreement" has the meaning set forth in the Recitals.

"Environmental Law" means any Law relating to the environment, occupational health and safety, or exposure of Persons or property to Materials of Environmental Concern, including any statute, regulation, administrative decision or order pertaining to: (a) the presence of or the treatment, storage, disposal, generation, transportation, handling, distribution, manufacture, processing, use, import, export, labeling, recycling, registration, investigation or remediation of Materials of Environmental Concern or documentation related to the foregoing; (b) air, water and noise pollution; (c) groundwater and soil contamination; (d) the release, threatened release, or accidental release into the environment, the workplace or other areas of Materials of Environmental Concern, including emissions, discharges, injections, spills, escapes or dumping of Materials of Environmental Concern; (e) transfer of interests in or control of real property which may be contaminated; (f) community or worker right-to-know disclosures with respect to Materials of Environmental Concern; (g) the protection of wild life, marine life and wetlands, and endangered and threatened species; (h) storage tanks, vessels, containers, abandoned or discarded barrels and other closed receptacles; and (i) health and safety of employees and other Persons. As used above, the term "release" shall have the meaning set forth in CERCLA.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.

"ERISA Affiliate" means any entity that is, or at any applicable time was, a member of (a) a controlled group of corporations (as defined in Section 414(b) of the Code), (b) a group of trades or businesses under common control (as defined in Section 414(c) of the Code), or (c) an affiliated service group (as defined under Section 414(m) of the Code or the regulations under Section 414(o) of the Code).

"Exchange and Paying Agent" means Broadridge Corporate Issuer Solutions, Inc., in its capacity as exchange and paying agent, or another bank or trust company reasonably and mutually acceptable to Public Company and Merger Partner.

"Exchange and Paying Agent Agreement" means an agreement, in form and substance reasonably acceptable to Merger Partner, to be entered into at or prior to the Effective Time by the Exchange and Paying Agent and Public Company, governing the disbursement of the Payment Fund.

"Exchange Act" means the Securities Exchange Act of 1934, as amended.

"Exchange Agreement" has the meaning set forth in the Recitals.

"Exchanges" has the meaning set forth in the Recitals.

"Excluded Units" has the meaning set forth in Section 2.5(b).

"Executive Severance Plan" has the meaning set forth in Section 7.1(a).

"Fifth Blocker Merger" has the meaning set forth in the Recitals.

"First Blocker Merger" has the meaning set forth in the Recitals.

"Fourth Blocker Merger" has the meaning set forth in the Recitals.

"Fraud" with respect to a party means intentional common law fraud under Delaware law in the making of the representations and warranties in this Agreement.

"Fundamental Representations" means, with respect to Merger Partner, the representations in Sections 3.1, 3.2, 3.3(c), 3.4 (other than 3.4(c)(iii) and 3.4(c)(iv)), 3.21, 3.22, 3.23, 3.24 and 3.25 (other than 3.25(b)(iii) and 3.25(b)(iv)) and, with respect to Public Company, the representations in Sections 4.1, 4.2, 4.4 (other than 4.4(c)(iii) and 4.4(c)(iv)), 4.5, 4.22 and 4.25).

"GAAP" means United States generally accepted accounting principles.

"Governmental Entity" means any federal, state, local or foreign government or any court, arbitrational tribunal, administrative agency or commission or government authority acting under the authority of the federal or any state, local or foreign government.

"HSR" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder.

"Indebtedness" with respect to any Person means (a) any indebtedness or other obligation for borrowed money or under bonds, notes, debentures or similar instruments, letters of credit, acceptance credit or similar facilities, as well as any cash advances; (b) any obligation incurred for all or any part of the purchase price of property or other assets (including earnout, milestone, royalty and similar obligations) or for the cost of property or other assets constructed or of improvements thereto, other than accounts payable included in current liabilities and incurred in respect of property purchased in the Ordinary Course of Business; (c) letters of credit issued for the account of such Person to the extent drawn; (d) lease obligations required to be capitalized by GAAP; (e) all guarantees and similar obligations of such Person with respect to obligations of another Person of the nature described in clauses (a), (b), (c) or (d) above; (f) all bankers acceptances and overdrafts; (g) declared but unpaid dividends or distributions; and (h) all interest, prepayment premiums and penalties, guarantees, and any other fees, expenses, indemnities and other amounts payable as a result of the prepayment or discharge of any indebtedness listed in clause (a) above.

"Intellectual Property" means, in any and all jurisdictions throughout the world, all (a) patents, patent applications, inventors' certificates, utility models, statutory invention registrations, and other indicia of ownership of an invention, discovery or improvement issued by a Governmental Authority, including reissues, divisionals, continuations, continuations-in-part, extensions, reexaminations and other pre-grant and post-grant forms of the foregoing (collectively, "Patents"), (b) trademarks, service marks, trade dress, slogans, logos, symbols, trade names, brand names and other identifiers of source or goodwill recognized by any Governmental Authority, including registrations and applications for registration thereof and including the goodwill symbolized thereby or associated therewith (collectively, "Trademarks"), and Internet domain names and associated uniform resource locators, (c) copyrights, whether in published and unpublished works of authorship, registrations, applications, renewals and extensions therefor, mask works, and any and all similar rights recognized in a work of authorship by a Governmental Authority (collectively, "Copyrights"), (d) any rights in any inventions, discoveries, improvements and all other confidential or proprietary Information (including know-how, data, formulas, processes and procedures, research records, records of inventions, test information, and market surveys), and all rights to limit the use or disclosure thereof (collectively, "Trade Secrets"), (e) registered and unregistered design rights (collectively, "Designs"), (f) rights of privacy and publicity, and (g) any and all other intellectual or industrial property rights recognized by any Governmental Authority under the Laws of any country throughout the world.

"Intended Tax Treatment" has the meaning set forth in Section 7.3(b).

"Interim Funding" has the meaning set forth in Section 5.17.

"Knowledge" and words of similar effect means (a) with respect to Public Company, the actual knowledge of each of the individuals identified in Section 9.01 of the Public Company Disclosure Schedule, in each case after due and reasonable inquiry of their respective direct reports and (b) with respect to Merger Partner and the Blockers, the actual knowledge of each of the individuals identified in Section 9.01 of the Merger Partner Disclosure Schedule, in each case after due and reasonable inquiry of their respective direct reports.

"<u>Law</u>" means any United States federal, state or local or foreign law, common law, statute, standard, ordinance, code, rule, regulation, resolution or promulgation, or any decree, order, injunction, rule, judgment, consent of or by any Governmental Entity, or any Permit or similar right granted under any of the foregoing, or any similar provision having the force or effect of law.

"<u>Lease</u>" means any lease, sublease, license, or occupancy agreement pursuant to which Merger Partner (or its Subsidiaries) or Public Company (or its Subsidiaries), as applicable, leases or subleases from or to another party any real property, or otherwise occupies any real property.

"<u>Legal Proceeding</u>" means any action, suit, proceeding (including administrative proceeding), claim, complaint, hearing, information request, notice of violation, arbitration, inquiry or investigation of or before any Governmental Entity or before any arbitrator.

"Letter of Transmittal" means a letter of transmittal in the form to be mutually agreed between Public Company and Merger Partner promptly following the execution and delivery of this Agreement.

"Lien" means any mortgage, pledge, security interest, encumbrance, charge or other lien (whether arising by contract or by operation of Law.

"Mailing Date" has the meaning set forth in Section 2.5(a).

"Management Services Agreement" means that certain Management Services Agreement, dated January 1, 2016, by and among Cellectis SA, a French *Société Anonyme*, Cellectis, Inc., a Delaware corporation, and Public Company, as amended by that certain First Amendment to the Management Services Agreement, dated July 25, 2017, and that certain Second Amendment to Management Services Agreement, dated January 29, 2020.

"Materials of Environmental Concern" means any: pollutants, contaminants or hazardous substances (as such terms are defined under CERCLA), pesticides (as such term is defined under the Federal Insecticide, Fungicide and Rodenticide Act), solid wastes and hazardous wastes (as such terms are defined under the Resource Conservation and Recovery Act), chemicals, other hazardous, radioactive or toxic materials, oil, petroleum and petroleum products (and fractions thereof), or any other material (or article containing such material) listed or subject to regulation under any Law due to its potential, directly or indirectly, to harm the environment or the health of humans or other living beings.

"Measurement Date" has the meaning set forth in 4.2(a).

"Mergers" has the meaning set forth in the Recitals.

"Merger Consideration" means the shares of Public Company Class A Common Stock and Public Company Class B Common Stock and Merger Partner Common Units, as applicable, calculated in accordance with Section 2.1.

"Merger Partner" has the meaning set forth in the first paragraph of this Agreement.

"Merger Partner Benefit Plan" means any Employee Benefit Plan in respect of any employees, independent contractors, directors, officers or members of Merger Partner or any of its Subsidiaries that are sponsored or maintained by Merger Partner or any of its Subsidiaries or with respect to which Merger Partner or any of its Subsidiaries is required to make payments, transfers or contributions or has or may have any actual, potential or contingent liability.

"Merger Partner Board" means the board of managers of Merger Partner.

"Merger Partner Certificate of Merger" has the meaning set forth in Section 1.1(b)(i).

"Merger Partner Change in Circumstance" means a change in circumstances (other than an Acquisition Proposal) that affects the business, assets or operations of Merger Partner that occurs or arises after the date of this Agreement that was neither known to Merger Partner or Merger Partner Board nor reasonably foreseeable on, or prior to, the date of this Agreement.

"Merger Partner Common Unit" means the units of Merger Partner designated as "Common Units" with such rights and restrictions as to be set forth in the Amended Operating Agreement.

"Merger Partner Contract" has the meaning set forth in Section 3.11(a).

"Merger Partner Disclosure Schedule" means Merger Partner Disclosure Schedule provided by Merger Partner to Public Company on the date hereof.

"Merger Partner Employee" means any employee (whether current or former) of Merger Partner or any Subsidiary of Merger Partner.

"Merger Partner Financial Statements" means:

- (a) the consolidated audited balance sheets, statements of operations and comprehensive loss, statements of convertible preferred stock and stockholders' deficit and statements of cash flows of Merger Partner as of the end of and for each of the fiscal years ended December 31, 2020 and 2021, as certified without qualification by BDO USA, LLC, Merger Partner's independent public accountants; and
- (b) the consolidated unaudited balance sheets of Merger Partner for any interim periods after December 31, 2021, including at September 30, 2022, and the related consolidated unaudited statements of operations and comprehensive loss, statements of convertible preferred stock and stockholders' deficit and statements of cash flows for each of the interim period then ended.

"Merger Partner Intellectual Property" has the meaning set forth in Section 3.10(b).

"Merger Partner Insurance Policies" has the meaning set forth in Section 3.19(a).

"Merger Partner IT Systems" means any computer systems, networks, hardware, software and other information technology equipment used in connection with the business of Merger Partner and its Subsidiaries.

"Merger Partner Material Adverse Effect" means any change, event, effect, circumstance or development (collectively, "Effects") that, considered together with all other Effects, has or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), assets, liabilities or results of operations of Merger Partner or its Subsidiaries, taken as a whole; provided, however, that Effects resulting from the following shall not be taken into account in determining whether there has been a Merger Partner Material Adverse Effect: (a) general business or economic conditions generally affecting the industry in which Merger Partner and its Subsidiaries operate, (b) acts of war, the outbreak or escalation of armed hostilities, acts of terrorism (including cyberterrorism), earthquakes, wildfires, hurricanes or other natural disasters, health emergencies, including pandemics (including COVID-19) and related or associated epidemics, disease outbreaks and any governmental or industry responses thereto (including quarantine restrictions), (c) changes in financial, banking, securities markets, or general economic, regulatory, legislative or political conditions (including changes in interest or exchange rates), (d) any change in, or any compliance with or action taken for the purpose of complying with, any Law or GAAP (or interpretations of any Law or GAAP), (e) resulting from the announcement of this Agreement or the pendency of the Transactions, (g) the failure of Merger Partner to meet internal expectations or projections or the results of operations of Merger Partner in and of itself (it being understood, however, that any Effect causing or contributing to such failure may be taken into account in determining whether a Merger Partner Material Adverse Effect has occurred, unless such Effects are otherwise excepted from this definition) or (h) resulting from the taking of any action required to be taken by this Agreement, except in each case with respect to clauses (a) through (c), to the extent disproportionately affecting Merger Partner and its Subsidiaries, taken as a whole, relative to other similarly situated companies in the industries in which Merger Partner and its Subsidiaries operate (and then only the amount of such disproportionate impact shall be taken into account).

"Merger Partner Member Approval" means approval of (i) a majority of the issued and outstanding Series D Preferred Units in Merger Partner, (ii) a majority of the issued and outstanding Series F Preferred Units in Merger Partner, (iii) a majority of the issued and outstanding Series F Preferred Units in Merger Partner and (iv) a majority of the issued and outstanding Preferred Units in Merger Partner in each of cases (i) through (iv) to (a) agree to convert all issued and outstanding Preferred Units in Merger Partner into Voting Common Units on a one-for-one basis and (b) approve the Transactions.

"Merger Partner Merger Effective Time" has the meaning set forth in Section 1.1(b)(iii).

"Merger Partner PIU" means Common Units of the Merger Partner, which are "profits interests" (as described in IRS Revenue Procedure 93-27 and IRS Revenue Procedure 2001-43) in Merger Partner.

"Merger Partner Preferred Unit" means a membership unit representing a preferred membership interest in Merger Partner, which may be designated as a Series A Preferred Unit, Series B Preferred Unit, Series C Preferred Unit, Series D Preferred Unit, Series E Preferred Unit or Series F Preferred Unit.

"Merger Partner Privacy Requirements" has the meaning set forth in Section 3.20(a).

- "Merger Partner Support Agreement" has the meaning set forth in the Recitals.
- "Merger Partner Support Holders" has the meaning set forth in the Recitals.
- "Merger Partner Termination Fee" has the meaning set forth in Section 8.3(c).
- "Merger Partner Unit" means a membership unit representing a membership interest in Merger Partner, which may be designated as a Series A Preferred Unit, Series B Preferred Unit, Series B Preferred Unit, Series C Preferred Unit, Series E Preferred Unit, Series F Preferred Unit, Voting Common Units.
 - "Merger Partner Warrant" means a warrant to purchase membership units representing a membership interest in Merger Partner.
 - "Merger Subsidiary" has the meaning set forth in the first paragraph of this Agreement.
 - "Ninth Blocker Merger" has the meaning set forth in the Recitals.
 - "OFCCP" has the meaning set forth in Section 3.14(g).
- "Ordinary Course of Business" means the ordinary course of business consistent with past custom and practice (including with respect to frequency and amount) of the applicable Person, including, without limitation, with respect to the preparation and making of filings with the SEC.
- "Organizational Documents" means, with respect to any Person (other than an individual), (a) the certificate or articles of incorporation or organization and any joint venture, limited liability company, operating or partnership agreement and other similar documents adopted or filed in connection with the creation, formation or organization of such Person and (b) all by-laws, stockholder agreements, voting agreements and similar documents, instruments or agreements relating to the organization or governance of such Person, in each case, as amended or supplemented.
 - "Outside Date" means May 15, 2023, as may be extended pursuant to Section 8.1(b).
- "Participating Member" means any member of Merger Partner prior to the Merger Partner Merger Effective Time, other than the Blockers, the Electing Members or Public Company or its Subsidiaries.
- "Pass-Through Income Tax" means any income Tax with respect to which the holders of units of Merger Partner outstanding as of immediately prior to the Closing (or any of their direct or indirect owners) would be primarily liable as a matter of Tax Law (e.g., the income Tax liability for items of income, gain, loss, deduction and credit passed-through to owners of an entity treated as a partnership for U.S. federal income Tax purposes).
 - "Pass-Through Return" has the meaning set forth in Section 7.3(e).
 - "Patent" has the meaning set forth in the definition of "Intellectual Property" herein.

"Payment Fund" has the meaning set forth in Section 2.2(a).

"<u>Permits</u>" means all permits, licenses, registrations, certificates, orders, exemptions, approvals, franchises, variances, clearances and similar rights issued by or obtained from any Governmental Entity (including those issued or required under Environmental Laws and those relating to the occupancy or use of owned or leased real property).

"Permitted Lien" means than (a) mechanic's, material men's and similar liens to secure claims for labor, materials or supplies arising in the Ordinary Course of Business of the applicable party and its Subsidiaries, (b) liens arising under worker's compensation, unemployment insurance, social security, retirement and similar legislation, (c) liens on goods in transit incurred pursuant to documentary letters of credit, in each case arising in the Ordinary Course of Business of the applicable Party and its Subsidiaries, (d) any liens for current Taxes not yet due and payable or for Taxes that are being contested in good faith and, in each case, for which adequate reserves have been made in accordance with GAAP on the applicable party's unaudited interim balance sheet, as applicable, (e) statutory liens to secure obligations to landlords, lessors or renters under leases or rental agreements, (f) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance or similar programs mandated by Law in the Ordinary Course of Business of the applicable party and its Subsidiaries, (g) non-exclusive licenses of Intellectual Property granted by a party or any of its Subsidiaries, as applicable, in the Ordinary Course of Business and that do not (in any case or in the aggregate) materially detract from the value of the Intellectual Property subject thereto, and (h) liens that, individually or in the aggregate, are not material to Merger Partner (and its Subsidiaries) or Public Company (and its Subsidiaries), as applicable, taken as a whole.

"Person" means any natural Person, firm, limited liability company, general or limited partnership, association, corporation, unincorporated organization, company, joint venture, trust, Governmental Entity or other entity.

"Personal Information" has the definition that "Personal information," "Personal data," "protected health information" and similar terms are assigned under applicable Law.

"Pre-Closing Period" means the period commencing on the date of this Agreement and ending at the Effective Time or such earlier date as this Agreement is terminated in accordance with its terms.

"Pre-Closing Tax Period" means any taxable period ending on or before the Closing Date and the portion of any Straddle Period through and including the Closing Date.

"Privacy Laws" means any Laws that govern the receipt, collection, compilation, use, storage, sharing, safeguarding, security, disposal, destruction, disclosure, transfer or other processing of Personal Information, and any Laws governing privacy, data security, data or security breach notification, and any order of a Governmental Entity dealing with same, including, without limitation: (i) Section 5 of the Federal Trade Commission Act, the Health Information Portability and Accountability Act, the CAN-SPAM Act, the Telephone Consumer Protection Act, the California Online Privacy Protection Act, the California Consumer Privacy Act, the Illinois Biometric Information Privacy Act and other Laws governing the processing of

biometric data, the Massachusetts Data Security Regulations set forth at 201 CMR 17.00, and any other United States federal and state Laws implementing, supplementing or amending the foregoing or otherwise concerning privacy, data protection, data security, or electronic marketing; and (ii) any Laws in any relevant jurisdiction concerning privacy, data protection, data security, or electronic marketing.

"Proxy Statement/Prospectus" means the proxy statement/prospectus, which shall be included in the Registration Statement, to be sent to Public Company's stockholders in connection with the Public Company Stockholders' Meeting.

"Public Company" has the meaning set forth in the first paragraph of this Agreement.

"Public Company Benefit Plan" means any Employee Benefit Plan in respect of any employees, independent contractors, directors, officers or shareholders of Public Company or any of its Subsidiaries that are sponsored or maintained by Public Company or any Subsidiary or with respect to which Public Company or any of its Subsidiaries is required to make payments, transfers or contributions or has or may have any actual, potential or contingent liability.

"Public Company Board" means the board of directors of Public Company.

"Public Company Board Adverse Recommendation Change" has the meaning set forth in Section 5.9(c).

"Public Company Board Recommendation" has the meaning set forth in Section 5.9(c).

"<u>Public Company Change in Circumstance</u>" means a change in circumstances (other than an Acquisition Proposal) that affects the business, assets or operations of Public Company that occurs or arises after the date of this Agreement that was neither known to Public Company or Public Company Board nor reasonably foreseeable on, or prior to, the date of this Agreement.

"Public Company Class A Common Stock" means the Class B Common Stock, par value \$0.0001 per share, of Public Company.

"Public Company Class B Common Stock" means the Class B Common Stock, par value \$0.0001 per share, of Public Company.

"Public Company Common Stock" means the common stock, \$0.0001 par value per share, of Public Company.

"Public Company Contract" has the meaning set forth in Section 4.12(a).

"Public Company Disclosure Schedule" means Public Company Disclosure Schedule provided by Public Company to Merger Partner on the date hereof.

"Public Company Employee" means any employee (whether current or former) of Public Company or any of its Subsidiaries.

"Public Company Financial Statements" means: each of the consolidated financial statements (including, in each case, any related notes and schedules) contained in Public Company SEC Reports.

"Public Company Insurance Policy" has the meaning set forth in Section 4.20(a).

"Public Company Intellectual Property" has the meaning set forth in Section 4.11.

"Public Company IT Systems" means any computer systems, networks, hardware, software and other information technology equipment used in connection with the business of Public Company and its Subsidiaries.

"Public Company Material Adverse Effect" means any Effect that, considered together with all other Effects, has or would reasonably be expected to have a material adverse effect on the business, condition (financial or otherwise), assets, liabilities or results of operations of Public Company or its Subsidiaries, taken as a whole; provided, however, that Effects resulting from the following shall not be taken into account in determining whether there has been a Public Company Material Adverse Effect; (a) general business or economic conditions generally affecting the industry in which Public Company and its Subsidiaries operate, (b) acts of war, the outbreak or escalation of armed hostilities, acts of terrorism (including cyberterrorism), earthquakes, wildfires, hurricanes or other natural disasters, health emergencies, including pandemics (including COVID-19) and related or associated epidemics, disease outbreaks and any governmental or industry responses thereto (including quarantine restrictions), (c) changes in financial, banking, securities markets, or general economic, regulatory, legislative or political conditions (including changes in interest or exchange rates), (d) any change in, or any compliance with or action taken for the purpose of complying with, any Law or GAAP (or interpretations of any Law or GAAP), (e) resulting from the announcement of this Agreement or the pendency of the Transactions, (g) the failure of Public Company to meet internal expectations or projections or the results of operations of Public Company in and of itself (it being understood, however, that any Effect causing or contributing to such failure may be taken into account in determining whether a Public Company Material Adverse Effect has occurred, unless such Effects are otherwise excepted from this definition) or (h) resulting from the taking of any action required to be taken by this Agreement, except in each case with respect to clauses (a) through (c), to the extent disproportionately affecting Public Company and its Subsidiaries, taken as a whole, relative to other similarly situated companies in the industries in which Public Company and its Subsidiaries operate (and then only the amount of such disproportionate impact shall be taken into account).

"Public Company Privacy Requirements" has the meaning set forth in Section 4.21(a).

"Public Company PSU" has the meaning set forth in Section 4.2(b).

"Public Company RSU" has the meaning set forth in Section 4.2(b).

"Public Company SEC Reports" has the meaning set forth in Section 4.6.

"Public Company Stock Option" has the meaning set forth in Section 4.2(b).

"Public Company Stock Plan" has the meaning set forth in Section 4.2(b).

"Public Company Stockholder Approval" means the approval of the Public Company Stockholder Matters and the Other Public Company Stockholder Matters by the stockholders of Public Company.

"Public Company Stockholder Matters" has the meaning set forth in Section 5.9(a).

"Public Company Stockholder Meeting" has the meaning set forth in Section 5.9(a).

"Public Company Support Stockholders" has the meaning set forth in the Recitals.

"Public Company Support Agreements" has the meaning set forth in the Recitals.

"Public Company Termination Fee" has the meaning set forth in Section 8.3(b).

"Public Company Transaction Litigation" has the meaning set forth in Section 7.4.

"Registration Rights Agreement" has the meaning set forth in the Recitals.

"Registration Statement" means the registration statement on Form S-4 (or any other applicable form under the Securities Act to register Public Company Common Stock) to be filed with the SEC by Public Company registering the public offering and sale of Public Company Common Stock to the equity holders of Blockers and Merger Partner in the Mergers, as may be amended prior to the time it is declared effective by the SEC.

"Representatives" means, with respect to any Person, such Person's directors, officers, employees, agents, attorneys, accountants, investment bankers, advisors and representatives.

"Required Public Company Stockholder Vote" means the affirmative vote of the holders of a majority of the issued and outstanding shares Public Company Common Stock entitled to vote thereon at a meeting of the holders of Public Company Common Stock duly called and convened in accordance with the Proxy Statement/Prospectus, Delaware Law, Public Company's Organizational Documents and the rules and regulations of Nasdaq.

"Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002.

"SEC" means the United States Securities and Exchange Commission.

"Second Blocker Merger" has the meaning set forth in the Recitals.

"Securities Act" means the Securities Act of 1933, as amended.

"Seventh Blocker Merger" has the meaning set forth in the Recitals.

"Sixth Blocker Merger" has the meaning set forth in the Recitals.

"Straddle Period" means any taxable period that beings on or before the Closing Date and ends after the Closing Date.

"Subsidiary" means, with respect to any Person, any corporation, partnership, trust, limited liability company or other non-corporate business enterprise in which such Person (or another Subsidiary of such Person) holds stock or other ownership interests representing (a) more than 50% of the voting power of all outstanding stock or ownership interests of such entity or (b) the right to receive more than 50% of the net assets of such entity available for distribution to the holders of outstanding stock or ownership interests upon a liquidation or dissolution of such entity.

"Superior Offer" means an unsolicited bona fide written Acquisition Proposal (with all references to 20% in the definition of Acquisition Transaction being treated as references to greater than 50% for these purposes) that: (a) was not obtained or made as a result of a breach of (or in violation of) this Agreement; (b) is on terms and conditions that Public Company Board (or, in the case of a Superior Offer with respect to Merger Partner, the managers of Merger Partner) determines in good faith, based on such matters that it deems relevant (including the likelihood of consummation thereof, the financing terms thereof, any break-up fees, and conditions to consummation), as well as any written offer by Merger Partner (or, in the case of a Superior Offer with respect to Merger Partner, by Public Company) to amend the terms of this Agreement, and following consultation with its outside legal counsel and outside financial advisors, if any, are more favorable, from a financial point of view, to Public Company's stockholders (or, in the case of a Superior Offer with respect to Merger Partner, the members of Merger Partner and the Blockers) than the terms of the Transactions; (c) is not subject to any financing conditions (and if financing is required, such financing is then fully committed to the third party); and (d) is reasonably capable of being completed on the terms proposed without unreasonable delay.

"Tax" or "Taxes" means any and all taxes, charges, fees, duties, contributions, levies or other similar assessments or liabilities, including income, gross receipts, corporation, ad valorem, premium, value-added, net worth, capital stock, capital gains, documentary, recapture, alternative or add-on minimum, disability, registration, recording, excise, real property, personal property, sales, use, license, lease, service, service use, transfer, withholding, employment, unemployment, insurance, social security, national insurance, business license, business organization, environmental, workers compensation, payroll, profits, severance, stamp, occupation, escheat, windfall profits, customs duties, franchise, estimated and other taxes of any kind whatsoever imposed by the United States of America or any state, local or foreign government, or any agency or political subdivision thereof, and any interest, fines, penalties, assessments or additions to tax imposed with respect to or related to such items.

"Tax Contest" has the meaning set forth in Section 7.3(g).

"Tax Receivable Agreement" has the meaning set forth in the Recitals.

"<u>Tax Returns</u>" means any and all reports, returns (including information returns), declarations, or statements relating to Taxes, including any schedule or attachment thereto and any amendment thereof, filed with or submitted to, or required to be filed with or submitted to, any Governmental Entity in connection with the determination, assessment, collection or payment of Taxes.

- "Tenth Blocker Merger" has the meaning set forth in the Recitals.
- "Termination Fee" means the Public Company Termination Fee or the Merger Partner Termination Fee, as applicable.
- "Third Blocker Merger" has the meaning set forth in the Recitals.
- "Top 99 Merger Partner Holder" means, any holder of Merger Partner Units (other than Blockers) that, prior to the Mailing Date, is among the 99 largest members of Merger Partner (or such lower number of largest members treated as 99 partners for purposes of the private placement safe harbor set forth in Treasury Regulations Section 1.7704-1(h)), measured based on each member's membership interest in Merger Partner (including warrants or options to purchase and instruments convertible into membership interests of Merger Partner), on a fully-diluted, as converted basis (including any Merger Partner Units subject to an irrevocable commitment as of the date hereof to purchase Merger Partner Units at any point prior to the Blocker Merger Effective Time for the First Blocker Merger). For purposes of this definition, "holder of Merger Partner Units" shall include any investor who has delivered an irrevocable commitment as of the date hereof to purchase Merger Partner Units at any point prior to the Blocker Merger Effective Time for the First Blocker Merger.
 - "Trade Secret" has the meaning set forth in the definition of "Intellectual Property" herein.
 - "Trademark" has the meaning set forth in the definition of "Intellectual Property" herein.
 - "Transactions" mean, collectively, the Mergers, the Asset Transfers and the other transactions contemplated by this Agreement.
 - "Transfer Taxes" has the meaning set forth in Section 7.3(a).
 - "Treasury Regulations" means the United States Treasury regulations promulgated under the Code.
- "<u>Unitholder Representative</u>" means Rory Riggs, or any successor to such Person pursuant to the Unitholder Representative Engagement Agreement, dated as of the Closing Date, by and between Unitholder Representative and Merger Partner, in substantially the form attached hereto as Exhibit I.
 - "Up-C Unit" has the meaning set forth in Section 2.1(b)(i)(B).
 - "WARN Act" has the meaning set forth in Section 3.15(g).
- "Willful Breach" means an intentional and material breach of this Agreement that is the consequence of an act or omission by the breaching party with the actual knowledge that the taking of such act or failure to take such act would cause or constitute such material breach and result in the consequences caused by such material breach.
 - "Written Consents" has the meaning set forth in the Recitals.

ARTICLE X MISCELLANEOUS

- 10.1 Notices. All notices and other communications hereunder shall be in writing and shall be deemed to have been duly delivered and received hereunder (a) one (1) Business Day after being sent for next Business Day delivery, fees prepaid, via a reputable international overnight courier service, (b) upon delivery in the case of delivery by hand or (c) on the date delivered in the place of delivery if sent by email or facsimile (with a written or electronic confirmation of delivery) prior to 6:00 p.m. New York City time, otherwise on the next succeeding Business Day, in each case to the intended recipient as set forth below:
 - (a) if to Public Company or Merger Subsidiary (prior to the Closing), to:

Calyxt, Inc.

2800 Mount Ridge Road

Roseville, MN 55113

Attention: Debra Frimerman, General Counsel

E-mail: [***]

with a copy (which shall not constitute notice) to:

Sidley Austin LLP

787 Seventh Ave

New York, New York 10019

Attention: John H. Butler

Email: [***]

(b) if to Merger Partner or to Public Company (after the Closing):

Cibus Global, LLC

6455 Nancy Ridge Drive

San Diego, CA 92121

Attention: Rory Riggs, Chief Executive Officer, and Wade King, MD, Chief Financial Officer

E-mail: [***] and [***]

with a copy (which shall not constitute notice) to:

Jones Day

4655 Executive Drive, Suite 1500

San Diego, CA 92121-3134

Attention: Cameron A. Reese

E-mail: [***]

(c) if (prior to the Closing) to any Blocker, to the applicable Blocker address (with a copy which shall not constitute notice) listed on Schedule C.

- 10.2 Entire Agreement. This Agreement (including the schedules and exhibits hereto and the documents and instruments referred to in this Agreement that are to be delivered at the Closing) constitutes the entire agreement among the parties to this Agreement and supersedes any prior understandings, agreements or representations by or among the parties hereto, or any of them, written or oral, with respect to the subject matter hereof; provided that the Confidentiality Agreement shall remain in effect in accordance with its terms.
- 10.3 <u>Third-Party Beneficiaries</u>. This Agreement is not intended to, and shall not, confer upon any other Person any rights or remedies hereunder, except that Indemnified Persons shall be third-party beneficiaries of Section 7.1.
- 10.4 <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations under this Agreement may be assigned or delegated, in whole or in part, by operation of Law or otherwise by any of the parties hereto without the prior written consent of the other parties, and any such assignment without such prior written consent shall be null and void. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of, and be enforceable by, the parties hereto and their respective successors and permitted assigns.
- 10.5 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term.
- 10.6 <u>Counterparts and Signature</u>. This Agreement may be executed in two (2) or more counterparts, each of which shall be deemed an original but all of which together shall be considered one (1) and the same agreement and shall become effective when counterparts have been signed by each of the parties hereto and delivered to the other parties, it being understood that all parties need not sign the same counterpart. This Agreement may be executed and delivered by facsimile or by an electronic scan delivered by electronic transmission.
- 10.7 Interpretation. Except where expressly stated otherwise in this Agreement, the following rules of interpretation apply to this Agreement: (a) "either" and "or" are not exclusive and "include," "includes" and "including" are not limiting; (b) "hereof," "hereby," "herein" and "hereunder" and words of similar import when used in this Agreement refer to this Agreement as a whole and not to any particular provision of this Agreement; (c) "date of this Agreement" refers to the date set forth in the initial caption of this Agreement; (d) "extent" in the phrase "to the extent" means the degree to which a subject or other thing extends, and such phrase does not mean simply "if"; (e) the descriptive headings and table of contents included

herein are included for convenience only and shall not affect in any way the meaning or interpretation of this Agreement or any provision hereof; (f) definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms; (g) references to a contract or agreement mean such contract or agreement as amended or otherwise supplemented or modified from time to time; (h) references to a Person are also to its permitted successors and assigns; (i) references to an "Article," "Section," "Exhibit" or "Schedule" refer to an Article or Section of, or an Exhibit or Schedule to, this Agreement; (j) references to "\$" or otherwise to dollar amounts refer to the lawful currency of the United States; (k) references to a federal, state, local or foreign Law include any rules, regulations and delegated legislation issued thereunder; (1) references to "Party" or "party" refer to a party to this Agreement; (m) references to accounting terms used and not otherwise defined herein have the meaning assigned to them under GAAP and (n) references to a "day" mean calendar days. When reference is made in this Agreement to information that has been "made available" to Public Company, that shall consist of only the information that was (i) contained in Merger Partner's electronic data room no later than 5:00 p.m., Eastern time, on the second (2nd) Business Day prior to the date of this Agreement or (ii) delivered to Public Company or its counsel prior to the date of this Agreement. When reference is made in this Agreement to information that has been "made available" to Merger Partner, that shall consist of only the information that was (i) contained in Public Company's electronic data room no later than 5:00 p.m., Eastern time, on the second (2nd) Business Day prior to the date of this Agreement or (ii) delivered to Merger Partner or its counsel prior to the date of this Agreement. The language used in this Agreement shall be deemed to be the language chosen by the parties hereto to express their mutual intent, and no rule of strict construction shall be applied against any party hereto. No summary of this Agreement prepared by any party shall affect the meaning or interpretation of this Agreement. If any date on which a party is required to make a payment or a delivery pursuant to the terms hereof is not a Business Day, then such party shall make such payment or delivery on the next succeeding Business Day. Time shall be of the essence in this Agreement.

10.8 Governing Law. This Agreement (and any claims or disputes arising out of or related hereto or the Transactions or to the inducement of any party to enter herein, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall be governed in all respects, including validity, interpretation, and effect, by and construed in accordance with the internal Laws of the State of Delaware (including in respect of the statute of limitations or other limitations period applicable to any claim, controversy or dispute) without giving effect to any choice or conflict of Law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of Laws of any jurisdictions other than those of the State of Delaware.

10.9 Remedies.

(a) Except as otherwise expressly provided in this Agreement, any and all remedies herein expressly conferred upon a party will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by Law or equity upon such party, and the exercise by a party of any one (1) remedy will not preclude the exercise of any other remedy.

- (b) The parties hereto agree that irreparable damage would occur in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. It is accordingly agreed that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, in each case without posting a bond or undertaking, this being in addition to any other remedy to which they are entitled at Law or in equity. Each of the parties agrees that it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that (i) the party seeking such remedy has an adequate remedy at Law or (ii) an award of specific performance is not an appropriate remedy for any reason at Law or equity.
- 10.10 <u>Confidentiality</u>. The parties acknowledge that Public Company and Merger Partner have previously executed the Confidentiality Agreement, which Confidentiality Agreement shall continue in full force and effect in accordance with its terms, except as expressly modified herein.
- 10.11 <u>Submission to Jurisdiction</u>. Each of the parties to this Agreement (a) consents to submit itself to the exclusive Personal jurisdiction of the Court of Chancery of the State of Delaware, New Castle County, or, if that court does not have jurisdiction, a federal court sitting in Wilmington, Delaware in any action or proceeding arising out of or relating to this Agreement or any of the Transactions, (b) agrees that all claims in respect of such action or proceeding may be heard and determined in any such court, (c) agrees that it shall not attempt to deny or defeat such Personal jurisdiction by motion or other request for leave from any such court, (d) agrees not to bring any action or proceeding arising out of or relating to this Agreement or any of the Transactions in any other court, and (e) waives any right it may have to a trial by jury with respect to any action or proceeding arising out of or relating to this Agreement. Each of the parties hereto waives any defense of improper venue or inconvenient forum to the maintenance of any action or proceeding so brought and waives any bond, surety or other security that might be required of any other party with respect thereto. Any party hereto may make service on another party by sending or delivering a copy of the process to the party to be served at the address and in the manner provided for the giving of notices in Section 10.1. Nothing in this Section 10.11, however, shall affect the right of any party to serve legal process in any other manner permitted by Law. Nothing in this Section 10.11 shall limit the right of a party to seek injunctive relief under Section 10.9(b) in any applicable jurisdiction.
- 10.12 Non-Survival of Representations, Warranties and Covenants. None of the representations, warranties, covenants, obligations, agreements or other provisions in this Agreement or in any certificate, statement or instrument delivered pursuant to this Agreement, including any rights arising out of any breach of such representations, warranties, covenants, obligations, agreements or other provisions, shall survive the Closing, and each such representation, warranty, covenant, obligation, agreement or other provision shall terminate and expire upon the occurrence of the Merger Partner Merger Effective Time (and there shall be no liability after the Closing in respect thereof), except for (a) those covenants and agreements contained herein that by their terms expressly apply in whole or in part at or after the Closing, and then only with respect to any breaches occurring at or after the Closing, and (b) this ARTICLE X.

[Remainder of the Page Intentionally Left Blank]

MERGER SUBSIDIARY:

CALYPSO MERGER SUBSIDIARY, LLC

By Its Managing Member:

CALYXT, INC.

By: <u>/s/ Debra Frimerman</u>

Name: Debra Frimerman

Title: General Counsel and Corporate Secretary

PUBLIC COMPANY:

CALYXT, INC.

By: /s/ Debra Frimerman

Name: Debra Frimerman Title: General Counsel and Corporate Secretary

MERGER PARTNER:

CIBUS GLOBAL, LLC

By: /s/ Rory Riggs Name: Rory Riggs Title: Chief Executive Officer

BLOCKER 1:

NEW VENTURES I HOLDINGS, LLC

By: /s/ Jonathan Finn
Name: Jonathan Finn

Title: Member and Managing Member of BV

Partners LLC, acting as Managing Member of New Ventures I LLC, acting as sole member and managing member of New Ventures I Holdings,

LLC

BLOCKER 2:

BCGF CB HOLDINGS LLC

By: Fidelity Securities Fund: Fidelity Blue Chip

Growth Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005

Email: BBH. Fidelity. CA. Notifications@BBH. com

Phone Number: 617-772-1070 Fax Number: 201-325-3651

BLOCKER 3:

BCGFCP CB HOLDINGS LLC

By: Fidelity Blue Chip Growth Commingled Pool,

its manager

By: Fidelity Management Trust Company, as

Trustee

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault

140 Broadway New York, NY 10005

Email: BBH.Fidelity.CA.Notifications@BBH.com

Phone Number: 617-772-1070 Fax Number: 201-325-3651

BLOCKER 4:

BCGFK CB HOLDINGS LLC

By: Fidelity Securities Fund: Fidelity Blue Chip

Growth K6 Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Fax number: 312-557-5417

Address:

The Northern Trust Company
Attn: Trade Securities Processing
333 South Wabash Ave, 32nd Floor
Chicago, Illinois 60604
Fidelity Securities Fund: Fidelity Blue Chip
Growth K6 Fund
Reference Account # F71082
Email: NTINQUIRY@NTRS.COM

BLOCKER 5:

FSBCGF CB HOLDINGS LLC

By: Fidelity Securities Fund: Fidelity Series Blue

Chip Growth Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

State Street Bank & Trust
PO Box 5756
Boston, Massachusetts 02206
Attn: WAVECHART + CO fbo Fidelity Securities
Fund: Fidelity Series Blue Chip Growth Fund
Email: SSBCORPACTIONS@StateStreet.com
Fax Number: 617-988-9110

BLOCKER 6:

PYLBCG CB HOLDINGS LLC

By: FIAM Target Date Blue Chip Growth

Commingled Pool, its manager

By: Fidelity Institutional Asset Management Trust

Company as Trustee

By: /s/ Chris Maher
Name: Chris Maher
Title: Authorized Signatory

Address:

State Street Bank & Trust PO Box 5756 Boston, Massachusetts 02206

Attn: WAVECHART + CO fbo Fidelity Securities Fund: Fidelity Series Blue Chip Growth Fund Email: SSBCORPACTIONS@StateStreet.com

Fax Number: 617-988-9110

BLOCKER 7:

FSGRWCO CB HOLDINGS LLC

By: Fidelity Mt. Vernon Street Trust: Fidelity Series

Growth Company Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005

Email: BBH. Fidelity. CA. Notifications@BBH. com

BLOCKER 8:

GROWTHCO CB HOLDINGS LLC By: Fidelity Mt. Vernon Street Trust: Fidelity Growth Company Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

BNY Mellon PO Box 392002 Pittsburgh PA 15230

Email: fidelitycorporateevents@bnymellon.com

BLOCKER 9:

GRTHCOCP CB HOLDINGS LLC

By: Fidelity Growth Company Commingled Pool,

its manager

By: Fidelity Management Trust Company, as

Trustee

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005

Email: BBH.Fidelity.CA.Notifications@BBH.com

BLOCKER 10:

GRWTHCOK CB HOLDINGS LLC By: Fidelity Mt. Vernon Street Trust: Fidelity Growth Company K6 Fund, its manager

By: /s/ Chris Maher Name: Chris Maher Title: Authorized Signatory

Address:

BNY Mellon PO Box 392002 Pittsburgh PA 15230

Email: fidelitycorporateevents@bnymellon.com

Schedule A Support Stockholders

- 1. Cellectis S.A.
- 2. Michael Carr
- 3. Philippe Dumont
- 4. Jonathan Fassberg
- 5. Ewa Kozicz
- 6. Kim Nelson
- 7. Chris Neugent
- 8. Yves Ribeill
- 9. Bill Koschak
- 10. Travis Frey
- 11. Debra Frimerman

Schedule B Support Holders

- 1. New Ventures I Holdings, LLC
- 2. BCGF CB Holdings LLC
- 3. BCGFCP CB Holdings LLC
- 4. BCGFK CB Holdings LLC
- 5. FSBCGF CB Holdings LLC
- 6. PYLBCG CB Holdings LLC
- 7. FSGRWCO CB Holdings LLC
- 8. GROWTHCO CB Holdings LLC
- 9. GRTHCOCP CB Holdings LLC
- 10. GRWTHCOK CB Holdings LLC
- 11. Rory Riggs
- 12. Wade King, MD
- 13. Peter Beetham, Ph.D.
- 14. Greg Gocal Ph.D.
- 15. Jean-Pierre Lehmann
- 16. Mark Finn
- 17. Gerhard Prante
- 18. Keith Walker
- 19. Cibus Investments LLC
- 20. Lufam Genetic Biotech Ltd.
- 21. Rowiss Biotech Investment Ltd.

Schedule C

Notice Information for Blocker 1 through Blocker 10 pursuant to Section 10.1 of the Agreement:

Blocker:	Name	Notice to:	With a copy (which shall not constitute notice) to:
Blocker 1	New Ventures I Holdings, LLC	New Ventures I Holdings, LLC Attn: Jonathan Finn c/o BV PARTNERS LLC 3500 Pacific Avenue Virginia Beach, VA 23451	Akin Gump Strauss Hauer & Feld LLP One Bryant Park Bank of America Tower New York, NY 10036-6745 Attention: Tuvia Tendler E-mail: [***]
Blocker 2	BCGF CB HOLDINGS LLC	Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005 BBH.Fidelity.CA.Notifications@BBH.com Phone Number: 617-772-1070 Fax Number: 201-325-3651	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 3	BCGFCP CB HOLDINGS LLC	Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005 BBH.Fidelity.CA.Notifications@BBH.com Phone Number: 617-772-1070 Fax Number: 201-325-3651	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]

Blocker:	Name	Notice to:	With a copy (which shall not constitute notice) to:
Blocker 4	BCGFK CB HOLDINGS LLC	The Northern Trust Company Attn: Trade Securities Processing 333 South Wabash Ave, 32nd Floor Chicago, Illinois 60604 Fidelity Securities Fund: Fidelity Blue Chip Growth K6 Fund Reference Account # F71082 Email: NTINQUIRY@NTRS.COM Fax number: 312-557-5417	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 5	FSBCGF CB HOLDINGS LLC	State Street Bank & Trust PO Box 5756 Boston, Massachusetts 02206 Attn: WAVECHART + CO fbo Fidelity Securities Fund: Fidelity Series Blue Chip Growth Fund Email: SSBCORPACTIONS@StateStreet.com Fax number: 617-988-9110	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 6	PYLBCG CB HOLDINGS LLC	State Street Bank & Trust PO Box 5756 Boston, Massachusetts 02206 Attn: FLAPPER CO fbo FIAM Target Date Blue Chip Growth Commingled Pool Email: SSBCORPACTIONS@StateStreet.com Fax number: 617-988-9110	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 7	FSGRWCO CB HOLDINGS LLC	Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005 BBH.Fidelity.CA.Notifications@BBH.com	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]

Blocker:	Name	Notice to:	With a copy (which shall not constitute notice) to:
Blocker 8	GROWTHCO CB HOLDINGS LLC	BNY Mellon PO Box 392002 Pittsburgh PA 15230 fidelitycorporateevents@bnymellon.com	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 9	GRTHCOCP CB HOLDINGS LLC	Mag & Co. c/o Brown Brothers Harriman & Co. Attn: Corporate Actions /Vault 140 Broadway New York, NY 10005 BBH.Fidelity.CA.Notifications@BBH.com	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]
Blocker 10	GRWTHCOK CB Holdings LLC	BNY Mellon PO Box 392002 Pittsburgh PA 15230 fidelitycorporateevents@bnymellon.com	Fenwick & West LLP 555 California Street San Francisco, CA 94104 Attention: Jordan Roberts E-mail: [***]

Exhibit A Public Company Support Agreements

[See attached.]

VOTING AGREEMENT

VOTING AGREEMENT, dated as of January 13, 2023 (this "<u>Agreement</u>"), by and between Cibus Global, LLC, a Delaware limited liability company ("<u>Merger Partner</u>"), and the undersigned stockholders (each, a "<u>Holder</u>" and collectively, "<u>Holders</u>") of Calyxt, Inc., a Delaware corporation (the "<u>Public Company</u>")[, and, solely for purposes of Section 1.6 and Annex A, the Public Company.] Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Merger Agreement (as herein defined).

RECITALS

WHEREAS, concurrently with the execution and delivery of this Agreement, Public Company, Calypso Merger Subsidiary, LLC, a Delaware limited liability company and wholly-owned subsidiary of Public Company ("Merger Subsidiary"), Merger Partner and the Blockers (as defined in the Merger Agreement) are entering into an Agreement and Plan of Merger, dated as of the date hereof (as the same may be amended or modified in accordance with its terms, the "Merger Agreement");

WHEREAS, as a condition to Merger Partner's willingness to enter into the Merger Agreement, Holders have agreed to enter into this Agreement with respect to the Subject Shares (as defined herein);

WHEREAS, as of the date hereof, each Holder is the beneficial owner of the number of shares of Public Company Common Stock ("<u>Public Company Common Stock</u>"), set forth opposite such Holder's name on <u>Exhibit A</u> hereto (all such shares set forth opposite each such Holder's name, together with any shares of Public Company Common Stock that are hereafter issued to (including upon the exercise of options or warrants) or otherwise acquired, or beneficially owned by such Holder prior to the termination of this Agreement (including the resulting shares of Public Company Common Stock resulting from any stock split, stock dividend, merger, reorganization, recapitalization, reclassification, combination, exchange of shares affecting the shares of Public Company Common Stock) being referred to as the "<u>Subject Shares</u>" of such Holder); and

WHEREAS, the Merger Agreement provides for, among other things, the Mergers and the other Transactions.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

NTD: To be included only in Cellectis version of this Agreement

ARTICLE I Voting and Transfer of Public Company Common Stock

1.1 <u>Voting</u>. Each Holder shall cause to be present and counted and shall vote (or cause to be voted or acted upon by written consent with respect to) all of its Subject Shares at any meeting of the stockholders of Public Company or any adjournment or postponement thereof, and in any action proposed to be taken by written consent of the stockholders of Public Company: (a) to approve the Public Company Stockholder Matters, the Other Public Company Stockholder Matters and any proposal to adjourn or postpone such meeting of stockholders of Public Company to a later date if there are not sufficient votes to approve the Public Company Stockholder Matters and the Other Public Company Stockholder Matters, provided that, for the avoidance of doubt, Holder shall not be required to approve anything other than the Public Company Stockholder Matters and the Other Public Company Stockholder Matters and (b) against any Acquisition Proposal (other than the Transactions) or any other action that would reasonably be expected to materially impede, interfere with, discourage, adversely affect, inhibit or delay the timely consummation of the Transactions or the fulfillment of Public Company or Merger Subsidiary's conditions to Closing under the Merger Agreement. Each Holder agrees that it will not revoke any consent previously executed in support of the Transactions.

1.2 No Transfer. During the period beginning on the date of this Agreement and ending immediately prior to the earlier of the Merger Partner Merger Effective Time and the termination of this Agreement, other than as contemplated by the Merger Agreement or with the written consent of Merger Partner, each Holder shall not, directly or indirectly, except as contemplated by the Merger Agreement: (a) sell, distribute, assign, convey, transfer, grant, pledge, hypothecate, convey any legal or beneficial ownership in (including by merger (including by conversion into securities or other consideration), by tendering into any tender or exchange offer), or otherwise encumber or dispose of any Subject Shares; (b) deposit any Subject Shares into a voting trust or enter into a voting agreement or any other arrangement with respect to any such Subject Shares or purport to grant any proxy with respect thereto; (c) enter into any contract, option or other arrangement or undertaking with respect to the direct or indirect acquisition or sale, assignment, transfer or other disposition of any Subject Shares; (d) otherwise permit any Liens to be created on any Subject Shares (other than Permitted Liens (as defined herein)) or (e) commit or agree to take any of the foregoing actions (any action described in clauses (a), (b), (c), (d) and (e), a "Transfer"); provided, that the foregoing shall not prohibit (i) Transfers between such Holder and any Affiliates of such Holder; (ii) Transfers in connection with any estate planning or charitable giving; (iii) Transfers pursuant to any existing 10b5-1 Plan that has been entered into by such Holder prior to the Merger Partner Merger Effective Time; (iv) exercising any stock option to purchase shares of Public Company Common Stock, or any securities convertible into or exercisable or exchangeable for Public Company Common Stock, or other similar awards granted pursuant to the Public Company Stock Plans; (v) Transfer Public Company Common Stock or any securities convertible into or exercisable or exchangeable for shares of Public Company Common Stock upon a vesting event of Public Company's securities or upon the exercise of options to purchase Public Company Common Stock, in each case on a "cashless" or "net exercise" basis or in a sale-to-cover transaction with respect to tax withholding obligations of such Holder in connection with such vesting or exercise, whether by means of a "net settlement" or otherwise with respect to awards granted pursuant to the Public Company Stock Plans; or (vi) any Transfer required by Law or required by a Legal Proceeding to which such Holder is a party; provided, further, that any Transfer pursuant to clauses (i), (ii) or (vi) of this Section 1.2 shall be permitted only if, prior to any such Transfer, and as a precondition to the effectiveness of any such Transfer, the transferee executes and delivers to Merger Partner a joinder to this Agreement in the form attached hereto as Exhibit B. Any Transfer or action in violation of this Section 1.2 shall be void ab initio.

- 1.3 <u>Solicitation</u>. Each Holder shall, and shall instruct and use its reasonable best efforts to cause the Representatives retained by it and acting on its behalf to:
 - a) immediately cease and cause to be terminated any existing discussions, negotiations and communications with any Person that relate to any Acquisition Proposal or Acquisition Inquiry that has not already been terminated as of the date of this Agreement; and
 - b) not to take any action that Public Company would then be prohibited from taking under Section 5.4(a) of the Merger Agreement as if such Holder were Public Company, provided that the foregoing shall not preclude any Holder from having discussions with Merger Partner or Public Company regarding the Merger Agreement or the transactions contemplated thereby.

Notwithstanding the foregoing, from and after the date of the Merger Agreement until the earlier to occur of (x) the Merger Partner Merger Effective Time and (y) such date and time as the Merger Agreement shall be validly terminated pursuant thereto, if (and only if) Public Company is permitted, pursuant to the Merger Agreement, to have discussions or negotiations in response to an Acquisition Proposal, each Holder and its Representatives shall be permitted to participate in such discussions or negotiations with such Person making such Acquisition Proposal to the same extent as Public Company is permitted to do so under the Merger Agreement (including with respect to providing a voting agreement with respect to such Holder's Subject Shares).

For the avoidance of doubt, and notwithstanding anything in this Agreement to the contrary, (a) no Holder's obligations under this Section 1.3 shall be construed to apply to any actions taken by any other stockholder of Public Company, Public Company or any of its Subsidiaries, the Board of Directors of Public Company or any committee thereof, any of the directors, officers or employees of any other stockholder of Public Company, Public Company or any of its Subsidiaries (in their capacities as such), or the Representatives of Public Company or any of their respective Subsidiaries, (b) no Holder shall be responsible for the actions of any of the foregoing and (c) nothing in this Section 1.3 shall prevent any Holder or its Representatives, in their capacity as a member of Public Company's Board of Directors or as an executive officer of Public Company, as applicable, from engaging in any activity permitted or required by the Merger Agreement in the performance of their fiduciary obligations to Public Company's stockholders.

- 1.4 <u>Further Assurances</u>. Each Holder shall execute and deliver, or cause to be executed and delivered, such further certificates, instruments and other documents as Merger Partner or Public Company may reasonably request for the purpose of effectively carrying out its obligations under this Agreement.
- 1.5 <u>Additional Public Company Stock</u>. Each Holder acknowledges and agrees that any additional shares of Public Company Common Stock with respect to which beneficial ownership is acquired by such Holder (other than pursuant to a vesting event pursuant to Public Company Stock Plans, if any, after the date hereof), by transfer or any other mechanism shall automatically become subject to the terms of this Agreement as though owned by such Holder as of the date hereof.

1.6 <u>Holder Guaranty</u>. [Each of Holder, Merger Partner and Public Company acknowledges and agrees to the matters set forth in <u>Annex A</u> attached hereto, which are hereby incorporated by reference into this Section 1.6.]²

ARTICLE II Representations and Warranties of Each Holder

Each Holder, severally and not jointly, hereby represents and warrants to Merger Partner as to itself (and not as to any other Holder) as follows:

2.1 <u>Authorization</u>. Such Holder has all requisite capacity, power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement. This Agreement has been duly executed and delivered by or on behalf of such Holder and, assuming the due authorization, execution and delivery of this Agreement by the Merger Partner, this Agreement constitutes a legal, valid and binding obligation of such Holder, enforceable against such Holder in accordance with its terms, except to the extent that enforcement may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Laws affecting creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law). Each Holder that is an entity (a) is duly organized, validly existing and in good standing (with respect to jurisdictions that recognize such concept or equivalent) under the Laws of the jurisdiction of its organization, (b) the execution and delivery of this Agreement and such Holder's performance of its obligations under this Agreement have been duly authorized by all necessary corporate or other organizational action on the part of such Holder and no other corporate or other organizational action on the part of such Holder is necessary to authorize the execution and delivery of this Agreement or for such Holder to perform its obligations under this Agreement and (c) no approval by any holder of such Holder's equity, membership or other interests is necessary to approve this Agreement.

2.2 Governmental Filings; No Violations; Certain Contracts.

(a) No consent, approval, order or authorization of, or registration, declaration or filing with or notice to, any Government Authority is required to be obtained or made by such Holder in connection with the execution and delivery of this Agreement, except for (i) such filings and reports as may be required pursuant to the applicable requirements of the Securities Act, the Exchange Act and any other applicable state or federal securities, takeover and "blue sky" Laws (collectively, "Required Filings") and (ii) such consents, authorizations, filings, approvals and registrations which, if not obtained or made, are not reasonably likely to prevent, materially delay or materially impair the performance of such Holder's obligations under this Agreement.

NTD: To be included only in Cellectis version of this Agreement

- (b) The execution and delivery by such Holder of this Agreement does not, and the compliance with the provisions hereof will not, conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of, or result in, termination, cancellation, modification or acceleration of any obligation or to the loss of a material benefit under, or result in the creation of any Lien in or upon any of the properties, assets or rights of such Holder under, or give rise to any increased, additional, accelerated or guaranteed rights or entitlements under, or require any consent, waiver or approval of any Person pursuant to, any provision of (i) any written agreement to which such Holder is a party or by which such Holder or any of his properties or assets may be bound, (ii) any Law applicable to such Holder or by which such Holder or any of its properties or assets may be bound or (iii) following a valid Transfer pursuant to Section 1.2, with respect to such Holder that is an entity, the organizational documents of such Holder, except as would not, individually or in the aggregate, be reasonably expected to prevent or materially delay the ability of such Holder to perform such Holder's obligations under this Agreement or to consummate the transactions contemplated hereby.
- 2.3 Ownership of Subject Shares; Voting Power. Exhibit A hereto correctly sets forth the number of Subject Shares held beneficially by such Holder, as of the date of this Agreement. Such Holder is the sole beneficial holder of all of its Subject Shares and has full voting power and power of disposition with respect to all such Subject Shares free and clear of any liens, claims, proxies, voting trusts or agreements, options or any other encumbrances or restrictions on title, transfer or exercise of any rights of a stockholder in respect of such Subject Shares (collectively, "Encumbrances"), except for any such Encumbrance that may be imposed pursuant to (a) (i) this Agreement, (ii) the organizational documents of Public Company, or (iii) any applicable restrictions on transfer under the Securities Act, including the rules and regulations promulgated thereunder, or any "blue sky" Laws of the various states of the United States or (b) expressly contemplated by the Merger Agreement (collectively, "Permitted Liens"). Except pursuant to this Agreement or the Merger Agreement or as set forth in the organizational documents of Public Company, no Person has any contractual or other right or obligation to purchase or otherwise acquire any of the Holder's Subject Shares.
- 2.4 <u>Disclosure of All Shares Owned</u>. Such Holder does not beneficially own any shares of Public Company Common Stock other than the Subject Shares any options, warrants, or other rights to acquire any additional shares of Public Company Common Stock or any security exercisable for or convertible into shares of Public Company Common Stock, in each case, as set forth on <u>Exhibit A</u> attached hereto.
- 2.5 <u>Reliance</u>. Such Holder understands and acknowledges that the Merger Partner is entering into the Merger Agreement in reliance upon such Holder's execution, delivery and performance of this Agreement.

ARTICLE III Representations and Warranties of Merger Partner

Merger Partner hereby represents and warrants to each Holder as follows:

(a) Merger Partner is an entity duly organized, validly existing and in good standing (with respect to jurisdictions that recognize such concept or equivalent) under the Laws of the jurisdiction of its organization.

(b) Merger Partner has all requisite capacity, power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement. This Agreement has been duly executed and delivered by or on behalf of Merger Partner and, assuming the due authorization, execution and delivery of this Agreement by the Holder, this Agreement constitutes a legal, valid and binding obligation of Merger Partner, enforceable against Merger Partner in accordance with its terms, except to the extent that enforcement may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Laws affecting creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law).

ARTICLE IV General Provisions

- 4.1 <u>Termination</u>. This Agreement and all obligations, covenants and agreements contained herein, including the voting agreements contemplated hereby and the proxies granted hereunder, shall automatically terminate and cease to be effective at the earliest to occur of: (a) the Merger Partner Merger Effective Time; (b) the termination of the Merger Agreement pursuant thereto; (c) the effective date of a written agreement of the parties hereto terminating this Agreement; (d) with respect to the Holder, the date of any modification or amendment to the Merger Agreement, as in effect on the date hereof, in a manner that reduces the amount and/or changes the form of consideration payable thereunder to the Holder or its equityholders, or otherwise adversely effects the Holder or its equityholders in any respect without the prior written approval of the Holder; and (e) a Public Company Board Adverse Recommendation Change in accordance with and to the extent permitted by the Merger Agreement; <u>provided, however</u>, that in the case of any termination pursuant to clause (a) of this sentence (or any other termination that is followed by the occurrence of the Merger Partner Merger Effective Time), Section 1.4 (Further Assurances)[, Section 1.6 (Holder Guaranty) (including Annex A)]³ and this ARTICLE IV shall survive such termination hereof.
- 4.2 <u>Notices</u>. All notices and other communications to any party hereunder shall be in writing and shall be deemed given if personally delivered, emailed (which is confirmed) or sent by overnight courier (providing proof of delivery) to the parties at the following addresses:

If to Merger Partner, to:

Cibus Global, LLC 6455 Nancy Ridge Drive San Diego, CA 92121

Attention: Rory Riggs, Chief Executive Officer, and Wade King, MD, Chief Financial Officer

E-mail: [***] and [***]

3 NTD: To be included only in Cellectis version of this Agreement

With a copy (which shall not constitute notice) to:

Jones Day

4655 Executive Drive

Suite 1500

San Diego, CA 92121-3134 Attention: Cameron A. Reese

E-mail: [***

If to any Holder, to it at: the address listed below such Holder's name on the signature page of such Holder,

With a copy (which shall not constitute notice) to:

[Public Company] 2800 Mount Ridge Road Roseville, MN 55113

Attention: Debra Frimerman, General Counsel

E-mail: [***]

and to:

Sidley Austin LLP 787 Seventh Ave New York, New York 10019

Attention: John H. Butler

Email: [***]

or such other address or email address as such party may hereafter specify by like notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of actual receipt by the recipient thereof if received prior to 5:00 p.m. local time in the place of receipt and such day is a business day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed not to have been received until the next succeeding business day in the place of receipt.

4.3 <u>Severability</u>. If any term, condition or other provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other terms, provisions and conditions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term, condition or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible to the fullest extent permitted by applicable Law.

4.4 Third Party Beneficiaries. This Agreement (including the documents and instruments referred to herein) is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, including the right to rely upon the representations and warranties set forth herein. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of such parties. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance herewith without notice or liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any of the parties hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date hereof or as of any other date.

4.5 Governing Law; Jurisdiction; Specific Performance; Waiver to Jury Trial.

- (a) This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware applicable to contracts executed in and to be performed entirely within that State, regardless of the Laws that might otherwise govern under any applicable conflict of Laws principles.
- (b) All Actions arising out of or relating to this Agreement or the Transactions shall be heard and determined in the Court of Chancery of the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over any Legal Proceeding, any state or federal court within the State of Delaware). The parties hereto hereby irrevocably (i) submit to the exclusive jurisdiction and venue of such courts in any such Legal Proceeding, (ii) waive the defense of an inconvenient forum or lack of jurisdiction to the maintenance of any such Legal Proceeding, (iii) agree to not attempt to deny or defeat such jurisdiction by motion or otherwise request for leave from any such court and (iv) agree to not bring any Legal Proceeding arising out of or relating to this Agreement or the Transactions in any court other than the Court of Chancery of the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over any Legal Proceeding, any state or federal court within the State of Delaware), except for Legal Proceedings brought to enforce the judgment of any such court. The consents to jurisdiction and venue set forth in this Section 4.5(b) shall not constitute general consents to service of process in the State of Delaware and shall have no effect for any purpose except as provided in this paragraph and shall not be deemed to confer rights on any Person other than the parties hereto. Each party hereto agrees that service of process upon such party in any Legal Proceeding arising out of or relating to this Agreement shall be effective if notice is given by overnight courier at the address set forth in Section 4.2 of this Agreement. The parties hereto agree that a final judgment in any such Legal Proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law; provided, however, that nothing in the foregoing shall restrict any party's rights to seek any post-judgment r
- (c) The parties hereto agree that irreparable damage for which monetary relief, even if available, might not be an adequate remedy, would occur in the event that any provision of this Agreement is not performed in accordance with its specific terms or is otherwise breached, including if the parties hereto fail to take any action required of them hereunder to consummate the Mergers or the other Transactions. Subject to the following sentence, the parties acknowledge

and agree that (a) the parties shall be entitled to seek an injunction or injunctions, specific performance or other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in Section 4.5(b), this being in addition to any other remedy to which they are entitled under this Agreement and (b) the right of specific enforcement is an integral part of the Transactions and without that right none of Holders or Merger Partner would have entered into this Agreement or the Merger Agreement. The parties hereto acknowledge and agree that any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 4.5(c) shall not be required to provide any bond or other security in connection with any such order or injunction.

- (d) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVER, (C) IT MAKES SUCH WAIVER VOLUNTARILY AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS SECTION 4.5(d).
- 4.6 Entire Agreement. This Agreement, the Merger Agreement and the other agreements related to the Transactions (and all exhibits and schedules hereto and thereto) collectively constitute the entire agreement, and supersedes all other prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof and thereof.
- 4.7 <u>Amendment or Supplement</u>. Subject to compliance with applicable Law, at any time prior to the Merger Partner Merger Effective Time, this Agreement may be amended or supplemented in any and all respects only by written agreement of the parties hereto.
- 4.8 <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 4.9 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts (including by facsimile or electronic mail), each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties hereto and delivered to the other parties hereto.

* * * * *

-10-

IN WITNESS	HEREOF, the parties hereto have caused this Agreement to be executed by their duly authorized respective officers as of
the date first written above.	

[Signature Page to Voting Agreement]

CIBUS GLOBAL, LLC
By:
Name:
Title:

Solely for purposes of Section 1.6 and Annex A:
CALYXT, INC.
Ву:
Name:
Title:

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their duly authorized respective officers as of the date first written above.
[STOCKHOLDER]
Signature of [STOCKHOLDER]
Printed Name of
[STOCKHOLDER]
Address of [STOCKHOLDER] for purposes of Section 4.2 (Notices) of this Agreement:

EXHIBIT A

Subject Shares

EXHIBIT B FORM OF JOINDER

ANNEX A COVENANTS REGARDING CELLECTIS GUARANTEE

Exhibit B Merger Partner Support Agreement

VOTING AGREEMENT

VOTING AGREEMENT, dated as of January 13, 2023 (this "<u>Agreement</u>"), by and between Calyxt, Inc., a Delaware corporation ("<u>Public Company</u>"), and the undersigned equityholders (each, a "<u>Holder</u>" and collectively, "<u>Holders</u>")¹ of Cibus Global, LLC, a Delaware limited liability company ("<u>Merger Partner</u>"). Capitalized terms used herein and not otherwise defined shall have the meanings assigned to them in the Merger Agreement (as herein defined).

RECITALS

WHEREAS, concurrently with the execution and delivery of this Agreement, Public Company, Calypso Merger Subsidiary, LLC, a Delaware limited liability company and wholly-owned subsidiary of Public Company ("Merger Subsidiary"), Merger Partner and the Blockers (as defined in the Merger Agreement) are entering into an Agreement and Plan of Merger, dated as of the date hereof (as the same may be amended or modified in accordance with its terms, the "Merger Agreement");

WHEREAS, as a condition to Public Company's willingness to enter into the Merger Agreement, Holders have agreed to enter into this Agreement with respect to the Subject Units (as defined herein);

WHEREAS, as of the date hereof, each Holder is the beneficial owner of the number of limited liability membership units of Merger Partner ("Merger Partner Units") set forth opposite such Holder's name on Exhibit A hereto (all such Merger Partner Units set forth opposite each such Holder's name, together with any Merger Partner Units that are hereafter issued to (including upon the exercise of options or warrants) or otherwise acquired, or beneficially owned by such Holder prior to the termination of this Agreement (including the Merger Partner Units resulting from any unit split, unit dividend, merger, reorganization, recapitalization, reclassification, combination, exchange of units affecting Merger Partner Units) being referred to as the "Subject Units" of such Holder); and

WHEREAS, the Merger Agreement provides for, among other things, the Mergers and the other Transactions.

NOW, THEREFORE, in consideration of the premises, and of the representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, agree as follows:

Merger Partner D&O group to sign a separate agreement from the 9 Fidelity Blockers. New Ventures to sign its own agreement.

ARTICLE I Voting and Conversion of Merger Partner Preferred Units

- 1.1 <u>Voting</u>. Each Holder shall cause to be present and counted and shall vote (or cause to be voted or acted upon by written consent with respect to) all of its Subject Units at any meeting of the equityholders of Merger Partner or any adjournment or postponement thereof, and in any action proposed to be taken by written consent of the equityholders of Merger Partner: (a) to approve the Transactions, provided that, for the avoidance of doubt, Holder shall not be required to approve anything other than the Merger Agreement and the consummation of the Transactions; (b) against any Acquisition Proposal (other than the Transactions); (c) to agree to convert all issued and outstanding preferred units of Merger Partner into voting common units of Merger Partner on a one-for-one basis; and (d) against any other action that would reasonably be expected to materially impede, interfere with, discourage, adversely affect, inhibit or delay the timely consummation of the Transactions or the fulfillment of Merger Partner's conditions to Closing under the Merger Agreement. Each Holder agrees that it will not revoke any consent previously executed in support of the Transactions.
- 1.2 No Transfer. During the period beginning on the date of this Agreement and ending immediately prior to the earlier of the applicable Blocker Merger Effective Time and the termination of this Agreement, other than as contemplated by the Merger Agreement or with the written consent of Public Company, each Holder shall not, directly or indirectly, except as contemplated by the Merger Agreement: (a) sell, distribute, assign, convey, transfer, grant, pledge, hypothecate, convey any legal or beneficial ownership in (including by merger (including by conversion into securities or other consideration), by tendering into any tender or exchange offer), or otherwise encumber or dispose of any Subject Units; (b) deposit any Subject Units into a voting trust or enter into a voting agreement or any other arrangement with respect to any such Subject Units or purport to grant any proxy with respect thereto; (c) enter into any contract, option or other arrangement or undertaking with respect to the direct or indirect acquisition or sale, assignment, transfer or other disposition of any Subject Units; (d) otherwise permit any Liens to be created on any Subject Units (other than Permitted Liens (as defined herein)) or (e) commit or agree to take any of the foregoing actions (any action described in clauses (a), (b), (c), (d) and (e), a "Transfer"); provided, that the foregoing shall not prohibit (i) Transfers between such Holder and any Affiliates of such Holder; (ii) Transfers in connection with any estate planning or charitable giving; (iii) exercising any option to purchase Merger Partner Units, or any securities convertible into or exercisable or exchangeable for Merger Partner Units; (iv) Transfer Merger Partner Units or any securities convertible into or exercisable or exchangeable for Merger Partner Units upon a vesting event of Merger Partner's securities or upon the exercise of options or warrants to purchase Merger Partner Units, in each case on a "cashless" or "net exercise" basis or in a sale-to-cover transaction with respect to tax withholding obligations of such Holder in connection with such vesting or exercise, whether by means of a "net settlement" or otherwise; or (v) any Transfer required by Law or required by a Legal Proceeding to which such Holder is a party; provided, further, that any Transfer pursuant to clauses (i), (ii) or (v) of this Section 1.2 shall be permitted only if, prior to any such Transfer, and as a precondition to the effectiveness of any such Transfer, the transferee executes and delivers to Merger Partner a joinder to this Agreement in the form attached hereto as Exhibit B. Any Transfer or action in violation of this Section 1.2 shall be void ab initio.
- 1.3 Solicitation. Each Holder shall, and shall instruct and use its reasonable best efforts to cause the Representatives retained by it and acting on its behalf to:
 - a) immediately cease and cause to be terminated any existing discussions, negotiations and communications with any Person that relate to any Acquisition Proposal or Acquisition Inquiry that has not already been terminated as of the date of this Agreement; and

b) not to take any action that Merger Partner would then be prohibited from taking under Section 5.5(a) of the Merger Agreement as if such Holder were Merger Partner, provided that the foregoing shall not preclude any Holder from having discussions with Merger Partner or Public Company regarding the Merger Agreement or the transactions contemplated thereby.

Notwithstanding the foregoing, from and after the date of the Merger Agreement until the earlier to occur of (x) the Merger Partner Merger Effective Time and (y) such date and time as the Merger Agreement shall be validly terminated pursuant thereto, if (and only if) Merger Partner is permitted, pursuant to the Merger Agreement, to have discussions or negotiations in response to an Acquisition Proposal, each Holder and its Representatives shall be permitted to participate in such discussions or negotiations with such Person making such Acquisition Proposal to the same extent as Merger Partner is permitted to do so under the Merger Agreement (including with respect to providing a voting agreement with respect to such Holder's Subject Units).

For the avoidance of doubt, and notwithstanding anything in this Agreement to the contrary, (a) no Holder's obligations under this Section 1.3 shall be construed to apply to any actions taken by any other equityholder of Merger Partner, Merger Partner or any of its Subsidiaries, the Board of Managers of Merger Partner or any committee thereof, any of the directors, managers, officers or employees of any other equityholder of Merger Partner, Merger Partner or any of their respective Subsidiaries (in their capacities as such), or the Representatives of any other equityholder of Merger Partner, Merger Partner or any of their respective Subsidiaries, (b) no Holder shall be responsible for the actions of any of the foregoing and (c) nothing in this Section 1.3 shall prevent any Holder or its Representatives, in their capacity as a member of Merger Partner's Board of Managers or as an executive officer of Merger Partner, as applicable, from engaging in any activity permitted or required by the Merger Agreement in the performance of their fiduciary obligations to Merger Partner's equityholders.

- 1.4 <u>Further Assurances</u>. Each Holder shall execute and deliver, or cause to be executed and delivered, such further certificates, instruments and other documents as Merger Partner or Public Company may reasonably request for the purpose of effectively carrying out its obligations under this Agreement.
- 1.5 <u>Additional Merger Partner Units</u>. Each Holder acknowledges and agrees that any additional Merger Partner Units with respect to which beneficial ownership is acquired by such Holder (other than pursuant to a vesting event pursuant to Merger Partner's profit interest units, if any, after the date hereof), by transfer or any other mechanism shall automatically become subject to the terms of this Agreement as though owned by such Holder as of the date hereof.

ARTICLE II Representations and Warranties of Each Holder

Each Holder, severally and not jointly, hereby represents and warrants to Public Company as to itself (and not as to any other Holder) as follows:

2.1 <u>Authorization</u>. Such Holder has all requisite capacity, power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement. This Agreement has been duly executed and delivered by or on behalf of such Holder and, assuming the due authorization, execution and delivery of this Agreement by Public Company, this Agreement constitutes a legal, valid and binding obligation of such Holder, enforceable against such Holder in accordance with its terms, except to the extent that enforcement may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Laws affecting creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law). Each Holder that is an entity (a) is duly organized, validly existing and in good standing (with respect to jurisdictions that recognize such concept or equivalent) under the Laws of the jurisdiction of its organization, (b) the execution and delivery of this Agreement and such Holder's performance of its obligations under this Agreement have been duly authorized by all necessary corporate or other organizational action on the part of such Holder and no other corporate or other organizational action on the part of such Holder is necessary to authorize the execution and delivery of this Agreement or for such Holder to perform its obligations under this Agreement and (c) no approval by any holder of such Holder's equity, membership or other interests is necessary to approve this Agreement.

2.2 Governmental Filings; No Violations; Certain Contracts.

- (a) No consent, approval, order or authorization of, or registration, declaration or filing with or notice to, any Government Authority is required to be obtained or made by such Holder in connection with the execution and delivery of this Agreement, except for (i) such filings and reports as may be required pursuant to the applicable requirements of the Securities Act, the Exchange Act and any other applicable state or federal securities, takeover and "blue sky" Laws (collectively, "Required Filings") and (ii) such consents, authorizations, filings, approvals and registrations which, if not obtained or made, are not reasonably likely to prevent, materially delay or materially impair the performance of such Holder's obligations under this Agreement.
- (b) The execution and delivery by such Holder of this Agreement does not, and the compliance with the provisions hereof will not, conflict with, or result in any violation or breach of, or default (with or without notice or lapse of time, or both) under, or give rise to a right of, or result in, termination, cancellation, modification or acceleration of any obligation or to the loss of a material benefit under, or result in the creation of any Lien in or upon any of the properties, assets or rights of such Holder under, or give rise to any increased, additional, accelerated or guaranteed rights or entitlements under, or require any consent, waiver or approval of any Person pursuant to, any provision of (i) any written agreement to which such Holder is a party or by which such Holder or any of his properties or assets may be bound, (ii) any Law applicable to such Holder or by which such Holder or any of its properties or assets may be bound or (iii) following a valid Transfer pursuant to Section 1.2, with respect to such Holder that is an entity, the organizational documents of such Holder, except as would not, individually or in the aggregate, be reasonably expected to prevent or materially delay the ability of such Holder to perform such Holder's obligations under this Agreement or to consummate the transactions contemplated hereby.

- 2.3 Ownership of Subject Units; Voting Power. Exhibit A hereto correctly sets forth the number of Subject Units held beneficially by such Holder, as of the date of this Agreement. Such Holder is the sole beneficial holder of all of its Subject Units and has full voting power and power of disposition with respect to all such Subject Units free and clear of any liens, claims, proxies, voting trusts or agreements, options or any other encumbrances or restrictions on title, transfer or exercise of any rights of an equityholder in respect of such Subject Units (collectively, "Encumbrances"), except for any such Encumbrance that may be imposed pursuant to (a) (i) this Agreement, (ii) the organizational documents of Merger Partner, or (iii) any applicable restrictions on transfer under the Securities Act, including the rules and regulations promulgated thereunder, or any "blue sky" Laws of the various states of the United States or (b) expressly contemplated by the Merger Agreement (collectively, "Permitted Liens"). Except pursuant to this Agreement or the Merger Agreement or as set forth in the organizational documents of Merger Partner, no Person has any contractual or other right or obligation to purchase or otherwise acquire any of the Holder's Subject Units.
- 2.4 <u>Disclosure of All Units Owned</u>. Such Holder does not beneficially own any equity securities of Merger Partner other than the Subject Units any options, warrants, or other rights to acquire any additional equity securities of Merger Partner or any security exercisable for or convertible into equity securities of Merger Partner, in each case, as set forth on <u>Exhibit A</u> attached hereto.
- 2.5 <u>Reliance</u>. Such Holder understands and acknowledges that Public Company is entering into the Merger Agreement in reliance upon such Holder's execution, delivery and performance of this Agreement.

ARTICLE III Representations and Warranties of Public Company

Public Company hereby represents and warrants to each Holder as follows:

- (a) Public Company is an entity duly organized, validly existing and in good standing (with respect to jurisdictions that recognize such concept or equivalent) under the Laws of the jurisdiction of its organization.
- (b) Public Company has all requisite capacity, power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement. This Agreement has been duly executed and delivered by or on behalf of Public Company and, assuming the due authorization, execution and delivery of this Agreement by the Holder, this Agreement constitutes a legal, valid and binding obligation of Public Company, enforceable against Public Company in accordance with its terms, except to the extent that enforcement may be limited by applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and other Laws affecting creditors' rights generally and by general equitable principles (whether considered in a proceeding in equity or at law).

ARTICLE IV General Provisions

- 4.1 <u>Termination</u>. This Agreement and all obligations, covenants and agreements contained herein, including the voting agreements contemplated hereby and the proxies granted hereunder, shall automatically terminate and cease to be effective at the earliest to occur of: (a) the Merger Partner Merger Effective Time; (b) the termination of the Merger Agreement pursuant thereto; (c) the effective date of a written agreement of the parties hereto terminating this Agreement; and (d) with respect to the Holder, the date of any modification or amendment to the Merger Agreement, as in effect on the date hereof, in a manner that reduces the amount and/or changes the form of consideration payable thereunder to the Holder or its equityholders, or otherwise adversely effects the Holder or its equityholders in any respect without the prior written approval of the Holder; <u>provided, however</u>, that in the case of any termination pursuant to clause (a) of this sentence, Section 1.4 (Further Assurances) and this ARTICLE IV shall survive such termination hereof.
- 4.2 <u>Notices</u>. All notices and other communications to any party hereunder shall be in writing and shall be deemed given if personally delivered, emailed (which is confirmed) or sent by overnight courier (providing proof of delivery) to the parties at the following addresses:

If to Public Company, to:

Calyxt, Inc. 2800 Mount Ridge Road Roseville, MN 55113

Attention: Debra Frimerman, General Counsel

E-mail: [***]

and to:

Sidley Austin LLP 787 Seventh Ave New York, New York 10019 Attention: John H. Butler Email: [***]

If to any Holder, to it at: the address listed below such Holder's name on the signature page of such Holder,

With a copy (which shall not constitute notice) to:

Cibus Global, LLC 6455 Nancy Ridge Drive San Diego, CA 92121

Attention: Rory Riggs, Chief Executive Officer, and Wade King, MD, Chief Financial Officer

E-mail: [***] and [***]

and to:
Jones Day
4655 Executive Drive
Suite 1500
San Diego, CA 92121-3134
Attention: Cameron A. Reese

E-mail: [***]

or such other address or email address as such party may hereafter specify by like notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of actual receipt by the recipient thereof if received prior to 5:00 p.m. local time in the place of receipt and such day is a business day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed not to have been received until the next succeeding business day in the place of receipt.

- 4.3 <u>Severability</u>. If any term, condition or other provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other terms, provisions and conditions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any term, condition or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible to the fullest extent permitted by applicable Law.
- 4.4 Third Party Beneficiaries. This Agreement (including the documents and instruments referred to herein) is not intended to, and does not, confer upon any Person other than the parties hereto any rights or remedies hereunder, including the right to rely upon the representations and warranties set forth herein. The representations and warranties in this Agreement are the product of negotiations among the parties hereto and are for the sole benefit of such parties. Any inaccuracies in such representations and warranties are subject to waiver by the parties hereto in accordance herewith without notice or liability to any other Person. In some instances, the representations and warranties in this Agreement may represent an allocation among the parties hereto of risks associated with particular matters regardless of the knowledge of any of the parties hereto. Consequently, Persons other than the parties hereto may not rely upon the representations and warranties in this Agreement as characterizations of actual facts or circumstances as of the date hereof or as of any other date.

4.5 Governing Law; Jurisdiction; Specific Performance; Waiver to Jury Trial.

- (a) This Agreement shall be governed by, and construed in accordance with, the Laws of the State of Delaware applicable to contracts executed in and to be performed entirely within that State, regardless of the Laws that might otherwise govern under any applicable conflict of Laws principles.
- (b) All Actions arising out of or relating to this Agreement or the Transactions shall be heard and determined in the Court of Chancery of the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over any Legal Proceeding, any state or federal court within the State of Delaware). The parties hereto hereby irrevocably (i) submit to the exclusive jurisdiction and venue of such courts in any such Legal Proceeding, (ii) waive the defense of an inconvenient forum or lack of jurisdiction to the maintenance of any such Legal Proceeding, (iii) agree to not attempt to deny or defeat such jurisdiction by motion or otherwise request for leave from any such court and (iv) agree to not bring any Legal Proceeding arising out of or relating to this Agreement or the Transactions in any court other than the Court of Chancery of the State of Delaware (or, if the Court of Chancery of the State of Delaware declines to accept jurisdiction over any Legal Proceeding, any state or federal court within the State of Delaware), except for Legal Proceedings brought to enforce the judgment of any such court. The consents to jurisdiction and venue set forth in this Section 4.5(b) shall not constitute general consents to service of process in the State of Delaware and shall have no effect for any purpose except as provided in this paragraph and shall not be deemed to confer rights on any Person other than the parties hereto. Each party hereto agrees that service of process upon such party in any Legal Proceeding arising out of or relating to this Agreement shall be effective if notice is given by overnight courier at the address set forth in Section 4.2 of this Agreement. The parties hereto agree that a final judgment in any such Legal Proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law, provided, however, that nothing in the foregoing shall restrict any party's rights to seek any post-judgment r
- (c) The parties hereto agree that irreparable damage for which monetary relief, even if available, might not be an adequate remedy, would occur in the event that any provision of this Agreement is not performed in accordance with its specific terms or is otherwise breached, including if the parties hereto fail to take any action required of them hereunder to consummate the Mergers or the other Transactions. Subject to the following sentence, the parties acknowledge and agree that (a) the parties shall be entitled to seek an injunction or injunctions, specific performance or other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof in the courts described in Section 4.5(b), this being in addition to any other remedy to which they are entitled under this Agreement and (b) the right of specific enforcement is an integral part of the Transactions and without that right none of Holders or Public Company would have entered into this Agreement or the Merger Agreement. The parties hereto acknowledge and agree that any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement in accordance with this Section 4.5(c) shall not be required to provide any bond or other security in connection with any such order or injunction.

(d) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT AND ANY OF THE AGREEMENTS DELIVERED IN CONNECTION HEREWITH OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVER, (C) IT MAKES SUCH WAIVER VOLUNTARILY AND (D) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVER AND CERTIFICATIONS IN THIS SECTION 4.5(d).

- 4.6 Entire Agreement. This Agreement, the Merger Agreement and the other agreements related to the Transactions (and all exhibits and schedules hereto and thereto) collectively constitute the entire agreement, and supersedes all other prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof and thereof.
- 4.7 <u>Amendment or Supplement</u>. Subject to compliance with applicable Law, at any time prior to the Merger Partner Merger Effective Time, this Agreement may be amended or supplemented in any and all respects only by written agreement of the parties hereto.
- 4.8 <u>Headings</u>. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 4.9 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts (including by facsimile or electronic mail), each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties hereto and delivered to the other parties hereto.

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to the date first written above.	be executed by their duly authorized respective officers as of
	CALYXT, INC.
	Ву:
	Name:
	Title:

[Signature Page to Voting Agreement]

IN WITNESS WHEREOF, the parties hereto have the date first written above.	e caused this Agreement to be executed by their duly authorized respective officers as of
[EQUITYHOLDER]	
Signature of [EQUITYHOLDER]	
Printed Name of [EQUITYHOLDER]	
Address of [EQUITYHOLDER] for purposes of Section 4.2 (N	lotices) of this Agreement:

EXHIBIT A

Subject Units

EXHIBIT B FORM OF JOINDER

Exhibit C-1 First Blocker Merger Certificate of Merger

Exhibit C-2 Form of Blocker Certificate of Merger

Exhibit C-3 Merger Partner Certificate of Merger

Exhibit D-1 Public Company Directors Exhibit D-2 Public Company Officers

Exhibit E Interim Funding Term Sheet

Project Calypso

Proposed Calvoso Funding Terms

All terms not otherwise defined herein shall have the meanings assigned to them in the Merger Agreement

- 1. **Amount**: line of credit up to \$3,000,000; provided, however, that the line of credit may be increased to \$4,000,000 at the Merger Partner's sole discretion at which point the Outside Date would automatically be extended to June 30, 2023.
- 2. **Availability**: becomes available to Public Company upon the earlier to occur of (a) March 15, 2023 and (b) Public Company's unrestricted cash balance dropping below \$1,500,000.
- 3. **Draws**: Public Company may draw in \$500,000 increments to the extent necessary to maintain its unrestricted cash balance at \$1,500,000 or above. Merger Partner will wire funds within two business days of a written request for a draw from Public Company.
- 4. **Use of Funds**: Public Company may only use the funds for operating expenses incurred in the ordinary course of business consistent with past practice and consistent with the negative covenants in the Merger Agreement. For the avoidance of doubt, this will exclude capex that has not been disclosed other than equipment repairs in the ordinary course of business, bonuses to officers and employees, dividends to stockholders, and loans to affiliates.
- 5. **Maturity**: Third Business Day after Closing of the Transactions (subject to acceleration described in Section 7 and termination of the Merger Agreement by Merger Partner as described in Section 8).¹
- 6. **Interest**: No interest
- 7. **Acceleration Event**. The Loan will automatically accelerate and become immediately due and payable in connection with a voluntary bankruptcy or involuntary bankruptcy that is not dismissed in 180 days related to . No further draws may be made under the Loan following the occurrence of a bankruptcy related event.
- 8. **Repayment**: 100% forgiven if the Merger Agreement is terminated other than in connection with (a) a termination of the Agreement by Merger Partner resulting from the Fraud or Willful Breach of Public Company, in which case Public Company will no longer be permitted to make draws under the loan and loan will become due and payable on the date that is two years from the date funds are first drawn or (b) (i) a termination of the Agreement by Public Company pursuant to Section 8.1(g) of the Agreement or (ii) a termination of the Agreement by Merger Partner pursuant to Section 8.1(f) of the Agreement, in which case Public Company will no longer be permitted to make draws under the loan and the loan will become due and payable on the date that is 30 days of termination of the Agreement. The Closing of the Transactions contemplated by the Merger Agreement will not constitute a termination for purposes of this paragraph.

NTD: It is the parties expectation that the Loan will be 100% repaid immediately prior to the closing of the Merger Agreement pursuant to a Contribution Agreement pursuant to which all the assets of Public Company will be transferred to Merger Partner first in repayment of the Loan and thereafter in exchange for the Merger Partner equity provided in the Merger Agreement (or such other structure or documentation requested by Merger Partner providing substantially similar economics to Public Company).

- 9. **Security**: Loan will be unsecured.
- 10. **Other Provisions**: The Loan will not have any other events of default or events of acceleration that are not expressly provided for in this term sheet.

Exhibit F Amendment to License Agreement

[See attached.]

FIRST AMENDMENT TO THE LICENSE AGREEMENT

This FIRST AMENDMENT TO THE LICENSE SERVICES AGREEMENT (the "First Amendment") is made and entered into effective as of _____, 2023 (the "First Amendment Effective Date") by and between Cellectis S.A., a corporation existing and registered under the laws of France ("Cellectis"), and Calyxt, Inc., a corporation existing and registered under the laws of Delaware ("Calyxt").

WHEREAS, the Parties entered into that certain License Agreement dated July 25, 2017 (the "Agreement").

WHEREAS, the Parties wish to amend the Agreement in accordance with the terms as set forth in this First Amendment.

NOW, THEREFORE, the Parties agree as follows:

ARTICLE 1 DEFINITIONS

Section 1.01 *Generally*. All Capitalized terms used in this First Amendment not otherwise defined herein have the meaning assigned to them in the Agreement.

ARTICLE 2 AMENDMENTS

Section 2.01 Bare Sublicense Revenue. The definition of "Bare Sublicense Revenue" in Article 1 of the Agreement is hereby in its entirety and replaced with the following:

"Bare Sublicense Revenue" means any and all consideration, payments and revenue (including the fair market value of any non-cash consideration) received by Calyxt pursuant to any Bare Sublicense to the extent attributable to the rights in the Licensed Cellectis Patents and the Licensed Plant Patents that are granted to a third party in such Bare Sublicense.

Section 2.02 Calyxt Field. The definition of "Calyxt Field" in Article 1 of the Agreement is hereby deleted in its entirety and replaced with the following:

"Calyxt Field" means the field of researching, developing and commercializing microorganism, agricultural and food products, including, but not limited to traits, seeds, proteins, oils, carbohydrates, food, and food and animal feed ingredients, excluding (i) any application in connection with animals and animal cells and (ii) therapeutic applications.

Section 2.03 Licensed Plant Patents. The following definition is inserted into Article 1 of the Agreement in alphabetical order:

"Licensed Plant Patents" means (i) each Patent set forth on Exhibit A to the First Amendment, (ii) all foreign equivalents and counterparts to each Patent described in clause (i), and (iii) each Patent claiming priority to or having common priority with each Patent described in clauses (i) and (ii) anywhere in the world.

Section 2.04 *Licensed Cellectis Patents*. The definition of "Licensed Cellectis Patents" in Article 1 of the Agreement is hereby deleted in its entirety and replaced with the following:

"Licensed Cellectis Patents" means any and all Patents that are: (i) related to the Calyxt Field; (ii) necessary for Calyxt to operate in the Calyxt Field; and (iii) Licensable by Cellectis and existing as of the Effective Date, in each case excluding Licensed Plant Patents.

Section 2.05 *Patent-Related Expenses*. The definition of "**Patent-Related Expenses**" in Article 1 of the Agreement is hereby deleted in its entirety and replaced with the following:

"Patent-Related Expenses" means (i) external costs and expenses (including out-of-pocket attorneys' fees, patent agent fees and governmental filing fees) that Cellectis or any of its Affiliates actually incurs in prosecuting and maintaining the Licensed Cellectis Patents which are exclusively and solely related to the Calyxt Field, and (ii) the Pro Rata Share.

Section 2.06 Pro Rata Share. The following definition is inserted into Article 1 of the Agreement in alphabetical order:

"Pro Rata Share" means (i) external costs and expenses (including out-of-pocket attorneys' fees, patent agent fees and governmental filing fees) that Cellectis or any of its Affiliates actually incurs in prosecuting and maintaining a Licensed Plant Patent which are exclusively and solely related to the Calyxt Field *multiplied by* (ii) a fraction, (A) the *numerator* of which is one, and (B) the *denominator* of which is (x) the number of express third party licensees of such Licensed Plant Patent, *plus* (y) one (1). By way of example only, if a Licensed Plant Patent is licensed to four third party licensees, then subclause (ii) of the definition of Pro Rata Share for this Licensed Plant Patent shall be 1/5. For the avoidance of doubt, if a Licensed Plant Patent is licensed to a third party and the grant of license collectively includes such third party's Affiliates, for the purposes of calculating the Pro Rata Share, such third party and its Affiliates shall be counted collectively as one third party licensee.

Section 2.07 Calyxt License. Section 2.01 of the Agreement is hereby deleted in its entirety and replaced with the following:

Section 2.01. Calyxt License.

- (i) Licensed Cellectis IP. Subject to the terms and conditions of this Agreement, Cellectis hereby grants to Calyxt an exclusive (except as otherwise provided herein and subject to existing licenses granted by Cellectis to third parties prior to the Effective Date) worldwide, perpetual license, with the right to sublicense (in accordance with Section 2.03) under the Licensed Cellectis IP to use, have used, make, have made, sell, have sold, offer for sale, export, import and otherwise exploit any and all Calyxt Licensed Products within the Calyxt Field (the "Calyxt License"). Notwithstanding the foregoing, the Calyxt License shall be non-exclusive solely in the Non-Exclusive Field, such that the rights as set forth above in this Section 2.01 are granted to Calyxt on a non-exclusive basis in the Non-Exclusive Field under the Calyxt License.
- (ii) *Licensed Plant Patents*. Subject to the terms and conditions of this Agreement, Cellectis hereby grants to Calyxt a non-exclusive, worldwide, perpetual, irrevocable, royalty-free and fully paid-up license (with the right to sublicense to any Person through multiple tiers and without any of the restrictions imposed by Section 2.03) under the Licensed Plant Patents to use, make, have made, sell, offer to sell, export, import and otherwise exploit the Licensed Plant Patents within the Calyxt Field.

Section 2.08 UMinn License. Section 2.04 of the Agreement is hereby deleted in its entirety and replaced with the following:

Section 2.04. UMinn license. The Parties acknowledge and agree that Calyxt has received a copy of the UMinn License and certain Licensed Cellectis IP is owned by the University of Minnesota ("UMinn IP") and the Calyxt License with respect to the UMinn IP is granted as a sublicense under, and subject to the terms and conditions of, the UMinn License. Accordingly, in exercising its rights under the Calyxt License, Calyxt shall comply with any and all terms and conditions of the UMinn License as they would apply to Calyxt as a sublicensee with respect to any UMinn IP. Without limiting the generality of the foregoing, promptly following receipt of written notice thereof, Calyxt shall reimburse Cellectis for any and all payments required to be made by Cellectis to the University of Minnesota pursuant to Sections 11.6.4 (Milestone Payments), 11.11 (Annual Fee), and 11.12 (Commercialization Fee) of the UMinn License, but solely to the extent that any such payments are required as a result of the applicable activities of Calyxt hereunder or thereunder. Calyxt shall not be liable to the University of Minnesota for any other payments and as party to the UMinn License, as between the Parties, Cellectis will be responsible for all such payments owed to UMinn under the UMinn License. Notwithstanding anything to the contrary in this Agreement, Calyxt can deduct from any payment owed by Calyxt to Cellectis hereunder for any particular activity any amount reimbursed to Cellectis under this Section 2.04 that arises from the same activity. Without the prior written consent of Calyxt, Cellectis shall not (A) terminate the UMinn License, or (B) amend or waive any rights under the UMinn License in any manner that would reasonably be expected to have a material adverse effect on any of Calyxt's rights under this Agreement. In addition, Calvxt shall provide, and shall cause its sublicensees to provide, to Cellectis all reports, information, and other assistance in connection with Calyxt's and its sublicensees' activities pursuant to this Agreement that are reasonably required to enable Cellectis to comply with its obligations under the UMinn License solely to the extent that such obligations relate to Calyxt's activities pursuant to this Agreement.

Section 2.09 Calyxt Improvements. Section 2.05(b) of the Agreement is hereby deleted in its entirety.

Section 2.10 No Other Licenses. Section 2.07 of the Agreement is hereby deleted in its entirety and replaced with the following:

Section 2.07. No Other Licenses. Except as expressly provided in this Agreement, no other licenses are granted to either Party under this Agreement. Each Party acknowledges and agrees that (a) any use by Calyxt or any of its sublicensees of the Licensed Cellectis IP or Licensed Plant Patents outside the scope of the Calyxt License or the Licensed TALEN Mark outside the scope of the Calyxt TM License is expressly prohibited and (b) any use by Cellectis or any of its sublicensees of the Calyxt Improvements is expressly prohibited.

Section 2.11 *Prosecution and Maintenance*. Section 9.1(a) of the Agreement is hereby amended by inserting the words "and Licensed Plant Patents" following the words "Licensed Cellectis IP," in the first sentence of such section.

Section 2.12 Prosecution and Maintenance. Section 9.1(b) is deleted in its entirety and replaced with the following:

(b) As between the Parties, Calyxt shall have the sole and exclusive right to prosecute and maintain all Intellectual Property Rights owned or otherwise controlled by Calyxt or any of its Affiliates, including all Intellectual Property Rights in or to any Calyxt Improvements.

Section 2.13 Prosecution and Maintenance. The following section (c) is added in alphabetical order to Section 9.01 of the Agreement:

(c) Within sixty (60) days of the First Amendment Effective Date, and thereafter on an annual basis, at least thirty (30) days prior to the beginning of each calendar year, Cellectis will provide to Calyxt a budget for the estimated Patent-Related Expenses that are expected to be incurred during such calendar year. Cellectis will consider in good faith to implement any changes to such budget that are reasonably requested by Calyxt (each an "Annual Prosecution Budget"). Cellectis will use commercially reasonable efforts to adhere to the Annual Prosecution Budget and will not incur any Patent-Related Expenses in an unreasonable manner (including by filing Patents subject to Patent-Related Expenses in multiple countries without first obtaining Calyxt's prior written consent). If practicable, Cellectis will notify Calyxt before incurring any additional Patent-Related Expense and the Parties will discuss strategies to mitigate any such additional Patent-Related Expense. Calyxt will be responsible for reimbursing Cellectis for Patent-Related Expenses in accordance with Section 5.03.

Section 2.14 Survival. Section 10.04 of the Agreement is hereby deleted in its entirety and replaced with the following:

Section 10.04. *Survival*. Notwithstanding anything in this Agreement to the contrary, Sections 2.01(ii), 2.05(a), 2.07, 3.02, 5.04 (only to the extent any royalties or Bare Sublicense Revenue are due to Cellectis during the applicable reporting period), 6.02, 6.03, 9.01(b) and 10.04, and Articles 7, 8 and 11 shall survive any expiration or termination of this Agreement.

Section 2.15 Change of Control. The Agreement is hereby amended by adding the following as a new Section 11.13:

Section 11.13. Change of Control. Unless that certain Lease Guaranty, dated as of September 1, 2017, executed by Cellectis, S.A. in favor of NLD Mount Ridge LLC as landlord under that certain Lease Agreement, dated as of the same date, with Calyxt, Inc. (as the same may be amended or restated from time to time, the "Lease Guaranty") has been terminated such that Cellectis S.A. and its Affiliates have no further liabilities or obligations thereunder, and with no amounts having been paid by Cellectis S.A. or its Affiliates thereunder other than those amounts that have been fully repaid to Cellectis S.A. or its Affiliates prior to the consummation of such Change of Control, Cellectis may terminate this Agreement by giving written notice to Calyxt, such termination to take effect forthwith or as otherwise stated in the notice, in case of any Change of Control of Calyxt. For the purpose of this Section 11.13: (a) "Change of Control" means the occurrence of any of the following events: (i) any third party acquires direct or indirect Control of Calyxt, by any means (including acquisition of shares, share exchange or share transfer); or (ii) Calyxt, directly or indirectly, conveys, transfers, divests or leases (including general succession and all types of corporate split) in one or more transactions to any third party either: (x) all or substantially all of the assets of Calyxt or (y) all or substantially all of its assets that are material to the purpose of performance of its obligations under this Agreement, and (b) "Control" means the ownership of more than 50% (fifty percent) of the issued share capital or other equity interest or the legal power to direct or cause the direction of the general management and policies of a Person, whether directly or indirectly.

ARTICLE 3 MISCELLANEOUS

Section 3.01 Other Provisions. All provisions of the Agreement not expressly amended herein shall remain in full force and effect.

Section 3.02 No Calyxt Improvements. Calyxt represents and warrants to Cellectis that as of the date hereof there are no Calyxt Improvements.

Section 3.03 Miscellaneous. The provisions of Article 11 of the Agreement apply to the First Amendment mutatis mutandis.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this First Amendment to be duly executed by their respective authorized officers as of the date first written above.

By:
Name:
Title:
CALYXT, INC.
By:
Name:
Title:

CELLECTIS S.A.

[Signature Page to First Amendment to License Agreement]

Exhibit A Licensed Plant Patents

[See Attached.]

Exhibit G Allocation Schedule

Exhibit H — TERM SHEET TAX RECEIVABLES AGREEMENT

Exhibit I Form of Unitholder Representative Engagement Agreement

[See attached.]

Amendment to Calyxt, Inc. 2021 Executive Severance Plan

Effective January 13, 2023

That certain Calyxt, Inc. 2021 Executive Severance Plan (the "Plan") is hereby amended pursuant to Section 8.4 thereof effective as of the date first written above as follows.

Section 4.1(b) is deleted and replaced in its entirety with the following:

- Time and Form of Payment. If a Participant is entitled to a Severance Benefit, the Severance Benefit will be paid as follows, unless otherwise specified in the Participation Agreement:
 - (i) **In General**. Except as otherwise provided in paragraphs (ii) and (iii), below, the Participant's Severance Benefit will be paid in substantially equal installments over the Severance Coverage Period and in accordance with the Company's payroll practices. Each such installment shall be considered a separate payment for purposes of Section 409A.
 - (ii) **Change-in-Control Period**. If a Participant's Qualifying Termination occurs during a Change-in-Control Period after the applicable Change in Control, (A) the portion of such Participant's Severance Benefit that is not exempt from Section 409A under Treasury Regulation section 1.409A-1(b)(4) or 1.409A-1(b)(9)(iii) will be paid in a lump sum on or before the 60th day following the Participant's Qualifying Termination date, and (B) the portion of such Participant's Severance Benefit that is exempt from Section 409A under Treasury Regulation section 1.409A-1(b)(4) or 1.409A-1(b)(9)(iii) will be paid as set forth in paragraph (i).
 - (iii) **Time of Payment under Section 409A**. To comply with Section 409A (A) any payment under the Plan that is subject to Section 409A and that is contingent on a termination of employment is contingent on a "separation from service" within the meaning of Section 409A and (B) if, upon separation from service, the Participant is a "specified employee" within the meaning of Section 409A, any payment under the Plan that is subject to Section 409A and would otherwise be paid within six months after the Participant's separation from service will instead be paid in the seventh month following the Participant's separation from service."

Calyxt and Cibus Announce Definitive Merger Agreement to Create Industry-Leading Precision Gene Editing and Trait Licensing Company

- The merger will bring together the technology platforms and facilities of two pioneering companies to create a leader in precision gene editing
 focused on driving sustainable agriculture and the availability of renewable low-carbon ingredients –
- Upon closure of the merger, Rory Riggs, Cibus Co-Founder, will become Chief Executive Officer and Chairman of the combined company and the Cibus leadership team will lead the new organization –

ROSEVILLE, Minn., and SAN DIEGO – January 17, 2023 – Calyxt, Inc. (Nasdaq: CLXT), a plant-based synthetic biology company, and Cibus, a leader in precision gene editing in agriculture, today announced that both companies have entered into a definitive merger agreement under which Calyxt and Cibus will merge in an all-stock transaction. The merger will create a new industry-leading company that combines the two pioneers in agriculture-based gene editing and establishes one of the world's most sophisticated facilities for trait development and next-generation plant breeding.

The combined company will be a leader in two key applications for gene editing in agriculture:

- 1) **Productivity Traits**: Productivity traits are a key basis of competition in the "seed and trait" business. The key focus of Cibus' patented gene editing platform, the Rapid Trait Development System™ (*RTDS*®), is the development of a new class of productivity traits in seeds addressing the sustainability of farming by increasing crop yields and reducing inputs such as fungicides, herbicides, pesticides, and fertilizers.
- 2) Renewable Low-Carbon Ingredients: Gene editing is a key tool in the development of sustainable low-carbon ingredients that can replace fossil fuel-based ingredients and diesel fuel. This is a key pillar of the Net Carbon Zero Climate 2040 goals and the global movement to reduce greenhouse gas emissions.

Cibus has a broad pipeline of productivity traits and collaborations with several leading seed companies. It is currently launching three important productivity traits: one in canola and two in rice with transfers to customers for commercialization beginning in the first half of 2023. In addition, Cibus and Calyxt are both working with leading consumer product companies to develop more sustainable ingredients.

Under the terms of the merger agreement, Calyxt will issue shares of its common stock to Cibus shareholders in an exchange ratio such that upon completion of the merger, Calyxt shareholders will own approximately 5% of the combined company, subject to adjustments permitted by the merger agreement. The Boards of Directors of both companies have unanimously approved the transaction. Concurrent with the execution of the merger agreement, certain officers of Calyxt, all of Calyxt's directors, and Cellectis, S.A., Calyxt's largest shareholder, executed support agreements in favor of the merger. These support agreements provide 49.9% approval from Calyxt shareholders. A majority of Cibus' shareholders have also provided support agreements in favor of the transaction. The merger is expected to close in the second quarter of 2023, subject to customary closing conditions, including approval of the merger by the shareholders of Calyxt.

"Cibus is an excellent strategic fit for Calyxt given our complementary technology platforms, and the merger provides a great opportunity to leverage multiple synergies to drive innovation and shareholder value," said Michael A. Carr, President and Chief Executive Officer of Calyxt. "I am deeply proud of the significant accomplishments made by our team and their commitment to further science in significant ways."

"We believe 2023 will be a transformative year for Cibus due to many of our major accomplishments that are coming to fruition this year," said Rory Riggs, Co-Founder, Chief Executive Officer, and Chairman of Cibus. "Through our Trait Machine™, which is planned to come on-line in a stand-alone facility in 2023, we are preparing to transfer our first product: a pod shatter reduction trait in canola in the elite germplasms of five different canola seed companies. In addition, in 2023, we are preparing to transfer two different herbicide resistance traits in rice in the elite germplasm of a leading North American rice seed company. Each of the three developed traits have been determined not to be regulated articles through the USDA's "Am I Regulated" process, which was replaced with the SECURE Rule's confirmation process. The successful development and transfer of these first traits to our customers will be an important milestone in the gene editing industry. It will also be a major commercial milestone in the growth of Cibus. We are very excited by the ability of this transaction – through the addition of both Calyxt's state-of-the-art gene editing facilities and its talented scientists – to accelerate our growth."

"2023 is also expected to be an important year for the regulations covering gene editing in agriculture with the European Commission and many other countries reviewing their laws for regulating certain new genomic techniques," added Peter Beetham, President and Chief Operating Officer of Cibus. "Notably, the UK is on course to introduce new laws broadly consistent with the laws from a growing list of important trading nations including the United States, Canada, Argentina, Brazil and Japan. These regulatory changes, if adopted, promise to enable more open and fair trade in agriculture for these new technologies and encourage research and innovation, greatly increasing the potential of gene editing to contribute globally to sustainable agriculture, food security and climate resilience. This merger positions the combined companies to be a leader in this important new industry and to accelerate our progress."

"We are excited to combine our joint wealth of experience that consolidates landmark intellectual property in agricultural gene editing including Transcription Activator-like Effector Nucleases ("TALENs") and Oligonucleotide Directed Mutagenesis ("ODM")," said Greg Gocal, Co-Founder, Executive Vice President and Chief Scientific Officer of Cibus. "The merger of two of the original pioneers in what has become the field of gene editing in agriculture, will enable new and expanded partnerships and opportunities to make farming more productive and to create a new generation of sustainable ingredients that are renewable and low-carbon."

"Consolidating our innovative platforms into one entity brings a myriad of opportunities for step change innovations in the areas of productivity and low-carbon ingredients. I'm particularly excited for the teams of talented scientists from both companies to combine. I truly believe we will see a force multiplier effect of scientific discovery and operationalization to deliver these customer-demanded products in record time," added Travis Frey, Chief Technology Officer of Calyxt.

Upon closing of the transaction, the combined company, renamed Cibus Inc., will trade on the Nasdaq Capital Market. The current Cibus management team will lead the new combined organization with Rory Riggs assuming the roles of Chairman of the Board of Directors and Chief Executive Officer. Mr. Riggs has been the founder and Chairman/Director of many successful healthcare and biotechnology companies

including Royalty Pharma, Fibrogen and Sugen and was the President and Director of Biomatrix (Nasdaq: BIOX) until its sale to Genzyme/Sanofi, where he launched Synvisc®, a leading biologic for the treatment osteoarthritis. Corporate headquarters for the combined company will be located in San Diego, California and Calyxt's offices, laboratory, and breeding facilities in Roseville, Minnesota will remain operational as a key site for the combined company.

Canaccord Genuity served as financial advisor and Sidley Austin served as legal counsel to Calyxt, and Jones Day served as legal counsel to Cibus with respect to the transaction.

About Calyxt

Calyxt (Nasdaq: CLXT) is a plant-based synthetic biology company. Calyxt leverages its proprietary PlantSpring[™] technology platform and Plant Cell Matrix[™] biomass to engineer plant metabolism to produce innovative high value plant-based chemistries for use in customers' materials and products. As plant-based solutions, Calyxt's synthetic biology products can be used in helping customers meet their sustainability targets and financial goals. Calyxt's diversified offerings are primarily delivered through its proprietary BioFactory[™] production system. For more information, visit www.calyxt.com.

About Cibus

Cibus® is a leading agricultural technology company that develops, and licenses gene edited plant traits to seed companies. Its traits enable farmers to manage productivity and sustainability challenges such as diseases, pests, weeds, fertilizer use and climate change. Cibus' goal is to use its trait technology to create a new generation of crops that are more adaptable to their environment and have increased yields while requiring less chemicals. Cibus' patented RTDS® technology platform has enabled agriculture's first standardized end-to-end gene editing trait prototype and production system: the Trait Machine™. The Trait Machine represents a technological breakthrough in plant breeding that broadens the range and scale of possible trait solutions that makes more diverse germplasm accessible, materially shortens trait breeding timelines, and shortens the time to market traits. Cibus' technologies and traits are accelerating agriculture's jump to a climate smart, more sustainable crop production system and the industry's move to sustainable low-carbon ingredients. For more information, visit www.cibus.com.

Cautionary Statement Regarding Forward-Looking Statements

The information included in this press release includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 (the "Securities Act") and Section 21E of the Securities Exchange Act of 1934. All statements, other than statements of present or historical fact included herein, regarding the transactions, the ability of the parties to the merger agreement to consummate the transactions, the benefits of the transactions, the combined company's future performance following the transactions, and the potential for global regulatory developments, as well as Calyxt's, Cibus' and the combined company's respective strategies, future operations, financial positions, prospects and plans as well as the objectives of management are forward-looking statements. Words such as "anticipate," "believes," "expects," "potential," "continues," "may," "will," "approximately," "plans," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words.

These forward-looking statements are based on the current expectations and assumptions of Cibus' and Calyxt's management about future events and are based on currently available information as to the outcome and timing of future events. Forward-looking statements are subject to risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of Calyxt and Cibus. These risks include, but are not limited to, (i) the risk that the conditions to the closing of the proposed transaction are not satisfied, including the failure to obtain stockholder approval of matters related to the proposed transaction in a timely manner or at all, (ii) uncertainties as to the timing of the consummation of the proposed transaction, (iii) risks related to Calyxt's capital resources and the ability of Calyxt and Cibus, respectively, to correctly estimate and manage their respective operating expenses and expenses associated with the proposed transaction, (iv) risks related to Calyxt's continued listing on the Nasdag Capital Market until closing of the proposed transaction, (v) risks associated with the possible failure to realize certain anticipated benefits of the proposed transaction, including with respect to future financial and operating results; (vi) uncertainties regarding the impact that any delay in the closing would have on the anticipated cash resources of the combined company upon closing and other events and unanticipated spending and costs that could reduce the combined company's cash resources; (vii) the potential for the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the merger agreement; (viii) the possible effect of the announcement, pendency or completion of the merger on Calyxt's or Cibus' business relationships, operating results and business generally; (ix) risks related to unexpected costs related to the merger; (x) the potential for, and uncertainty associated with the outcome of, any legal proceedings that may be instituted against Calvxt or Cibus or any of their respective directors or officers related to the merger agreement or the transactions contemplated thereby; (xi) risks associated with the ability of Calyxt and Cibus to protect their respective intellectual property rights; (xii) the potential impact of competitive responses to the proposed transaction and changes in expected or existing competition; (xiii) the possibility that Calyxt, Cibus or the combined company may be adversely affected by other economic, business, or competitive factors; (xiv) risks associated with the loss of key employees of Calyxt or Cibus; (xv) risks associated with changes in applicable laws or regulations and the potential impact of such changes on Calyxt's, Cibus' or the combined company's ability to advance product development and commercialization; and (xvi) other risks and uncertainties identified from time to time in documents filed or to be filed with the Securities and Exchange Commission (the "SEC") by Calyxt or the combined company, including those discussed in the "Risk Factors" section of Calyxt's Annual Report on Form 10-K, which was filed with the SEC on March 3, 2022, and Quarterly Reports on Form 10-Q, which were filed with the SEC on August 4, 2022 and November 3, 2022, respectively. Should one or more of the risks or uncertainties occur, or should underlying assumptions prove incorrect, actual results and plans could differ materially from those expressed in any forward-looking statements. In addition, the forward-looking statements included in this press release represent Calyxt's and Cibus' views as of the date hereof. Calyxt and Cibus anticipate that subsequent events and developments will cause the respective company's views to change. Calyxt and Cibus specifically disclaim any obligation to update such forward-looking statements in the future, except as required under applicable law. These forward-looking statements should not be relied upon as representing Calyxt's or Cibus' views as of any date subsequent to the date hereof.

Important Additional Information

In connection with the proposed transactions, Calyxt intends to file materials with the SEC, including a registration statement on Form S-4 (the "Form S-4"), which will include a proxy statement of Calyxt for the stockholders of Calyxt and that will serve as a prospectus of Calyxt and an information statement of Cibus, and other documents relating to the proposed transaction. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THESE MATERIALS, INCLUDING THE FORM S-4 AND THE PROXY STATEMENT / PROSPECTUS INCLUDED THEREIN, WHEN THEY BECOME AVAILABLE, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT CALYXT, CIBUS AND THE PROPOSED TRANSACTION. After the Form S-4 is declared effective, the definitive proxy statement to be included in the Form S-4 will be mailed to Calyxt stockholders as of a record date to be established for voting on the matters related to the proposed transaction to be considered at a special meeting of Calyxt stockholders. The Form S-4, the proxy statement / prospectus included therein, and other materials filed by Calyxt with the SEC may be obtained free of charge from the SEC's website (www.sec.gov) or from Calyxt by directing a request to: Calyxt, Inc., 2800 Mount Ridge Road, Roseville, MN 55113.

Participants in the Solicitation

Calyxt, Cibus and their respective directors, executive officers and other members of management may be deemed to be participants in the solicitation of proxies with respect to the proposed transaction under the rules of the SEC. Information about the directors and executive officers of Calyxt is set forth in Calyxt's Annual Report on Form 10-K for the fiscal year ended December 31, 2021, which was filed with the SEC on March 3, 2022, and its definitive proxy statement for its 2022 Annual Meeting of Shareholders, which was filed with the SEC on April 19, 2022.

Other information regarding persons who may, under the rules of the SEC, be deemed to be participants in the proxy solicitation and a description of their interests in the transaction, by security holdings or otherwise, will be included in the proxy statement/prospectus included in the Form S-4 and other relevant materials to be filed with the SEC regarding the proposed transaction when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from Calyxt or the SEC's website, as indicated above.

No Offer or Solicitation

This press release shall not constitute an offer to sell or a solicitation of an offer to buy the securities of Calyxt, Cibus or the combined company, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offering of securities shall be made in the United States except by means of a prospectus meeting the requirements of Section 10 of the Securities Act. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, a public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone or internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.

###

Contacts

For Calyxt:
Bill Koschak
investors@calyxt.com
651-425-1754

For Cibus:

Investor Relations: Karen Troeber <u>ktroeber@cibus.com</u> 858-450-2636

Media Relations: Colin Sanford colin@bioscribe.com 203-918-4347

Chris Tutino ctutino@cibus.com 919-356-9163

Filed by Calyxt, Inc.
Pursuant to Rule 425 under the Securities Act of 1933
and deemed filed pursuant to Rule 14a-12
under the Securities Exchange Act of 1934

Subject Company: Calyxt, Inc. Commission File No.: 001-38161

On January 17, 2023, Cibus Global, LLC published the following post on LinkedIn, Facebook and Twitter:

Today, Cibus and Calyxt are announcing a definitive merger agreement that will create the ag industry's leading company in plant gene editing and trait licensing. Learn more about this exciting development: [Insert Link to Release Here]

Cautionary Statement Regarding Forward-Looking Statements

The information included in this communication includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 (the "Securities Act") and Section 21E of the Securities Exchange Act of 1934. All statements, other than statements of present or historical fact included herein, regarding the transactions, the ability of the parties to the merger agreement to consummate the transactions, the benefits of the transactions, the combined company's future performance following the transactions, and the potential for global regulatory developments, as well as Calyxt's, Cibus's and the combined company's respective strategies, future operations, financial positions, prospects and plans as well as the objectives of management are forward-looking statements. Words such as "expects," "continues," "may," "will," "approximately," "intends," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words.

These forward-looking statements are based on the current expectations and assumptions of Cibus's and Calyxt's management about future events and are based on currently available information as to the outcome and timing of future events. Forward-looking statements are subject to risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of Calyxt and Cibus. These risks include, but are not limited to, (i) the risk that the conditions to the closing of the proposed transactions are not satisfied, including the failure to obtain stockholder approval of matters related to the proposed transactions in a timely manner or at all, (ii) uncertainties as to the timing of the consummation of the proposed transactions, (iii) risks related to Calyxt's capital resources and the ability of Calyxt and Cibus, respectively, to correctly estimate and manage their respective operating expenses and expenses associated with the proposed transactions, (iv) risks related to Calyxt's continued listing on the Nasdaq Capital Market until closing of the proposed transactions, (v) risks associated with the possible failure to realize certain anticipated benefits of the proposed transactions, including with respect to future financial and operating results; (vi) uncertainties regarding the impact that any delay in the closing would have on the anticipated cash resources of the combined company

upon closing and other events and unanticipated spending and costs that could reduce the combined company's cash resources; (vii) the potential for the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the merger agreement; (viii) the possible effect of the announcement, pendency or completion of the merger on Calyxt's or Cibus's business relationships, operating results and business generally; (ix) risks related to unexpected costs related to the merger; (x) the potential for, and uncertainty associated with the outcome of, any legal proceedings that may be instituted against Calyxt or Cibus or any of their respective directors or officers related to the merger Agreement or the transactions contemplated thereby; (xi) risks associated with the ability of Calvxt and Cibus to protect their respective intellectual property rights; (xii) the potential impact of competitive responses to the proposed transactions and changes in expected or existing competition; (xiii) the possibility that Calyxt, Cibus or the combined company may be adversely affected by other economic, business, or competitive factors; (xiv) risks associated with the loss of key employees of Calyxt or Cibus; (xv) risks associated with changes in applicable laws or regulations and the potential impact of such changes on Calyxt's, Cibus's or the combined company's ability to advance product development and commercialization, and (xvi) other risks and uncertainties identified from time to time in documents filed or to be filed with the SEC by Calyxt or the combined company, including those discussed in the "Risk Factors" section of Calvxt's Annual Report on Form 10-K, which was filed with the SEC on March 3, 2022, and Ouarterly Reports on Form 10-Q, which were filed with the SEC on August 4, 2022 and November 3, 2022, respectively. Should one or more of the risks or uncertainties occur, or should underlying assumptions prove incorrect, actual results and plans could differ materially from those expressed in any forward-looking statements. In addition, the forward-looking statements included in this communication represent Calyxt's and Cibus's views as of the date hereof. Calyxt and Cibus anticipate that subsequent events and developments will cause the respective company's views to change. Calyxt and Cibus specifically disclaim any obligation to update such forward-looking statements in the future, except as required under applicable law. These forward-looking statements should not be relied upon as representing Calyxt's or Cibus's views as of any date subsequent to the date hereof.

Important Additional Information

In connection with the proposed transactions, Calyxt intends to file materials with the Securities and Exchange Commission ("SEC"), including a registration statement on Form S-4 (the "Form S-4"), which will include a proxy statement of Calyxt for the stockholders of Calyxt and that will serve as a prospectus of Calyxt and an information statement of Cibus, and other documents relating to the proposed transactions. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THESE MATERIALS, INCLUDING THE FORM S-4 AND THE PROXY STATEMENT/PROSPECTUS INCLUDED THEREIN, WHEN THEY BECOME AVAILABLE, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT CALYXT, CIBUS AND THE PROPOSED TRANSACTIONS. After the Form S-4 is declared effective, the definitive proxy statement to be included in the Form S-4 will be mailed to Calyxt stockholders as of a record date to be established for voting on the Calyxt Stockholder Matters to be considered at the Calyxt Stockholder Meeting. The Form S-4, the proxy statement/prospectus included therein, and other materials filed by Calyxt with the SEC may be obtained free of charge from the SEC's website (www.sec.gov) or from Calyxt by directing a request to: Calyxt, Inc., 2800 Mount Ridge Road, Roseville, MN 55113.

Participants in the Solicitation

Calyxt, Cibus and their respective directors, executive officers and other members of management may be deemed to be participants in the solicitation of proxies with respect to the proposed transactions under the rules of the SEC. Information about the directors and executive officers of Calyxt is set forth in Calyxt's Annual Report on Form 10-K for the fiscal year ended December 31, 2021, which was filed with the SEC on March 3, 2022, and its definitive proxy statement for its 2022 Annual Meeting of Shareholders, which was filed with the SEC on April 19, 2022.

Other information regarding persons who may, under the rules of the SEC, be deemed to be participants in the proxy solicitation and a description of their interests in the transaction, by security holdings or otherwise, will be included in the proxy statement/prospectus included in the Form S-4 and other relevant materials to be filed with the SEC regarding the proposed transaction when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from Calyxt or the SEC's website, as indicated above.

No Offer or Solicitation

This press release shall not constitute an offer to sell or a solicitation of an offer to buy the securities of Calyxt, Cibus or the combined company, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offering of securities shall be made in the United States except by means of a prospectus meeting the requirements of Section 10 of the Securities Act. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, a public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone or internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.

Filed by Calyxt, Inc.
Pursuant to Rule 425 under the Securities Act of 1933
and deemed filed pursuant to Rule 14a-12
under the Securities Exchange Act of 1934

Subject Company: Calyxt, Inc. Commission File No.: 001-38161

On January 17, 2023, Calyxt, Inc. published the following post on LinkedIn:

We are excited to share that Calyxt and Cibus announced the signing of a definitive merger agreement to create an industry-leading precision #gene editing and #trait licensing company!

In the words of our President and CEO, Michael A. Carr, "Cibus is an excellent strategic fit for Calyxt given our complementary technology platforms, and the merger provides a great opportunity to leverage multiple synergies to drive innovation and shareholder value."

[Insert Link to Release Here]

On January 17, 2023, Calyxt, Inc. published the following post on Twitter:

We are excited to share that Calyxt and Cibus announced the signing of a definitive merger agreement to create an industry-leading precision #gene editing and #trait licensing company!

[Insert Link to Release Here]

Cautionary Statement Regarding Forward-Looking Statements

The information included in this communication includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 (the "Securities Act") and Section 21E of the Securities Exchange Act of 1934. All statements, other than statements of present or historical fact included herein, regarding the transactions, the ability of the parties to the merger agreement to consummate the transactions, the benefits of the transactions, the combined company's future performance following the transactions, and the potential for global regulatory developments, as well as Calyxt's, Cibus's and the combined company's respective strategies, future operations, financial positions, prospects and plans as well as the objectives of management are forward-looking statements. Words such as "expects," "continues," "may," "will," "approximately," "intends," the negative of such terms and other similar expressions are intended to identify forward-looking statements, although not all forward-looking statements contain such identifying words.

These forward-looking statements are based on the current expectations and assumptions of Cibus's and Calyxt's management about future events and are based on currently available information as to the outcome and timing of future events. Forward-looking statements are subject to risks and uncertainties, most of which are difficult to predict and many of which are beyond the control of Calyxt and Cibus. These risks include, but are not limited to, (i) the risk that the conditions to the closing of the proposed transactions are not satisfied, including the failure to obtain stockholder approval of matters related to the proposed transactions in a timely manner or at all, (ii) uncertainties as to the timing of the consummation of the proposed transactions. (iii) risks related to Calyxt's capital resources and the ability of Calyxt and Cibus, respectively, to correctly estimate and manage their respective operating expenses and expenses associated with the proposed transactions, (iv) risks related to Calyxt's continued listing on the Nasdag Capital Market until closing of the proposed transactions, (v) risks associated with the possible failure to realize certain anticipated benefits of the proposed transactions, including with respect to future financial and operating results; (vi) uncertainties regarding the impact that any delay in the closing would have on the anticipated cash resources of the combined company upon closing and other events and unanticipated spending and costs that could reduce the combined company's cash resources; (vii) the potential for the occurrence of any event, change or other circumstance or condition that could give rise to the termination of the merger agreement; (viii) the possible effect of the announcement, pendency or completion of the merger on Calyxt's or Cibus's business relationships, operating results and business generally; (ix) risks related to unexpected costs related to the merger; (x) the potential for, and uncertainty associated with the outcome of, any legal proceedings that may be instituted against Calyxt or Cibus or any of their respective directors or officers related to the merger Agreement or the transactions contemplated thereby; (xi) risks associated with the ability of Calyxt and Cibus to protect their respective intellectual property rights; (xii) the potential impact of competitive responses to the proposed transactions and changes in expected or existing competition; (xiii) the possibility that Calyxt, Cibus or the combined company may be adversely affected by other economic, business, or competitive factors; (xiv) risks associated with the loss of key employees of Calyxt or Cibus; (xv) risks associated with changes in applicable laws or regulations and the potential impact of such changes on Calyxt's, Cibus's or the combined company's ability to advance product development and commercialization; and (xvi) other risks and uncertainties identified from time to time in documents filed or to be filed with the SEC by Calyxt or the combined company, including those discussed in the "Risk Factors" section of Calyxt's Annual Report on Form 10-K, which was filed with the SEC on March 3, 2022, and Quarterly Reports on Form 10-Q, which were filed with the SEC on August 4, 2022 and November 3, 2022, respectively. Should one or more of the risks or uncertainties occur, or should underlying assumptions prove incorrect, actual results and plans could differ materially from those expressed in any forward-looking statements. In addition, the forward-looking statements included in this communication represent Calyxt's and Cibus's views as of the date hereof. Calyxt and Cibus anticipate that subsequent events and developments will cause the respective company's views to change. Calyxt and Cibus specifically disclaim any obligation to update such forward-looking statements in the future, except as required under applicable law. These forward-looking statements should not be relied upon as representing Calyxt's or Cibus's views as of any date subsequent to the date hereof.

Important Additional Information

In connection with the proposed transactions, Calyxt intends to file materials with the Securities and Exchange Commission ("SEC"), including a registration statement on Form S-4 (the "Form S-4"), which will include a proxy statement of Calyxt for the stockholders of Calyxt and that will serve as a prospectus of Calyxt and an information statement of Cibus, and other documents

relating to the proposed transactions. INVESTORS AND SECURITY HOLDERS ARE URGED TO READ THESE MATERIALS, INCLUDING THE FORM S-4 AND THE PROXY STATEMENT/PROSPECTUS INCLUDED THEREIN, WHEN THEY BECOME AVAILABLE, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT CALYXT, CIBUS AND THE PROPOSED TRANSACTIONS. After the Form S-4 is declared effective, the definitive proxy statement to be included in the Form S-4 will be mailed to Calyxt stockholders as of a record date to be established for voting on the Calyxt Stockholder Matters to be considered at the Calyxt Stockholder Meeting. The Form S-4, the proxy statement/prospectus included therein, and other materials filed by Calyxt with the SEC may be obtained free of charge from the SEC's website (www.sec.gov) or from Calyxt by directing a request to: Calyxt, Inc., 2800 Mount Ridge Road, Roseville, MN 55113.

Participants in the Solicitation

Calyxt, Cibus and their respective directors, executive officers and other members of management may be deemed to be participants in the solicitation of proxies with respect to the proposed transactions under the rules of the SEC. Information about the directors and executive officers of Calyxt is set forth in Calyxt's Annual Report on Form 10-K for the fiscal year ended December 31, 2021, which was filed with the SEC on March 3, 2022, and its definitive proxy statement for its 2022 Annual Meeting of Shareholders, which was filed with the SEC on April 19, 2022.

Other information regarding persons who may, under the rules of the SEC, be deemed to be participants in the proxy solicitation and a description of their interests in the transaction, by security holdings or otherwise, will be included in the proxy statement/prospectus included in the Form S-4 and other relevant materials to be filed with the SEC regarding the proposed transaction when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from Calyxt or the SEC's website, as indicated above.

No Offer or Solicitation

This press release shall not constitute an offer to sell or a solicitation of an offer to buy the securities of Calyxt, Cibus or the combined company, nor shall there be any sale of any such securities in any state or jurisdiction in which such offer, solicitation, or sale would be unlawful prior to registration or qualification under the securities laws of such state or jurisdiction. No offering of securities shall be made in the United States except by means of a prospectus meeting the requirements of Section 10 of the Securities Act. Subject to certain exceptions to be approved by the relevant regulators or certain facts to be ascertained, a public offer will not be made directly or indirectly, in or into any jurisdiction where to do so would constitute a violation of the laws of such jurisdiction, or by use of the mails or by any means or instrumentality (including without limitation, facsimile transmission, telephone or internet) of interstate or foreign commerce, or any facility of a national securities exchange, of any such jurisdiction.